# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

# **FORM 10-Q**

# $/\underline{X}/\,$ QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2011

OR

	NT TO SECTION 13 OR 15(d) OF THE HANGE ACT OF 1934
For the transition peri	od from to

Commission file number 1-11840

### THE ALLSTATE CORPORATION

(Exact name of registrant as specified in its charter)

36-3871531

(I.R.S. Employer Identification No.)

**Delaware** 

(State or other jurisdiction of incorporation or organization)

	Sanders Road, Northbrook, Ill (Address of principal executive offices		
(Regist	(847) 402-5000 strant's telephone number, including a	rea code)	
Indicate by check mark whether the registrant (1) has filed all report conths (or for such shorter period that the registrant was required to file s			
Yes X	No		
Indicate by check mark whether the registrant has submitted electrosted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) arch files).			
Yes X	No		
Indicate by check mark whether the registrant is a large accelerated celerated filer," "accelerated filer" and "smaller reporting company" in		elerated filer, or a smaller rep	porting company. See the definitions of "large
arge accelerated filer X		Accelerated filer	
on-accelerated filer (Do not check if a smaller reporting company	y)	Smaller reporting company	
Indicate by check mark whether the registrant is a shell company (as	s defined in Rule 12b-2 of the Exchar	ige Act).	
Yes	No X		
As of July 20, 2011, the registrant had 516,608,362 common shares,	, \$.01 par value, outstanding.		

# THE ALLSTATE CORPORATION INDEX TO QUARTERLY REPORT ON FORM 10-Q June 30, 2011

PART I	FINANCIAL INFORMATION	PAGE
Item 1.	Financial Statements	
	Condensed Consolidated Statements of Operations for the Three-Month and Six-Month Periods Ended June 30, 2011 and 2010 (unaudited)	1
	Condensed Consolidated Statements of Financial Position as of June 30, 2011 (unaudited) and December 31, 2010	2
	Condensed Consolidated Statements of Cash Flows for the Six-Month Periods Ended June 30, 2011 and 2010 (unaudited)	3
	Notes to Condensed Consolidated Financial Statements (unaudited)	4
	Report of Independent Registered Public Accounting Firm	49
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	

	Highlights Consolidated Net (Loss) Income Property-Liability Highlights Allstate Protection Segment Discontinued Lines and Coverages Segment Property-Liability Investment Results Allstate Financial Highlights Allstate Financial Segment Investments Highlights Investments	50 51 52 56 66 66 67 67 74
	Capital Resources and Liquidity Highlights Capital Resources and Liquidity Recent Developments	75 92 93 96
Item 4.  PART II	Controls and Procedures  OTHER INFORMATION	97
Item 1.	Legal Proceedings	98
Item 1A.	Risk Factors	98
Item 2.	Unregistered Sales of Equity Securities and Use of Proceeds	98
Item 6.	Exhibits	99

#### PART I. FINANCIAL INFORMATION

# ITEM 1. FINANCIAL STATEMENTS

# THE ALLSTATE CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(\$ in millions, except per share data)		7	Three Mo Jui		Six Months Ended June 30,				
		2011			2010	-	2011		2010
			(una	udited	l)	_	(una	ıdited)	
Revenues	Φ.		C 455	Φ.	6.540	Φ.	42.005	ф	12.016
Property-liability insurance premiums	\$		6,457 547	\$	6,513 545	\$	12,905	\$	13,016 1,089
Life and annuity premiums and contract charges Net investment income			1,020		1,049		1,116 2,002		2,099
Realized capital gains and losses:			1,020		1,049		2,002		2,099
Total other-than-temporary impairment losses			(82)		(288)		(238)		(538)
Portion of loss recognized in other comprehensive income			(4)		(18)		(31)		(23)
Net other-than-temporary impairment losses recognized in			(4)		(10)	_	(31)		(23)
earnings			(86)		(306)		(269)		(561)
Sales and other realized capital gains and losses			143		(145)		422		(238)
Total realized capital gains and losses			57		(451)	_	153		(799)
Total Tealized Capital gains and losses				_	(431)	_	133		(799)
			8,081		7,656	_	16,176		15,405
Costs and expenses									
Property-liability insurance claims and claims expense			6,355		4,714		10,831		9,506
Life and annuity contract benefits			422		485		876		927
Interest credited to contractholder funds			417		450		835		913
Amortization of deferred policy acquisition costs			1,018		949		2,069		1,963
Operating costs and expenses			802		789		1,640		1,618
Restructuring and related charges			11		13		20		24
Interest expense			91		92		183		184
			9,116		7,492	_	16,454		15,135
Gain (loss) on disposition of operations			6	_	2	_	(17)		3
(Loss) income from operations before income tax (benefit) expense			(1,029)		166		(295)		273
Income tax (benefit) expense			(409)	_	21	_	(194)		8
Net (loss) income	\$		(620)	\$	145	\$_	(101)	\$	265
Earnings per share:									
Net (loss) income per share - Basic	\$		(1.19)	\$	0.27	\$_	(0.19)	\$	0.49
Weighted average shares - Basic			523.1	_	540.7	_	528.2		540.4
						_			

Net (loss) income per share - Diluted	\$_	(1.19)	\$_	0.27	\$ (0.19)	\$	0.49
Weighted average shares - Diluted	_	523.1	=	543.0	528.2	: :	542.4
Cash dividends declared per share	\$_	0.21	\$	0.20	\$ 0.42	\$	0.40

See notes to condensed consolidated financial statements.

1

# THE ALLSTATE CORPORATION AND SUBSIDIARIES

# CONDENSED CONSOLIDATED STATEMENTS OF FINANCIAL POSITION

(\$ in millions, except par value data)		June 30, 2011	_	December 31, 2010
Assets		(unaudited)	· ·	_
Investments				
Fixed income securities, at fair value (amortized cost \$76,502 and \$78,786)	\$	78,414	\$	79,612
Equity securities, at fair value (cost \$4,329 and \$4,228)		4,954		4,811
Mortgage loans		6,827		6,679
Limited partnership interests		4,400		3,816
Short-term, at fair value (amortized cost \$2,536 and \$3,279)		2,536		3,279
Other	_	2,158	_	2,286
Total investments		99,289		100,483
Cash		693		562
Premium installment receivables, net		4,869		4,839
Deferred policy acquisition costs		4,572		4,769
Reinsurance recoverables, net		6,446		6,552
Accrued investment income		875		809
Deferred income taxes		525		784
Property and equipment, net		914		921
Goodwill		874		874
Other assets		1,791		1,605
Separate Accounts		8,175	_	8,676
Total assets	\$	129,023	\$	130,874
Liabilities				
Reserve for property-liability insurance claims and claims expense	\$	20,456	\$	19,468
Reserve for life-contingent contract benefits		13,787		13,482
Contractholder funds		45,078		48,195
Unearned premiums		9,727		9,800
Claim payments outstanding		948		737
Other liabilities and accrued expenses		6,152		5,564
Long-term debt		5,907		5,908
Separate Accounts		8,175	_	8,676
Total liabilities	_	110,230	_	111,830
Commitments and Contingent Liabilities (Note 10) Equity				
Preferred stock, \$1 par value, 25 million shares authorized, none issued				
Common stock, \$.01 par value, 2.0 billion shares authorized and 900 million issued, 517 million and 533 million				
shares outstanding		9		9
Additional capital paid-in		3,165		3,176
Retained income		31,647		31,969
Deferred ESOP expense		(43)		(44)
Treasury stock, at cost (383 million and 367 million shares)		(16,387)		(15,910)
Accumulated other comprehensive income:		, ,		, ,
Unrealized net capital gains and losses:				
Unrealized net capital losses on fixed income securities with OTTI		(156)		(190)
Other unrealized net capital gains and losses		1,783		1,089
Unrealized adjustment to DAC, DSI and insurance reserves		(181)		36
Total unrealized net capital gains and losses	_	1,446	_	935
Unrealized foreign currency translation adjustments		83		69
Unrecognized pension and other postretirement benefit cost		(1,156)		(1,188)
Total accumulated other comprehensive income (loss)	_	373	_	(184)
Total shareholders' equity	_	18,764	_	19,016
Noncontrolling interest		10,704		28
Total equity	_	18,793	_	19,044
• •	<u>e</u>	129,023	s <sup>-</sup>	130,874
Total liabilities and equity	<sup>5</sup> =	129,023	Ψ=	150,074

See notes to condensed consolidated financial statements.

#### THE ALLSTATE CORPORATION AND SUBSIDIARIES

#### CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(\$ in millions)	Six Months Ended					
		June 30, 2011	2010			
Cash flows from operating activities			2010			
Net (loss) income	\$	(unaudited)				
	Ф	(101) \$	265			
Adjustments to reconcile net (loss) income to net cash provided by operating activities:						
Depreciation, amortization and other non-cash items		89	26			
Realized capital gains and losses		(153)	799			
Loss (gain) on disposition of operations		17	(3)			
Interest credited to contractholder funds		835	913			
Changes in:						
Policy benefits and other insurance reserves		665	306			
Unearned premiums		(87)	(135)			
Deferred policy acquisition costs		57	(70)			
Premium installment receivables, net		(22)	9			
Reinsurance recoverables, net		(40)	(206)			
Income taxes		(226)	74			
Other operating assets and liabilities		226	116			
Net cash provided by operating activities		1,260	2,094			
Cash flows from investing activities						
Proceeds from sales						
Fixed income securities		14,140	9,114			
Equity securities		854	3,046			
Limited partnership interests		335	278			
Mortgage loans		65	44			
Other investments		109	62			
Investment collections						
Fixed income securities		2,385	2,391			
Mortgage loans		308	638			
Other investments		92	44			
Investment purchases		02	• •			
Fixed income securities		(13,934)	(11,900)			
Equity securities		(781)	(1,501)			
Limited partnership interests		(765)	(616)			
Mortgage loans		(536)				
Other investments			(10)			
Change in short-term investments, net		(146)	(79)			
Change in other investments, net		1,166	439			
Purchases of property and equipment, net		(170)	(128)			
Disposition of operations		(106)	(69)			
Net cash provided by investing activities		(1)	4.550			
Cash flows from financing activities		3,015	1,753			
		445	40			
Repayment of long-term debt Contractholder fund deposits		(1)	(1)			
Contractholder fund deposits  Contractholder fund withdrawals		1,120	1,567			
		(4,508)	(5,112)			
Dividends paid		(218)	(215)			
Treasury stock purchases		(544)	(5)			
Shares reissued under equity incentive plans, net		17	25			
Excess tax benefits on share-based payment arrangements		(3)	(4)			
Other		(7)	(3)			
Net cash used in financing activities		(4,144)	(3,748)			
Net increase in cash		131	99			
Cash at beginning of period		562	612			
Cash at end of period	\$	693 \$	711			

See notes to condensed consolidated financial statements.

3

# THE ALLSTATE CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

#### 1. General

# **Basis of presentation**

The accompanying condensed consolidated financial statements include the accounts of The Allstate Corporation and its wholly owned subsidiaries, primarily Allstate Insurance Company ("AIC"), a property-liability insurance company with various property-liability and life and investment subsidiaries, including Allstate Life Insurance Company ("ALIC") (collectively referred to as the "Company" or "Allstate").

The condensed consolidated financial statements and notes as of June 30, 2011, and for the three-month and six-month periods ended June 30, 2011 and 2010 are unaudited. The condensed consolidated financial statements reflect all adjustments (consisting only of normal recurring accruals), which are, in the opinion of management, necessary for the fair presentation of the financial position, results of operations and cash flows for the interim periods. These condensed consolidated financial statements and notes should be read in conjunction with the consolidated financial statements and notes thereto included in the Company's Annual Report on Form 10-K for the year ended December 31, 2010. The results of operations for the interim periods should not be considered indicative of results to be expected for the full year.

#### Adopted accounting standards

Consolidation Analysis Considering Investments Held through Separate Accounts

In April 2010, the Financial Accounting Standards Board ("FASB") issued guidance clarifying that an insurer is not required to combine interests in investments held in a qualifying separate account with its interests in the same investments held in the general account when performing a consolidation evaluation. The adoption of this guidance as of January 1, 2011 had no impact on the Company's results of operations or financial position.

Disclosure of Supplementary Pro Forma Information for Business Combinations

In December 2010, the FASB issued disclosure guidance for entities that enter into business combinations that are material. The guidance specifies that if an entity presents comparative financial statements, the entity should disclose pro forma revenue and earnings of the combined entity as though the business combination that occurred during the current year had occurred as of the beginning of the comparable prior annual reporting period only. The guidance expands the supplemental pro forma disclosures to include a description of the nature and amount of material, nonrecurring pro forma adjustments directly attributable to the business combination. The Company will apply the guidance to any business combinations entered into on or after January 1, 2011.

#### Pending accounting standards

Accounting for Costs Associated with Acquiring or Renewing Insurance Contracts

In October 2010, the FASB issued guidance modifying the definition of the types of costs incurred by insurance entities that can be capitalized in the acquisition of new and renewal contracts. The guidance specifies that the costs must be based on successful efforts. The guidance also specifies that advertising costs should be included as deferred acquisition costs only when the direct-response advertising accounting criteria are met. If application of the guidance would result in the capitalization of acquisition costs that had not been capitalized prior to adoption, the entity may elect not to capitalize those additional costs. The new guidance is effective for reporting periods beginning after December 15, 2011 and should be applied prospectively, with retrospective application permitted. The Company is in the process of evaluating the impact of adoption on the Company's results of operations and financial position.

Criteria for Classification as a Troubled Debt Restructuring ("TDR")

In April 2011, the FASB issued clarifying guidance related to determining whether a loan modification or restructuring should be classified as a TDR. The additional guidance provided pertains to the two criteria used to determine whether a TDR exists, specifically whether the creditor has granted a concession and whether the debtor is experiencing financial difficulties. The new guidance is effective for reporting periods beginning on or after June 15, 2011 with early adoption permitted. The guidance related to the identification of a TDR is to be applied retrospectively to the beginning of the annual period of adoption. The measurement of impairment on a TDR identified under this guidance is effective prospectively. Disclosures about the credit quality of financing receivables and the allowance for credit losses previously deferred for TDRs, is also effective for reporting periods

4

beginning on or after June 15, 2011. The Company is in the process of evaluating the impact of adoption, which is not expected to be material to the Company's results of operations and financial position.

Criteria for Determining Effective Control for Repurchase Agreements

In April 2011, the FASB issued guidance modifying the assessment criteria of effective control for repurchase agreements. The new guidance removes the criterion requiring an entity to have the ability to repurchase or redeem financial assets on substantially the agreed terms and the collateral maintenance implementation guidance related to that criterion. The guidance is to be applied prospectively to transactions or modifications of existing transactions that occur during reporting periods beginning on or after December 15, 2011. Early adoption is not permitted. The impact of adoption is not expected to be material to the Company's results of operations and financial position.

Amendments to Fair Value Measurement and Disclosure Requirements

In May 2011, the FASB issued guidance that clarifies the application of existing fair value measurement and disclosure requirements and amends certain fair value measurement principles, requirements and disclosures. To improve consistency in global application, changes in wording were made. The guidance is to be applied prospectively for reporting periods beginning after December 15, 2011. Early adoption is not permitted. The impact of adoption is not expected to be material to the Company's results of operations and financial position.

Presentation of Comprehensive Income

In June 2011, the FASB issued guidance amending the presentation of comprehensive income and its components. Under the new guidance, an entity has the option to present comprehensive income in a single continuous statement or in two separate but consecutive statements. Both options require an entity to present reclassification adjustments for items reclassified from other comprehensive income to net income in the statement(s) where the components of net income and the components of comprehensive income are presented. The guidance is effective for reporting periods beginning after December 15, 2011 and is to be applied retrospectively. Early adoption is permitted. The impact of adoption is related to presentation only and will have no impact on the Company's results of operations and financial position.

#### 2. Earnings per share

Basic earnings per share is computed using the weighted average number of common shares outstanding, including unvested participating restricted stock units. Diluted earnings per share is computed using the weighted average number of common and dilutive potential common shares outstanding. For the Company, dilutive potential common shares consist of outstanding stock options and unvested non-participating restricted stock units.

The computation of basic and diluted earnings per share is presented in the following table.

(\$ in millions, except per share data)		Three mo Jur	onth ne 3		Six months ended June 30,				
		2011		2010		2011		2010	
Numerator:				_					
Net (loss) income	\$ _	(620)	\$	145	\$	(101)	\$	265	
Denominator: Weighted average common shares outstanding Effect of dilutive potential common shares:		523.1		540.7		528.2		540.4	
Stock options				2.1				2.0	
Restricted stock units (non-participating)				0.2					
Weighted average common and dilutive potential common shares outstanding	=	523.1	: :	543.0	=	528.2	_	542.4	
Earnings per share - Basic	\$	(1.19)	\$	0.27	\$	(0.19)	\$	0.49	
Earnings per share - Diluted	\$	(1.19)	\$	0.27	\$	(0.19)	\$	0.49	

As a result of the net loss for the three-month and six-month periods ended June 30, 2011, weighted average dilutive potential common shares outstanding resulting from 2.1 million stock options and 0.5 million restricted stock options (non-participating) in both periods were not included in the computation of diluted earnings per share since inclusion of these securities would have an anti-dilutive effect. In the absence of the net loss, weighted

5

average common and dilutive potential common shares would have totaled 525.7 million and 530.8 million for the three-month and six-month periods ended June 30, 2011, respectively.

The effect of dilutive potential common shares does not include the effect of options with an anti-dilutive effect on earnings per share because their exercise prices exceed the average market price of Allstate common shares during the period or for which the unrecognized compensation cost would have an anti-dilutive effect. Options to purchase 28.3 million and 27.7 million Allstate common shares, with exercise prices ranging from \$27.36 to \$62.84 and \$28.52 to \$62.84, were outstanding for the three-month periods ended June 30, 2011 and 2010, respectively, but were not included in the computation of diluted earnings per share in those periods. Options to purchase 28.4 million and 26.1 million Allstate common shares, with exercise prices ranging from \$27.36 to \$62.84 and \$27.36 to \$64.53, were outstanding for the six-month periods ended June 30, 2011 and 2010, respectively, but were not included in the computation of diluted earnings per share in those periods.

#### 3. Supplemental Cash Flow Information

Non-cash investment exchanges, including modifications of certain mortgage loans (primarily refinances at maturity with no concessions granted to the borrower), fixed income securities, limited partnerships and other investments, as well as mergers completed with equity securities, totaled \$513 million and \$353 million for the six months ended June 30, 2011 and 2010, respectively.

Liabilities for collateral received in conjunction with the Company's securities lending program and over-the-counter ("OTC") derivatives are reported in other liabilities and accrued expenses or other investments. The accompanying cash flows are included in cash flows from operating activities in the Condensed Consolidated Statements of Cash Flows along with the activities resulting from management of the proceeds, which are as follows:

(\$ in millions)	Six mor Ju	nths en ne 30,	ded
	 2011		2010
Net change in proceeds managed			
Net change in short-term investments	\$ (421)	\$	211
Operating cash flow (used) provided	 (421)		211
Net change in cash	(2)		2
Net change in proceeds managed	\$ (423)	\$	213
Net change in liabilities			
Liabilities for collateral, beginning of year	\$ (484)	\$	(658)
Liabilities for collateral, end of period	(907)		(445)
Operating cash flow provided (used)	\$ 423	\$	(213)

# 6

#### 4. Investments

The amortized cost, gross unrealized gains and losses and fair value for fixed income securities are as follows:

(\$ in millions)	Amortized	Gross	Fair		
	cost	 Gains		Losses	value
June 30, 2011					 
U.S. government and agencies	\$ 5,872	\$ 318	\$	(3)	\$ 6,187
Municipal	14,557	491		(375)	14,673
Corporate	40,610	2,014		(255)	42,369
Foreign government	2,720	327		(4)	3,043
Residential mortgage-backed					
securities ("RMBS")	6,356	203		(569)	5,990
Commercial mortgage-backed					
securities ("CMBS")	2,083	57		(154)	1,986
Asset-backed securities ("ABS")	4,281	95		(234)	4,142
Redeemable preferred stock	 23	1			 24
Total fixed income securities	\$ 76,502	\$ 3,506	\$	(1,594)	\$ 78,414
December 31, 2010					
U.S. government and agencies	\$ 8,320	\$ 327	\$	(51)	\$ 8,596
Municipal	16,201	379		(646)	15,934
Corporate	36,260	1,816		(421)	37,655
Foreign government	2,821	347		(10)	3,158
RMBS	8,509	216		(732)	7,993
CMBS	2,213	58		(277)	1,994
ABS	4,425	113		(294)	4,244
Redeemable preferred stock	 37	 1	<u> </u>		 38
Total fixed income securities	\$ 78,786	\$ 3,257	\$	(2,431)	\$ 79,612

# **Scheduled maturities**

The scheduled maturities for fixed income securities are as follows as of June 30, 2011:

(\$ in millions)		Fair	
		cost	value
Due in one year or less	\$	2,921	\$ 2,965
Due after one year through five years		24,089	25,112
Due after five years through ten years		19,237	20,311
Due after ten years		19,618	19,894
	_	65,865	68,282
RMBS and ABS	_	10,637	10,132
Total	\$	76,502	\$ 78,414

Actual maturities may differ from those scheduled as a result of prepayments by the issuers. Because of the potential for prepayment on RMBS and ABS, they are not categorized by contractual maturity. CMBS are categorized by contractual maturity because they generally are not subject to prepayment risk.

7

# Net investment income

Net investment income is as follows:

(\$ in millions)	Three m Ju	onths e ine 30,	ended		nths e ne 30,	s ended 30,		
	 2011		2010	 2011		2010		
Fixed income securities	\$ 899	\$	955	\$ 1,799	\$	1,914		
Equity securities	34		25	53		46		
Mortgage loans	87		99	176		203		
Limited partnership interests	18		7	28		13		
Short-term investments	1		2	3		4		
Other	26		6	37		7		
Investment income, before expense	 1,065		1,094	 2,096		2,187		
Investment expense	(45)		(45)	(94)		(88)		
Net investment income	\$ 1,020	\$	1,049	\$ 2,002	\$	2,099		

#### Realized capital gains and losses

(\$ in millions)

Realized capital gains and losses by asset type are as follows:

		J 1

Three mon	ths ended	Six mont	ths ended
June	2 30,	Jun	e 30,
2011	2010	2011	2010

Fixed income securities	\$ 39	\$ (188)	\$ 12	\$ (324)
Equity securities	15	45	137	59
Mortgage loans	(3)	(28)	(9)	(53)
Limited partnership interests	53	26	121	5
Derivatives	(53)	(308)	(120)	(493)
Other	6	2	12	7
Realized capital gains and losses	\$ 57	\$ (451)	\$ 153	\$ (799)

Realized capital gains and losses by transaction type are as follows:

(\$ in millions)		Three moi Jun	nths ei e 30,	ıded		Six mon Jun	ded	
	_	2011		2010	_	2011		2010
Impairment write-downs	\$	(70)	\$	(239)	\$	(184)	\$	(462)
Change in intent write-downs		(16)		(67)		(85)		(99)
Net other-than-temporary impairment losses recognized in	_				_			
earnings		(86)		(306)		(269)		(561)
Sales		141		145		424		233
Valuation of derivative instruments		(50)		(283)		(28)		(438)
Settlements of derivative instruments		(3)		(27)		(92)		(57)
Equity method of accounting ("EMA") limited partnership								
income		55		20		118		24
Realized capital gains and losses	\$	57	\$	(451)	\$	153	\$	(799)

Gross gains of \$177 million and \$144 million and gross losses of \$98 million and \$113 million were realized on sales of fixed income securities during the three months ended June 30, 2011 and 2010, respectively. Gross gains of \$388 million and \$286 million and gross losses of \$186 million and \$187 million were realized on sales of fixed income securities during the six months ended June 30, 2011 and 2010, respectively.

8

Other-than-temporary impairment losses by asset type are as follows:

(\$ in millions)			e months end	led			Six months ended				
		J	une 30, 2011			_		J	une 30, 2011		
			Included			Included					
	Gross		in OCI		Net		Gross		in OCI		Net
Fixed income securities:				_		_					
Municipal	\$ (15)	\$	(1)	\$	(16)	\$	(42)	\$	(3)	\$	(45)
Corporate							(5)		1		(4)
Foreign government							(1)				(1)
RMBS	(35)				(35)		(107)		(25)		(132)
CMBS	(10)		(3)		(13)		(26)		(7)		(33)
ABS							(7)		3		(4)
Total fixed income securities	(60)	_	(4)	_	(64)	_	(188)		(31)		(219)
Equity securities	(13)				(13)		(33)				(33)
Mortgage loans	(7)				(7)		(13)				(13)
Limited partnership interests	(1)				(1)		(2)				(2)
Other	(1)				(1)		(2)				(2)
Other-than-temporary impairment		_		_		_				_	
losses	\$ (82)	\$ =	(4)	\$	(86)	\$ =	(238)	\$ _	(31)	\$ =	(269)

	Three months ended June 30, 2010						Six months ended June 30, 2010						
			Included										
	Gross		in OCI		Net		Gross	in OCI			Net		
Fixed income securities:	 			_		_		_					
Municipal	\$ (68)	\$	4	\$	(64)	\$	(105)	\$	4	\$	(101)		
Corporate	(6)		(1)		(7)		(53)		2		(51)		
RMBS	(124)		5		(119)		(212)		(2)		(214)		
CMBS	(17)		(11)		(28)		(43)		(11)		(54)		
ABS	(6)		(15)		(21)		(9)		(16)		(25)		
Total fixed income securities	 (221)		(18)		(239)		(422)	_	(23)		(445)		
Equity securities	(31)				(31)		(37)				(37)		
Mortgage loans	(28)				(28)		(47)				(47)		
Limited partnership interests	(8)				(8)		(32)				(32)		
Other-than-temporary impairment	 			_		_		_					
losses	\$ (288)	\$_	(18)	\$_	(306)	\$	(538)	\$	(23)	\$	(561)		

The total amount of other-than-temporary impairment losses included in accumulated other comprehensive income at the time of impairment for fixed income securities, which were not included in earnings, are presented in the following table. The amount excludes \$249 million and \$322 million as of June 30, 2011 and December 31, 2010, respectively, of net unrealized gains related to changes in valuation of the fixed income securities subsequent to the impairment measurement date.

(\$ in millions)		June 30, 2011	December 31, 2010
Municipal	\$	(13)	\$ (27)
Corporate		(32)	(31)
RMBS		(411)	(467)
CMBS		(11)	(49)
ABS		(22)	(41)
Total	\$	(489)	\$ (615)
	9		

Rollforwards of the cumulative credit losses recognized in earnings for fixed income securities held as of the end of the period are as follows:

(\$ in millions)	Three months ended June 30,				Six mo Ju	ıded	
	2011 2010			2011	2010		
Beginning balance	\$ (963)	\$	(1,236)	\$	(1,046)	\$	(1,187)
Additional credit loss for securities previously other-than-temporarily							
impaired	(31)		(101)		(90)		(180)
Additional credit loss for securities not previously other-than-temporarily							
impaired	(17)		(71)		(44)		(172)
Reduction in credit loss for securities disposed or collected	94		95		247		226
Reduction in credit loss for securities the Company has made the decision							
to sell or more likely than not will be required to sell			1		15		1
Change in credit loss due to accretion of increase in cash flows	5		3		6		3
Ending balance	\$ (912)	\$	(1,309)	\$	(912)	\$	(1,309)

The Company uses its best estimate of future cash flows expected to be collected from the fixed income security, discounted at the security's original or current effective rate, as appropriate, to calculate a recovery value and determine whether a credit loss exists. The determination of cash flow estimates is inherently subjective and methodologies may vary depending on facts and circumstances specific to the security. All reasonably available information relevant to the collectability of the security, including past events, current conditions, and reasonable and supportable assumptions and forecasts, are considered when developing the estimate of cash flows expected to be collected. That information generally includes, but is not limited to, the remaining payment terms of the security, prepayment speeds, foreign exchange rates, the financial condition and future earnings potential of the issue or issuer, expected defaults, expected recoveries, the value of underlying collateral, vintage, geographic concentration, available reserves or escrows, current subordination levels, third party guarantees and other credit enhancements. Other information, such as industry analyst reports and forecasts, sector credit ratings, financial condition of the bond insurer for insured fixed income securities, and other market data relevant to the realizability of contractual cash flows, may also be considered. The estimated fair value of collateral will be used to estimate recovery value if the Company determines that the security is dependent on the liquidation of collateral for ultimate settlement. If the estimated recovery value is less than the amortized cost of the security, a credit loss exists and an other-than-temporary impairment for the difference between the estimated recovery value and amortized cost is recorded in earnings. The portion of the unrealized loss related to factors other than credit remains classified in accumulated other comprehensive income. If the Company may conclude that the entire decline in fair v

10

#### Unrealized net capital gains and losses

Unrealized net capital gains and losses included in accumulated other comprehensive income are as follows:

(\$ in millions)	Fair Gross unrealized				alized	<b>Unrealized net</b>		
June 30, 2011	value	Gains Losse			Losses	gains (losses)		
Fixed income securities	\$ 78,414	\$	3,506	\$	(1,594)	\$	1,912	
Equity securities	4,954		705		(80)		625	
Short-term investments	2,536							
Derivative instruments (1)	(31)				(36)		(36)	
EMA limited partnership interests (2)							7	
Unrealized net capital gains and losses, pre-tax							2,508	
Amounts recognized for:								
Insurance reserves (3)							(217)	
DAC and DSI (4)							(61)	
Amounts recognized							(278)	
Deferred income taxes							(784)	
Unrealized net capital gains and losses, after-tax						\$	1,446	

<sup>(1)</sup> Included in the fair value of derivative instruments are \$(5) million classified as assets and \$26 million classified as liabilities.

Unrealized net capital gains and losses for limited partnership interests represent the Company's share of EMA limited partnerships' other comprehensive income. Fair value and gross gains

and losses are not applicable.

The insurance reserves adjustment represents the amount by which the reserve balance would increase if the net unrealized gains in the applicable product portfolios were realized and reinvested at current lower interest rates, resulting in a premium deficiency. Although the Company evaluates premium deficiencies on the combined performance of life insurance and immediate annuities with life contingencies, the adjustment primarily relates to structured settlement annuities with life contingencies, in addition to annuity buy-outs and certain payout annuities with life contingencies.

The DAC and DSI adjustment balance represents the amount by which the amortization of DAC and DSI would increase or decrease if the unrealized gains or losses in the respective product

	Fair	Gross unrealized			Unrealized net			
December 31, 2010	value		Gains		Losses	ga	ins (loss	ses)
Fixed income securities	\$ 79,612	\$	3,257	\$	(2,431)		\$	826
Equity securities	4,811		646		(63)			583
Short-term investments	3,279							
Derivative instruments (1)	(17)		2		(24)			(22)
Unrealized net capital gains and losses, pre-tax								1,387
Amounts recognized for:								
Insurance reserves								(41)
DAC and DSI								97
Amounts recognized								56
Deferred income taxes								(508)
Unrealized net capital gains and losses, after-tax							\$	935

<sup>(1)</sup> Included in the fair value of derivative instruments are \$2 million classified as assets and \$19 million classified as liabilities.

11

#### Change in unrealized net capital gains and losses

The change in unrealized net capital gains and losses for the six months ended June 30, 2011 is as follows:

(\$ in millions)		
Fixed income securities	\$	1,086
Equity securities		42
Derivative instruments		(14)
EMA limited partnership interests		7
Total	_	1,121
Amounts recognized for:		
Insurance reserves		(176)
DAC and DSI		(158)
Amounts recognized	_	(334)
Deferred income taxes		(276)
Increase in unrealized net capital gains and losses	\$	511

#### Portfolio monitoring

The Company has a comprehensive portfolio monitoring process to identify and evaluate each fixed income and equity security whose carrying value may be other-than-temporarily impaired.

For each fixed income security in an unrealized loss position, the Company assesses whether management with the appropriate authority has made the decision to sell or whether it is more likely than not the Company will be required to sell the security before recovery of the amortized cost basis for reasons such as liquidity, contractual or regulatory purposes. If a security meets either of these criteria, the security's decline in fair value is considered other than temporary and is recorded in earnings.

If the Company has not made the decision to sell the fixed income security and it is not more likely than not the Company will be required to sell the fixed income security before recovery of its amortized cost basis, the Company evaluates whether it expects to receive cash flows sufficient to recover the entire amortized cost basis of the security. The Company calculates the estimated recovery value by discounting the best estimate of future cash flows at the security's original or current effective rate, as appropriate, and compares this to the amortized cost of the security. If the Company does not expect to receive cash flows sufficient to recover the entire amortized cost basis of the fixed income security, the credit loss component of the impairment is recorded in earnings, with the remaining amount of the unrealized loss related to other factors recognized in other comprehensive income.

For equity securities, the Company considers various factors, including whether it has the intent and ability to hold the equity security for a period of time sufficient to recover its cost basis. Where the Company lacks the intent and ability to hold to recovery, or believes the recovery period is extended, the equity security's decline in fair value is considered other than temporary and is recorded in earnings. For equity securities managed by a third party, the Company has contractually retained its decision making authority as it pertains to selling equity securities that are in an unrealized loss position.

The Company's portfolio monitoring process includes a quarterly review of all securities to identify instances where the fair value of a security compared to its amortized cost (for fixed income securities) or cost (for equity securities) is below established thresholds. The process also includes the monitoring of other impairment indicators such as ratings, ratings downgrades and payment defaults. The securities identified, in addition to other securities for which the Company may have a concern, are evaluated for potential other-than-temporary impairment using all reasonably available information relevant to the collectability or recovery of the security. Inherent in the Company's evaluation of other-than-temporary impairment for these fixed income and equity securities are assumptions and estimates about the financial condition and future earnings potential of the issue or issuer. Some of the factors considered in evaluating whether a decline in fair value is other than temporary are: 1) the financial condition, near-term and long-term prospects of the issue or issuer, including relevant industry specific market conditions and trends, geographic location and implications of rating agency actions and offering prices; 2) the specific reasons that a security is in an unrealized loss position, including overall market conditions which could affect liquidity; and 3) the length of time and extent to which the fair value has been less than amortized cost or cost.

The following table summarizes the gross unrealized losses and fair value of fixed income and equity securities by the length of time that individual securities have been in a continuous unrealized loss position.

(\$ in millions)		Le	ess than 12 mon	ths				Total				
	Number		Fair		Unrealized	Number		Fair		Unrealized		unrealized
I 20 2011	of issues	_	value	-	losses	of issues	_	value		losses		losses
June 30, 2011												
Fixed income securities	0	\$	198	\$	(2)		\$		ď		ď	(2)
U.S. government and agencies	9 417	Ф	1,985	Ф	(3)	 3F7	Ф	2 222	Э	(200)	Ф	(3)
Municipal	417 385				(75)	357		2,222		(300)		(375)
Corporate			4,980		(95)	113		1,552		(160)		(255)
Foreign government	25		114		(4)	200		1 220		(557)		(4)
RMBS	120		368		(12)	296		1,338		(557)		(569)
CMBS	29		314		(16)	75		609		(138)		(154)
ABS	30	-	441	-	(5)	121	-	1,235		(229)		(234)
Total fixed income securities	1,015		8,400		(210)	962		6,956		(1,384)		(1,594)
Equity securities	895		666		(60)	87		73		(20)		(80)
Total fixed income and equity securities	1,910	\$	9,066	\$	(270)	1,049	= \$	7,029	\$	(1,404)	\$	(1,674)
Investment grade fixed income securities	836	\$	6,786	\$	(156)	618	\$	4,665	\$	(607)	\$	(763)
Below investment grade fixed income securities	179		1,614		(54)	344		2,291		(777)		(831)
Total fixed income securities	1,015	\$	8,400	\$	(210)	962	\$	6,956	\$	(1,384)	\$	(1,594)
December 31, 2010												
Fixed income securities												
U.S. government and agencies	32	\$	2,081	\$	(51)		\$		\$		\$	(51)
Municipal	847	-	4,130	-	(175)	411	-	2,715	-	(471)	-	(646)
Corporate	438		5,994		(186)	150		1,992		(235)		(421)
Foreign government	33		277		(9)	1		10		(1)		(10)
RMBS	280		583		(12)	422		1,939		(720)		(732)
CMBS	14		158		(3)	114		835		(274)		(277)
ABS	68		762		(8)	133		1,313		(286)		(294)
Total fixed income securities	1,712	-	13,985	-	(444)	1,231	_	8,804		(1,987)		(2,431)
Equity securities	773		610		(48)	44		91		(15)		(63)
Total fixed income and equity securities	2,485	\$	14,595	\$	(492)	1,275	- \$	8,895	\$	(2,002)	\$	(2,494)
Total fixed meome and equity securities	2,403	= "	14,000	: "	(432)	1,270	= "	5,655		(2,002)	4	(2,434)
Investment grade fixed income securities	1,607	\$	13,280	\$	(408)	857	\$	6,217	\$	(943)	\$	(1,351)
Below investment grade fixed income securities	105		705		(36)	374		2,587		(1,044)		(1,080)
Total fixed income securities	1,712	\$	13,985	\$	(444)	1,231	\$	8,804	\$	(1,987)	\$	(2,431)

As of June 30, 2011, \$750 million of unrealized losses are related to securities with an unrealized loss position less than 20% of amortized cost or cost, the degree of which suggests that these securities do not pose a high risk of being other-than-temporarily impaired. Of the \$750 million, \$532 million are related to unrealized losses on investment grade fixed income securities. Investment grade is defined as a security having a rating of Aaa, Aa, A or Baa from Moody's, a rating of AAA, AA, A or BBB from Standard & Poor's ("S&P"), Fitch, Dominion or Realpoint, a rating of aaa, aa, a or bbb from A.M. Best, or a comparable internal rating if an externally provided rating is not available. Unrealized losses on investment grade securities are principally related to widening credit spreads or rising interest rates since the time of initial purchase.

As of June 30, 2011, the remaining \$924 million of unrealized losses are related to securities in unrealized loss positions greater than or equal to 20% of amortized cost or cost. Investment grade fixed income securities comprising \$231 million of these unrealized losses were evaluated based on factors such as expected cash flows and the financial condition and near-term and long-term prospects of the issue or issuer and were determined to have adequate resources to fulfill contractual obligations. Of the \$924 million, \$660 million are related to below investment grade fixed income securities and \$33 million are related to equity securities. Of these amounts, \$542 million of the below investment grade fixed income securities had been in an unrealized loss position greater than or equal to 20% of amortized cost for a period of twelve or more consecutive months as of June 30, 2011. Unrealized losses on below investment grade securities are principally related to RMBS, CMBS and ABS and were the result of

13

wider credit spreads resulting from higher risk premiums since the time of initial purchase, largely due to macroeconomic conditions and credit market deterioration, including the impact of lower real estate valuations.

RMBS, CMBS and ABS in an unrealized loss position were evaluated based on actual and projected collateral losses relative to the securities' positions in the respective securitization trusts, security specific expectations of cash flows, and credit ratings. This evaluation also takes into consideration credit enhancement, measured in terms of (i) subordination from other classes of securities in the trust that are contractually obligated to absorb losses before the class of security the Company owns, (ii) the expected impact of other structural features embedded in the securitization trust beneficial to the class of securities the Company owns, such as overcollateralization and excess spread, and (iii) for RMBS and ABS in an unrealized loss position, credit enhancements from reliable bond insurers, where applicable. Municipal bonds in an unrealized loss position were evaluated based on the quality of the underlying securities, taking into consideration credit enhancements from reliable bond insurers, where applicable. Unrealized losses on equity securities are primarily related to equity market fluctuations.

As of June 30, 2011, the Company has not made the decision to sell and it is not more likely than not the Company will be required to sell fixed income securities with unrealized losses before recovery of the amortized cost basis. As of June 30, 2011, the Company had the intent and ability to hold equity securities with unrealized losses for a period of time sufficient for them to recover.

#### Limited partnerships

As of June 30, 2011 and December 31, 2010, the carrying value of equity method limited partnership interests totaled \$2.99 billion and \$2.47 billion, respectively. The Company recognizes an impairment loss for equity method investments when evidence demonstrates that the loss is other than temporary. Evidence of a loss in value that is other than temporary may include the absence of an ability to recover the carrying amount of the investment or the inability of the investee to sustain a level of earnings that would justify the carrying amount of the investment. The Company had no write-downs related to equity method limited partnership interests for the three months and six months ended June 30, 2011 and \$1 million for the three months and six months ended June 30, 2010.

As of June 30, 2011 and December 31, 2010, the carrying value for cost method limited partnership interests was \$1.41 billion and \$1.35 billion, respectively. To determine if an other-than-temporary impairment has occurred, the Company evaluates whether an impairment indicator has occurred in the period that may have a significant adverse effect on the carrying value of the investment. Impairment indicators may include: significantly reduced valuations of the investments held by the limited partnerships; actual recent cash flows received being significantly less than expected cash flows; reduced valuations based on financing completed at a lower value; completed sale of a material underlying investment at a price significantly lower than expected; or any other adverse events since the last financial statements received that might affect the fair value of the investee's capital. Additionally, the Company's portfolio monitoring process includes a quarterly review of all cost method limited partnerships to identify instances where the net asset value is below established thresholds for certain periods of time, as well as investments that are performing below expectations, for further impairment consideration. If a cost method limited partnership is other-than-temporarily impaired, the carrying value is written down to fair value, generally estimated to be equivalent to the reported net asset value of the underlying funds. The Company had write-downs related to cost method investments of \$1 million and \$7 million for the three months ended June 30, 2011 and 2010, respectively, and \$2 million and \$31 million for the six months ended June 30, 2011 and 2010, respectively.

#### Mortgage loans

Mortgage loans are evaluated for impairment on a specific loan basis through a quarterly credit monitoring process and review of key credit quality indicators. Mortgage loans are considered impaired when it is probable that the Company will not collect the contractual principal and interest. Valuation allowances are established for impaired loans to reduce the carrying value to the fair value of the collateral less costs to sell or the present value of the loan's expected future repayment cash flows discounted at the loan's original effective interest rate. Impaired mortgage loans may not have a valuation allowance when the fair value of the collateral less costs to sell is higher than the carrying value. Mortgage loan valuation allowances are charged off when there is no reasonable expectation of recovery. The impairment evaluation is non-statistical in respect to the aggregate portfolio but considers facts and circumstances attributable to each loan. It is not considered probable that additional impairment losses, beyond those identified on a specific loan basis, have been incurred as of June 30, 2011.

14

Accrual of income is suspended for mortgage loans that are in default or when full and timely collection of principal and interest payments is not probable. Cash receipts on mortgage loans on nonaccrual status are generally recorded as a reduction of carrying value.

Debt service coverage ratio is considered a key credit quality indicator when mortgage loans are evaluated for impairment. Debt service coverage ratio represents the amount of estimated cash flows from the property available to the borrower to meet principal and interest payment obligations. Debt service coverage ratio estimates are updated annually or more frequently if conditions are warranted based on the Company's credit monitoring process. The following table reflects the carrying value of non-impaired fixed rate and variable rate mortgage loans summarized by debt service coverage ratio distribution:

(\$ in millions)		June 30, 2011		December 31, 2010						
	_	Fixed rate mortgage	Variable rate mortgage		 Fixed rate mortgage		Variable rate mortgage			
Debt service coverage ratio distribution		loans	loans	Total	loans		loans		Total	
Below 1.0	\$	327	\$ 	\$ 327	\$ 280	\$		\$	280	
1.0 - 1.25		1,622		1,622	1,583		16		1,599	
1.26 - 1.50		1,672	26	1,698	1,520		5		1,525	
Above 1.50		2,740	266	3,006	2,540		546		3,086	
Total non-impaired mortgage loans	\$	6,361	\$ 292	\$ 6,653	\$ 5,923	\$	567	\$	6,490	

Mortgage loans with a debt service coverage ratio below 1.0 that are not considered impaired primarily relate to instances where the borrower has the financial capacity to fund the revenue shortfalls from the properties for the foreseeable term, the decrease in cash flows from the properties is considered temporary, or there are other risk mitigating circumstances such as additional collateral, escrow balances or borrower guarantees.

The net carrying value of impaired mortgage loans is as follows:

(\$ in millions)	June 30, 2011	December 31, 2010
Impaired mortgage loans with a valuation allowance	\$ 153	\$ 168
Impaired mortgage loans without a valuation allowance	 21	21
Total impaired mortgage loans	\$ 174	\$ 189
Valuation allowance on impaired mortgage loans	\$ 68	\$ 84

The average balance of impaired loans was \$178 million during the six months ended June 30, 2011.

The rollforward of the valuation allowance on impaired mortgage loans is as follows:

(\$ in millions)	Tl	ree months ended June 30, 2011	Six months ended June 30, 2011
Beginning balance	\$	77	\$ 84
Net increase in valuation allowance		7	13
Charge offs		(16)	(29)
Ending balance	\$	68	\$ 68

The carrying value of past due mortgage loans is as follows:

(\$ in millions)		June 30, 2011	_	December 31, 2010
Less than 90 days past due	\$	27	\$	12
90 days or greater past due	_	48	_	78

Total past due
Current loans

Total mortgage loans

\$\frac{6}{6}\$

75	90
6,752	6,589
\$ 6,827 \$	6,679

15

#### 5. Fair Value of Assets and Liabilities

Fair value is defined as the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date. The hierarchy for inputs used in determining fair value maximizes the use of observable inputs and minimizes the use of unobservable inputs by requiring that observable inputs be used when available. Assets and liabilities recorded on the Condensed Consolidated Statements of Financial Position at fair value are categorized in the fair value hierarchy based on the observability of inputs to the valuation techniques as follows:

- Level 1: Assets and liabilities whose values are based on unadjusted quoted prices for identical assets or liabilities in an active market that the Company can access.
- Level 2: Assets and liabilities whose values are based on the following:
  - (a) Quoted prices for similar assets or liabilities in active markets;
  - (b) Quoted prices for identical or similar assets or liabilities in markets that are not active; or
  - (c) Valuation models whose inputs are observable, directly or indirectly, for substantially the full term of the asset or liability.
- Level 3: Assets and liabilities whose values are based on prices or valuation techniques that require inputs that are both unobservable and significant to the overall fair value measurement. Unobservable inputs reflect the Company's estimates of the assumptions that market participants would use in valuing the assets and liabilities.

The availability of observable inputs varies by instrument. In situations where fair value is based on internally developed pricing models or inputs that are unobservable in the market, the determination of fair value requires more judgment. The degree of judgment exercised by the Company in determining fair value is typically greatest for instruments categorized in Level 3. In many instances, valuation inputs used to measure fair value fall into different levels of the fair value hierarchy. The category level in the fair value hierarchy is determined based on the lowest level input that is significant to the fair value measurement in its entirety. The Company uses prices and inputs that are current as of the measurement date, including during periods of market disruption. In periods of market disruption, the ability to observe prices and inputs may be reduced for many instruments.

The Company has two types of situations where investments are classified as Level 3 in the fair value hierarchy. The first is where quotes continue to be received from independent third-party valuation service providers and all significant inputs are market observable; however, there has been a significant decrease in the volume and level of activity for the asset when compared to normal market activity such that the degree of market observability has declined to a point where categorization as a Level 3 measurement is considered appropriate. The indicators considered in determining whether a significant decrease in the volume and level of activity for a specific asset has occurred include the level of new issuances in the primary market, trading volume in the secondary market, the level of credit spreads over historical levels, applicable bid-ask spreads, and price consensus among market participants and other pricing sources.

The second situation where the Company classifies securities in Level 3 is where specific inputs significant to the fair value estimation models are not market observable. This occurs in two primary instances. The first relates to the Company's use of broker quotes. The second relates to auction rate securities ("ARS") backed by student loans for which a key input, the anticipated date liquidity will return to this market, is not market observable.

Certain assets are not carried at fair value on a recurring basis, including investments such as mortgage loans, limited partnership interests, bank loans and policy loans. Accordingly, such investments are only included in the fair value hierarchy disclosure when the investment is subject to remeasurement at fair value after initial recognition and the resulting remeasurement is reflected in the condensed consolidated financial statements. In addition, derivatives embedded in fixed income securities are not disclosed in the hierarchy as free-standing derivatives since they are presented with the host contracts in fixed income securities.

In determining fair value, the Company principally uses the market approach which generally utilizes market transaction data for the same or similar instruments. To a lesser extent, the Company uses the income approach which involves determining fair values from discounted cash flow methodologies. For the majority of Level 2 and Level 3 valuations, a combination of the market and income approaches is used.

16

Summary of significant valuation techniques for assets and liabilities measured at fair value on a recurring basis

#### Level 1 measurements

- · <u>Fixed income securities:</u> Comprise U.S. Treasuries. Valuation is based on unadjusted quoted prices for identical assets in active markets that the Company can access.
- · <u>Equity securities:</u> Comprise actively traded, exchange-listed U.S. and international equity securities. Valuation is based on unadjusted quoted prices for identical assets in active markets that the Company can access.
- · Short-term: Comprise actively traded money market funds that have daily quoted net asset values for identical assets that the Company can access.
- <u>Separate account assets:</u> Comprise actively traded mutual funds that have daily quoted net asset values for identical assets that the Company can access. Net asset values for the actively traded mutual funds in which the separate account assets are invested are obtained daily from the fund managers.

# Level 2 measurements

· Fixed income securities:

*U.S. government and agencies*: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields and credit spreads.

*Municipal:* The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields and credit spreads.

Corporate, including privately placed: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields and credit spreads. Also included are privately placed securities valued using a discounted cash flow model that is widely accepted in the financial services industry and uses market observable inputs and inputs derived principally from, or corroborated by, observable market data. The primary inputs to the discounted cash flow model include an interest rate yield curve, as well as published credit spreads for similar assets in markets that are not active that incorporate the credit quality and industry sector of the issuer.

*Foreign government:* The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields and credit spreads.

RMBS - U.S. government sponsored entities ("U.S. Agency"), Prime residential mortgage-backed securities ("Prime") and Alt-A residential mortgage-backed securities ("Alt-A"); ABS - other: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields, prepayment speeds, collateral performance and credit spreads.

*CMBS*: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields, collateral performance and credit spreads.

*Redeemable preferred stock:* The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields, underlying stock prices and credit spreads.

- · <u>Equity securities:</u> The primary inputs to the valuation include quoted prices or quoted net asset values for identical or similar assets in markets that are not active.
- · <u>Short-term:</u> The primary inputs to the valuation include quoted prices for identical or similar assets in markets that are not active, contractual cash flows, benchmark yields and credit spreads. For certain short-term investments, amortized cost is used as the best estimate of fair value.
- · Other investments: Free-standing exchange listed derivatives that are not actively traded are valued based on quoted prices for identical instruments in markets that are not active.

OTC derivatives, including interest rate swaps, foreign currency swaps, foreign exchange forward contracts, and certain credit default swaps, are valued using models that rely on inputs such as interest rate

17

yield curves, currency rates, and counterparty credit spreads that are observable for substantially the full term of the contract. The valuation techniques underlying the models are widely accepted in the financial services industry and do not involve significant judgment.

#### Level 3 measurements

#### Fixed income securities:

*Municipal:* ARS primarily backed by student loans that have become illiquid due to failures in the auction market are valued using a discounted cash flow model that is widely accepted in the financial services industry and uses significant non-market observable inputs, including estimates of future coupon rates if auction failures continue, the anticipated date liquidity will return to the market and illiquidity premium. Also included are municipal bonds that are not rated by third party credit rating agencies but are rated by the National Association of Insurance Commissioners ("NAIC"), and other high-yield municipal bonds. The primary inputs to the valuation of these municipal bonds include quoted prices for identical or similar assets in markets that exhibit less liquidity relative to those markets supporting Level 2 fair value measurements, contractual cash flows, benchmark yields and credit spreads.

Corporate, including privately placed: Primarily valued based on non-binding broker quotes. Also included are equity-indexed notes which are valued using a discounted cash flow model that is widely accepted in the financial services industry and uses significant non-market observable inputs, such as volatility. Other inputs include an interest rate yield curve, as well as published credit spreads for similar assets that incorporate the credit quality and industry sector of the issuer.

RMBS - Subprime residential mortgage-backed securities ("Subprime"), Prime and Alt-A: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that exhibit less liquidity relative to those markets supporting Level 2 fair value measurements, contractual cash flows, benchmark yields, prepayment speeds, collateral performance and credit spreads. Also included are Subprime, Prime and Alt-A securities that are valued based on non-binding broker quotes. Due to the reduced availability of actual market prices or relevant observable inputs as a result of the decrease in liquidity that has been experienced in the market for these securities, Subprime and certain Alt-A securities are categorized as Level 3.

CMBS: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that exhibit less liquidity relative to those markets supporting Level 2 fair value measurements, contractual cash flows, benchmark yields, collateral performance and credit spreads. Also included are CMBS that are valued based on non-binding broker quotes. Due to the reduced availability of actual market prices or relevant observable inputs as a result of the decrease in liquidity that has been experienced in the market for these securities, certain CMBS are categorized as Level 3.

ABS - Collateralized debt obligations ("CDO"): Valued based on non-binding broker quotes received from brokers who are familiar with the investments. Due to the reduced availability of actual market prices or relevant observable inputs as a result of the decrease in liquidity that has been experienced in the market for these securities, all CDO are categorized as Level 3.

ABS - other: The primary inputs to the valuation include quoted prices for identical or similar assets in markets that exhibit less liquidity relative to those markets supporting Level 2 fair value measurements, contractual cash flows, benchmark yields, prepayment speeds, collateral performance and credit spreads. Also included are ABS that are valued based on non-binding broker quotes. Due to the reduced availability of actual market prices or relevant observable inputs as a result of the decrease in liquidity that has been experienced in the market for these securities, certain ABS are categorized as Level 3.

- Other investments: Certain OTC derivatives, such as interest rate caps and floors, certain credit default swaps and OTC options (including swaptions), are valued using models that are widely accepted in the financial services industry. These are categorized as Level 3 as a result of the significance of non-market observable inputs such as volatility. Other primary inputs include interest rate yield curves and credit spreads.
- <u>Contractholder funds</u>: Derivatives embedded in certain life and annuity contracts are valued internally using models widely accepted in the financial services industry that determine a single best estimate of fair value for the embedded derivatives within a block of contractholder liabilities. The models primarily use

stochastically determined cash flows based on the contractual elements of embedded derivatives, projected option cost and applicable market data, such as interest rate yield curves and equity index volatility assumptions. These are categorized as Level 3 as a result of the significance of non-market observable inputs.

Assets and liabilities measured at fair value on a non-recurring basis

(\$ in millions)

Mortgage loans written-down to fair value in connection with recognizing impairments are valued based on the fair value of the underlying collateral less costs to sell. Limited partnership interests written-down to fair value in connection with recognizing other-than-temporary impairments are valued using net asset values.

The following table summarizes the Company's assets and liabilities measured at fair value on a recurring and non-recurring basis as of June 30, 2011:

Quoted prices

(\$ in millions)	_	Quoted prices in active markets for identical assets (Level 1)	Significant other observable inputs (Level 2)		Significant unobservable inputs (Level 3)		Counterparty and cash collateral netting			Balance as of June 30, 2011
Assets Fixed income securities:										
U.S. government and agencies	\$	4,298	\$	1,889	\$				\$	6,187
Municipal	Ψ	4,230	Ψ	13,119	Ψ	1,554			Ψ	14,673
Corporate				40,649		1,720				42,369
Foreign government				3,043		-,· =• 				3,043
RMBS				4,796		1,194				5,990
CMBS				1,048		938				1,986
ABS				1,975		2,167				4,142
Redeemable preferred stock	_			23	_	1				24
Total fixed income securities		4,298		66,542		7,574				78,414
Equity securities		4,248		664		42				4,954
Short-term investments		271		2,265						2,536
Other investments:				425		22	\$	(00)		348
Free-standing derivatives Separate account assets		8,175		425		22	Ф	(99)		8,175
Other assets		0,173				1				3
Total recurring basis assets	_	16,994	_	69,896	-	7,639	_	(99)	_	94,430
Non-recurring basis (1)		10,334		05,050		62		(33)		62
Total assets at fair value	¢	16,994	\$	69,896	\$	7,701	s -	(99)	\$	94,492
% of total assets at fair value	Ψ=	18.0 %	<b>=</b>	74.0 %	Ψ=	8.1 %	~ =	(0.1) %	_	100.0 %
		10.0 /0		74.0 70		0.1 /0		(0.1) /0		100.0 /0
Liabilities										
Contractholder funds:	\$		\$		\$	(629)			\$	(629)
Derivatives embedded in life and annuity contracts Other liabilities:	Ф		Ф	<b></b>	Ф	(029)			Ф	(029)
Free-standing derivatives				(217)		(78)	\$	95		(200)
Total liabilities at fair value	s -		<u>e</u> —	(217)	¢ -	(707)	φ_	95	<u> </u>	(829)
% of total liabilities at fair value	Φ=	%	Ψ=		Ψ=		Ψ=		Ψ	
70 OI total habilates at rail value		%		26.2 %		85.3 %		(11.5) %		100.0 %

<sup>(1)</sup> Includes \$50 million of mortgage loans, \$1 million of limited partnership interests and \$11 million of other investments written-down to fair value in connection with recognizing other-than-temporary impairments.

19

The following table summarizes the Company's assets and liabilities measured at fair value on a recurring and non-recurring basis as of December 31, 2010:

(\$ in millions)  Assets	_	Quoted prices in active markets for identical assets (Level 1)	_	Significant other observable inputs (Level 2)	_	Significant unobservable inputs (Level 3)	_	Counterparty and cash collateral netting	_	Balance as of December 31, 2010
Fixed income securities:										
U.S. government and agencies	\$	4,976	\$	3,620	\$				\$	8,596
Municipal		·		13,918		2,016				15,934
Corporate				35,747		1,908				37,655
Foreign government				3,158						3,158
RMBS				6,199		1,794				7,993
CMBS				1,071		923				1,994
ABS				1,827		2,417				4,244
Redeemable preferred stock				37	_	11				38
Total fixed income securities		4,976		65,577		9,059				79,612
Equity securities		4,316		432		63				4,811
Short-term investments		174		3,105						3,279
Other investments:										
Free-standing derivatives				651		74	\$	(286)		439
Separate account assets		8,676								8,676
Other assets			_		_	1	_		_	1_
Total recurring basis assets		18,142		69,765		9,197		(286)		96,818
Non-recurring basis (1)					_	120				120
Total assets at fair value	\$	18,142	\$	69,765	\$	9,317	\$	(286)	\$	96,938
% of total assets at fair value		18.7 %	-	72.0 %	_	9.6 %	_	(0.3) %	_	100.0 %
Liabilities										
Contractholder funds:										
Derivatives embedded in life and annuity contracts	\$		\$		\$	(653)			\$	(653)
Other liabilities:	•		•		-	()			•	()
Free-standing derivatives		(2)		(529)		(95)	\$	263		(363)
Total liabilities at fair value	\$	(2)	\$	(529)	\$	(748)	\$	263	\$	(1,016)
% of total liabilities at fair value	<b>~</b> =	0.2 %	-	52.1 %		73.6 %	-	(25.9) %	-	100.0 %

<sup>(1)</sup> Includes \$111 million of mortgage loans and \$9 million of limited partnership interests written-down to fair value in connection with recognizing other-than-temporary impairments.

The following table presents the rollforward of Level 3 assets and liabilities held at fair value on a recurring basis during the three months ended June 30, 2011.

Total realized and

(13)

(13)

48

48

Settlements

2011

Issuances

(629)

(629)

unrealized gains (losses) included in: OCI on Statement Transfers Transfers Balance as of Net of Financial into out of Position Assets Fixed income securities: Municipal 1,864 (13)45 (22)Corporate RMBS 2,035 1,398 23 (26) 87 (117) (68) 8 CMBS 995 (21)10 (10)2,091 ABS 11 12 (9)Redeemable preferred stock Total fixed income securities 8,384 (26) 70 97 (226) Equity securities 43 Other investments: Free-standing derivatives, net (71) (3) Other assets Total recurring Level 3 assets (29) Liabilities Contractholder funds: Derivatives embedded in life and annuity contracts (630)**Total recurring Level 3 liabilities** (630) (34) Balance as of June 30, 2011 Settlements Assets Fixed income securities: Municipal (321)\$ (2) 1,554 Corporate RMBS (4) (51) 1,720 1,194 35 (347) (60)2 CMBS (41) 938 2,167 ABS 213 (49)(102)Redeemable preferred stock Total fixed income securities (818) (160)7,574 Equity securities (1) 42 Other investments: Free-standing derivatives, net (56) <sup>(2)</sup> 19 (1) Total recurring Level 3 assets (819)(161)7,561 Liabilities Contractholder funds:

Derivatives embedded in life and annuity contracts

**Total recurring Level 3 liabilities** 

(\$ in millions)

21

The following table presents the rollforward of Level 3 assets and liabilities held at fair value on a recurring basis during the six months ended June 30, 2011.

(\$ in millions)				Total re unrealized incl						
	Balance as of December 31, 2010			Net income (1)		OCI on Statement of Financial Position	Transfers into Level 3			Transfers out of Level 3
Assets										
Fixed income securities:				(B.1)			_			(=0)
Municipal	\$	2,016	\$	(24) 35	\$	66	\$	100	\$	(59)
Corporate RMBS		1,908 1,794		(87)		18 106		182		(164) (113)
CMBS		923		(42)		118		66		(69)
ABS		2,417		`55 <sup>´</sup>		28				(313)
Redeemable preferred stock		1								<u> </u>
Total fixed income securities		9,059		(63)		336		248		(718)
Equity securities		63		(10)						(10)
Other investments:		(21)		(24)						
Free-standing derivatives, net Other assets		(21)		(34)						
Total recurring Level 3 assets	\$	9,102	\$	(107)	\$	336	\$	248	\$	(728)
Liabilities										
Contractholder funds:	_		_		_		_		_	
Derivatives embedded in life and annuity contracts	\$	(653)	\$	(26)	\$		\$		\$_	
Total recurring Level 3 liabilities	\$	(653)	\$	(26)	\$_		\$		\$_	
										Balance as of June 30,

Purchases

Sales

Assets
Fixed income securities:

<sup>(1)</sup> The effect to net income totals \$(63) million and is reported in the Condensed Consolidated Statements of Operations as follows: \$(38) million in realized capital gains and losses, \$9 million in net investment income, \$(26) million in interest credited to contractholder funds and \$(8) million in life and annuity contract benefits.

<sup>(2)</sup> Comprises \$22 million of assets and \$78 million of liabilities.

Municipal	\$ 13	\$ (455)	\$ 	\$ (3)	\$ 1,554
Corporate	131	(378)		(12)	1,720
RMBS		(378)		(128)	1,194
CMBS	10	(66)		(2)	938
ABS	303	(163)		(160)	2,167
Redeemable preferred stock				·	1
Total fixed income securities	 457	(1,440)		(305)	7,574
Equity securities		(1)		`	42
Other investments:					
Free-standing derivatives, net	67			(68)	(56) (2)
Other assets				`	1
Total recurring Level 3 assets	\$ 524	\$ (1,441)	\$ 	\$ (373)	\$ 7,561
Liabilities					
Contractholder funds:					
Derivatives embedded in life and annuity contracts	\$ 	\$ 	\$ (27)	\$ 77	\$ (629)
Total recurring Level 3 liabilities	\$ 	\$ 	\$ (27)	\$ 77	\$ (629)

<sup>(1)</sup> The effect to net income totals \$(133) million and is reported in the Condensed Consolidated Statements of Operations as follows: \$(123) million in realized capital gains and losses, \$16 million in net investment income, \$(63) million in interest credited to contractholder funds and \$37 million in life and annuity contract benefits.

(\$ in millions)

22

The following table presents the rollforward of Level 3 assets and liabilities held at fair value on a recurring basis during the three months ended June 30, 2010.

Total realized and

				inch	ı ganı uded									
Assets	_	Balance as of March 31, 2010	_	Net income (1)	_	OCI on Statement of Financial Position		Purchases, sales, issuances and settlements, net	_	Transfers into Level 3	_	Transfers out of Level 3	. <u>-</u>	Balance as of June 30, 2010
Fixed income securities:	_		_		_		_		_		_			
Municipal	\$	2,482	\$	(31)	\$	21	\$	(236)	\$	16	\$	(55)	\$	2,197
Corporate RMBS		2,177 2,079		(10) (111)		26 180		45 (124)		151		(164) (14)		2,225 2,010
CMBS		1,130		(73)		192		(165)				(204)		2,010 880
ABS		2,403		3		6		141				(123)		2,430
Redeemable preferred stock		2						(1)						1
Total fixed income securities	_	10,273	_	(222)	_	425		(340)	_	167	_	(560)	_	9,743
Equity securities		72		(6)		1		(1)				·		66
Other investments:		(2.0)		(00)										(0.0)
Free-standing derivatives, net		(38)		(99)				38						(99) (2)
Other assets  Total recurring Level 3 assets	_	10 200	φ-	(227)	φ-	426	φ.	(202)	<sub>e</sub> –	167	<u>-</u>	(FCO)	. <sub>e</sub> –	0.712
Total recurring Level 3 assets	\$_	10,309	»=	(327)	Ф=	420	Ф	(303)	э <sub>=</sub>	10/	» =	(560)	, D	9,712
Liabilities Contractholder funds: Derivatives embedded in life and														
annuity contracts	\$	(90)	\$	(30)	\$		\$	1	\$		\$		\$	(119)
Total recurring Level 3	-	` '	. —	· ´	_		φ.		, _		_			<u> </u>
liabilities	\$_	(90)	\$_	(30)	\$_		\$	1	\$		\$		\$	(119)

<sup>(1)</sup> The effect to net income totals \$(357) million and is reported in the Condensed Consolidated Statements of Operations as follows: \$(345) million in realized capital gains and losses, \$22 million in net investment income, \$(4) million in interest credited to contractholder funds and \$(30) million in life and annuity contract benefits.

23

The following table presents the rollforward of Level 3 assets and liabilities held at fair value on a recurring basis during the six months ended June 30, 2010.

(\$ in millions)			_	Total ro unrealized incli		s (losses)								
		Balance as of December 31, 2009		Net income (1)		OCI on Statement of Financial Position		Purchases, sales, issuances and settlements, net		Transfers into Level 3		Transfers out of Level 3		Balance as of June 30, 2010
Assets Fixed income securities:	_	2005	_	meome	_	1 00111011	•	settlements, net	_	Levers	_	Levers	_	
Municipal	\$	2,706	\$	(47)	\$	58	\$	(452)	\$	16	\$	(84)	\$	2,197
Corporate		2,241		(37)		101		34		163		(277)		2,225
Foreign government		20		(100)		242		(20) 179				(1.4)		2.010
RMBS CMBS		1,671 1,404		(169) (107)		343 300		(328)		24		(14) (413)		2,010 880
ABS		2,001		18		99		472				(160)		2,430
Redeemable preferred stock		2,001						(1)				(100)		1
Total fixed income securities	_	10,045	_	(342)	_	901		(116)	_	203	_	(948)	_	9,743
Equity securities Other investments:		69		(6)		4		3				(4)		66
Free-standing derivatives, net		55		(232)				78						(99) (2)
Other assets	_	2	_		_				_		_			2
Total recurring Level 3 assets	\$_	10,171	\$_	(580)	\$_	905	\$	(35)	\$_	203	\$_	(952)	\$=	9,712
Liabilities														
Contractholder funds:														
Derivatives embedded in life and	\$	(110)	\$	(12)	\$		¢	3	¢		¢		¢	(119)
annuity contracts  Total recurring Level 3	Φ_	(110)	Φ_	(12)	Φ_	<del></del>	Ф	3	Φ_		<b>a</b>		. <b>.</b> _	(119)
liabilities	\$_	(110)	\$_	(12)	\$		\$	3	\$		\$_		\$_	(119)

<sup>(2)</sup> Comprises \$22 million of assets and \$78 million of liabilities.

 $<sup>^{\</sup>left(2\right)}$  Comprises \$27 million of assets and \$126 million of liabilities.

(1) The effect to net income totals \$(592) million and is reported in the Condensed Consolidated Statements of Operations as follows: \$(631) million in realized capital gains and losses, \$54 million in net investment income, \$(3) million in interest credited to contractholder funds and \$(12) million in life and annuity contract benefits.

(2) Comprises \$27 million of assets and \$126 million of liabilities.

Transfers between level categorizations may occur due to changes in the availability of market observable inputs, which generally are caused by changes in market conditions such as liquidity, trading volume or bid-ask spreads. Transfers between level categorizations may also occur due to changes in the valuation source. For example, in situations where a fair value quote is not provided by the Company's independent third-party valuation service provider and as a result the price is stale or has been replaced with a broker quote, the security is transferred into Level 3. Transfers in and out of level categorizations are reported as having occurred at the beginning of the quarter in which the transfer occurred. Therefore, for all transfers into Level 3, all realized and changes in unrealized gains and losses in the quarter of transfer are reflected in the Level 3 rollforward table.

There were no transfers between Level 1 and Level 2 during the three and six months ended June 30, 2011 or 2010.

During the six months ended June 30, 2011 and the three and six months ended June 30, 2010, certain CMBS and ABS were transferred into Level 2 from Level 3 as a result of increased liquidity in the market and the availability of market observable quoted prices for similar assets. When transferring these securities into Level 2, the Company did not change the source of fair value estimates or modify the estimates received from independent third-party valuation service providers or the internal valuation approach. Accordingly, for securities included within this group, there was no change in fair value in conjunction with the transfer resulting in a realized or unrealized gain or loss.

Transfers into Level 3 during the three and six months ended June 30, 2011 and 2010 included situations where a fair value quote was not provided by the Company's independent third-party valuation service provider and as a result the price was stale or had been replaced with a broker quote resulting in the security being classified as Level

24

3. Transfers out of Level 3 during the three and six months ended June 30, 2011 and 2010 included situations where a broker quote was used in the prior period and a fair value quote became available from the Company's independent third-party valuation service provider in the current period. A quote utilizing the new pricing source was not available as of the prior period, and any gains or losses related to the change in valuation source for individual securities were not significant.

The following table provides the total gains and (losses) included in net income for Level 3 assets and liabilities still held as of June 30.

(\$ in millions)	Three m	onths one 30,	ended		Six moi Jui	nths e	
	 2011		2010	_	2011		2010
Assets	 						
Fixed income securities:							
Municipal	\$ (5)	\$	(31)	\$	(15)	\$	(43)
Corporate	6		(14)		10		(50)
RMBS	(27)		(106)		(63)		(164)
CMBS	(11)		(19)		(16)		(43)
ABS	5		14		7		15
Total fixed income securities	(32)		(156)		(77)		(285)
Equity securities			(7)		(10)		(6)
Other investments:							
Free-standing derivatives, net			(55)		3		(141)
Total recurring Level 3 assets	\$ (32)	\$	(218)	\$	(84)	\$	(432)
Liabilities							
Contractholder funds:							
Derivatives embedded in life and annuity contracts	\$ (34)	\$	(30)	\$	(26)	\$	(12)
Total recurring Level 3 liabilities	\$ (34)	\$	(30)	\$	(26)	\$	(12)

The amounts in the table above represent gains and losses included in net income for the period of time that the asset or liability was determined to be in Level 3. These gains and losses total \$(66) million for the three months ended June 30, 2011 and are reported as follows: \$(41) million in realized capital gains and losses, \$9 million in net investment income, \$(26) million in interest credited to contractholder funds and \$(8) million in life and annuity contract benefits. These gains and losses total \$(248) million for the three months ended June 30, 2010 and are reported as follows: \$(233) million in realized capital gains and losses, \$20 million in net investment income, \$(5) million in interest credited to contractholder funds and \$(30) million in realized capital gains and losses, \$13 million in net investment income, \$(63) million in interest credited to contractholder funds and \$37 million in life and annuity contract benefits. These gains and losses total \$(444) million for the six months ended June 30, 2010 and are reported as follows: \$(461) million in realized capital gains and losses, \$34 million in net investment income, \$(5) million in interest credited to contractholder funds and \$(12) million in life and annuity contract benefits.

25

Presented below are the carrying values and fair value estimates of financial instruments not carried at fair value.

Financial assets

(\$ in millions) June 30, 2011 December 31, 2010

Carrying Fair Carrying Fair

	value	value	value	value
Mortgage loans	\$ 6,827	\$ 6,879	\$ 6,679	\$ 6,439
Limited partnership interests - cost basis	1,408	1,704	1,348	1,481
Bank loans	354	351	363	355

The fair value of mortgage loans is based on discounted contractual cash flows or, if the loans are impaired due to credit reasons, the fair value of collateral less costs to sell. Risk adjusted discount rates are selected using current rates at which similar loans would be made to borrowers with similar characteristics, using similar types of properties as collateral. The fair value of limited partnership interests accounted for on the cost basis is determined using reported net asset values of the underlying funds. The fair value of bank loans, which are reported in other investments, is based on broker quotes from brokers familiar with the loans and current market conditions.

#### **Financial liabilities**

(\$ in millions)	June	30, 20	11	Decemb	er 31,	2010
	Carrying value		Fair value	 Carrying value		Fair value
Contractholder funds on investment contracts	\$ 33,126	\$	32,269	\$ 36,163	\$	35,194
Long-term debt	5,907		6,222	5,908		6,325
Liability for collateral	907		907	484		484

The fair value of contractholder funds on investment contracts is based on the terms of the underlying contracts utilizing prevailing market rates for similar contracts adjusted for the Company's own credit risk. Deferred annuities included in contractholder funds are valued using discounted cash flow models which incorporate market value margins, which are based on the cost of holding economic capital, and the Company's own credit risk. Immediate annuities without life contingencies and fixed rate funding agreements are valued at the present value of future benefits using market implied interest rates which include the Company's own credit risk.

The fair value of long-term debt is based on market observable data (such as the fair value of the debt when traded as an asset) or, in certain cases, is determined using discounted cash flow calculations based on current interest rates for instruments with comparable terms and considers the Company's own credit risk. The liability for collateral is valued at carrying value due to its short-term nature.

#### 6. Derivative Financial Instruments

The Company primarily uses derivatives for risk management, to partially mitigate potential adverse impacts from changes in risk-free interest rates, negative equity market valuations and increases in credit spreads, and asset replication. In addition, the Company has derivatives embedded in non-derivative host contracts that are required to be separated from the host contracts and accounted for at fair value. With the exception of non-hedge derivatives used for asset replication and non-hedge embedded derivatives, all of the Company's derivatives are evaluated for their ongoing effectiveness as either accounting hedge or non-hedge derivative financial instruments on at least a quarterly basis. The Company does not use derivatives for trading purposes. Non-hedge accounting is generally used for "portfolio" level hedging strategies where the terms of the individual hedged items do not meet the strict homogeneity requirements to permit the application of hedge accounting.

Property-Liability uses interest rate swaps to mitigate municipal bond interest rate risk within the municipal bond portfolio. Equity index futures are used by Property-Liability to offset valuation losses in the equity portfolio during periods of declining equity market values. Credit default swaps are typically used to mitigate the credit risk within the Property-Liability fixed income portfolio. Prior to March 31, 2011, Property-Liability used interest rate swaption contracts and exchange traded options on interest rate futures to offset potential declining fixed income market values resulting from significant increases in interest rates. Prior to March 31, 2011, exchange traded equity put options were utilized by Property-Liability for overall equity portfolio protection from significant declines in equity market values below a targeted level.

26

Portfolio duration management is a risk management strategy that is principally employed by Property-Liability wherein financial futures and interest rate swaps are utilized to change the duration of the portfolio in order to offset the economic effect that interest rates would otherwise have on the fair value of its fixed income securities.

Property-Liability uses futures to hedge the market risk related to deferred compensation liability contracts and forward contracts to hedge foreign currency risk associated with holding foreign currency denominated investments and foreign operations.

Asset-liability management is a risk management strategy that is principally employed by Allstate Financial to balance the respective interest-rate sensitivities of its assets and liabilities. Depending upon the attributes of the assets acquired and liabilities issued, derivative instruments such as interest rate swaps, caps, floors, swaptions and futures are utilized to change the interest rate characteristics of existing assets and liabilities to ensure the relationship is maintained within specified ranges and to reduce exposure to rising or falling interest rates. Allstate Financial uses financial futures and interest rate swaps to hedge anticipated asset purchases and liability issuances and futures and options for hedging the equity exposure contained in its equity indexed life and annuity product contracts that offer equity returns to contractholders. In addition, Allstate Financial uses interest rate swaps to hedge interest rate risk inherent in funding agreements.

Allstate Financial uses foreign currency swaps primarily to reduce the foreign currency risk associated with issuing foreign currency denominated funding agreements and holding foreign currency denominated investments. Credit default swaps are also typically used to mitigate the credit risk within the Allstate Financial fixed income portfolio.

When derivatives meet specific criteria, they may be designated as accounting hedges and accounted for as fair value, cash flow, foreign currency fair value or foreign currency cash flow hedges. Allstate Financial designates certain of its interest rate and foreign currency swap contracts and certain investment risk transfer reinsurance agreements as fair value hedges when the hedging instrument is highly effective in offsetting the risk of changes in the fair value of the hedged item. Allstate Financial designates certain of its foreign currency swap contracts as cash flow hedges when the hedging instrument is highly effective in offsetting the exposure of variations in cash flows for the hedged risk that could affect net income. Amounts are reclassified to net investment income or realized capital gains and losses as the hedged item affects net income.

Asset replication refers to the "synthetic" creation of assets through the use of derivatives and primarily investment grade host bonds to replicate securities that are either unavailable in the cash markets or more economical to acquire in synthetic form. The Company replicates fixed income securities using a combination of a credit default swap and one or more highly rated fixed income securities to synthetically replicate the economic characteristics of one or more cash market securities. The Company also creates "synthetic" exposure to equity markets through the use of exchange traded equity index future contracts and an investment grade host bond.

The Company's primary embedded derivatives are equity options in Allstate Financial life and annuity product contracts, which provide equity returns to contractholders; equity-indexed notes containing equity call options, which provide a coupon payout that is determined using one or more equity-based indices; credit default swaps in synthetic collateralized debt obligations, which provide enhanced coupon rates as a result of selling credit protection; and conversion options in fixed income securities, which provide the Company with the right to convert the instrument into a predetermined number of shares of common stock. Substantially all of the fixed income securities with conversion options were sold in March 2011.

The notional amounts specified in the contracts are used to calculate the exchange of contractual payments under the agreements and are generally not representative of the potential for gain or loss on these agreements. However, the notional amounts specified in credit default swaps where the Company has sold credit protection represent the maximum amount of potential loss, assuming no recoveries.

Fair value, which is equal to the carrying value, is the estimated amount that the Company would receive or pay to terminate the derivative contracts at the reporting date. The carrying value amounts for OTC derivatives are further adjusted for the effects, if any, of legally enforceable master netting agreements and are presented on a net basis, by counterparty agreement, in the Condensed Consolidated Statements of Financial Position. For certain exchange traded derivatives, the exchange requires margin deposits as well as daily cash settlements of margin accounts. As of June 30, 2011, the Company pledged \$12 million of cash and securities in the form of margin deposits.

27

For those derivatives which qualify for fair value hedge accounting, net income includes the changes in the fair value of both the derivative instrument and the hedged risk, and therefore reflects any hedging ineffectiveness. For cash flow hedges, gains and losses are amortized from accumulated other comprehensive income and are reported in net income in the same period the forecasted transactions being hedged impact net income. For embedded derivatives in fixed income securities, net income includes the change in fair value of the embedded derivative and accretion income related to the host instrument. For non-hedge derivatives, net income includes changes in fair value and accrued periodic settlements, when applicable.

The following table provides a summary of the volume and fair value positions of derivative instruments as well as their reporting location in the Condensed Consolidated Statement of Financial Position as of June 30, 2011.

(\$ in millions, except number of contracts)				Asset deriva	tives					
			Volum	e (1)						
	Balance sheet location		Notional amount	Number of contracts		Fair value, net	_	Gross asset	_	Gross liability
Derivatives designated as accounting hedging instruments Interest rate swap agreements	Other investments	\$	207	n/a	\$	(15)	\$		\$	(15)
Foreign currency swap agreements	Other investments	Ψ	50	n/a	Ψ	(5)	Ψ	2	Ψ	(7)
Total	Other investments	\$	257	n/a	\$	(20)	\$	2	\$	(22)
1001		Ψ	207		Ψ	(20)	Ψ_		Ψ_	(22)
Derivatives not designated as accounting hedging instruments										
Interest rate contracts										
Interest rate swap agreements	Other investments	\$	5,585	n/a	\$	138	\$	161	\$	(23)
Interest rate swaption agreements	Other investments		2,250	n/a		13		13		
Interest rate cap and floor agreements	Other investments		1,383	n/a		(5)				(5)
Financial futures contracts and options	Other investments		n/a	9,000		5		5		
Financial futures contracts and options	Other assets		n/a	840						
Equity and index contracts										
Options, futures and warrants (2)	Other investments		160	14,931		221		221		
Options, futures and warrants	Other assets		n/a	1,183		2		2		
Embedded derivative financial instruments										
Conversion options	Fixed income securities		5	n/a						
Equity-indexed call options	Fixed income securities		150	n/a		16		16		
Credit default swaps	Fixed income securities		172	n/a		(78)				(78)
Other embedded derivative financial instruments	Other investments		1,000	n/a						
Credit default contracts										
Credit default swaps - buying protection	Other investments		121	n/a		(1)		1		(2)
Credit default swaps - selling protection	Other investments		60	n/a		(1)		1		(2)
Other contracts										
Other contracts	Other investments		6	n/a						
Other contracts	Other assets		5	n/a		1	_	1	_	
Total		\$	10,897	25,954	\$	311	\$	421	\$	(110)
Total asset derivatives		\$	11,154	25,954	\$	291	\$	423	\$	(132)

<sup>(1)</sup> Volume for OTC derivative contracts is represented by their notional amounts. Volume for exchange traded derivatives is represented by the number of contracts, which is the basis on which they are traded. (n/a = not applicable)

28

			Liu	omey derivatives					
			Volur	ne <sup>(1)</sup>					
	Balance sheet location	_	Notional amount	Number of contracts		Fair value, net	Gross asset		Gross liability
Derivatives designated as accounting hedging			,		_			_	-
instruments									
Interest rate swap agreements	Other liabilities & accrued expenses	\$	39	n/a	\$	(2)	\$ 	\$	(2)
Foreign currency swap agreements	Other liabilities & accrued expenses		152	n/a		(27)			(27)
Total	-	\$	191	n/a	\$	(29)	\$ 	\$	(29)

Liability derivatives

<sup>(2)</sup> In addition to the number of contracts presented in the table, the Company held 308,430 stock rights and 4,389,707 stock warrants. Stock rights and warrants can be converted to cash upon sale of those instruments or exercised for shares of common stock.

<b>▼</b>	0.1 11.1.01.1 0 1		D 450	,				20		(4.0)
Interest rate swap agreements	Other liabilities & accrued expenses	\$	2,479	n/a	\$	15	\$	28	\$	(13)
Interest rate swaption agreements	Other liabilities & accrued expenses		940	n/a		5		5		
Interest rate cap and floor agreements	Other liabilities & accrued expenses		1,669	n/a		(19)		1		(20)
Financial futures contracts and options	Other liabilities & accrued expenses		n/a	270		`				`
Equity and index contracts	•									
Options and futures	Other liabilities & accrued expenses		n/a	15,316		(104)				(104)
Foreign currency contracts	*									
Foreign currency swap agreements	Other liabilities & accrued expenses		50	n/a						
Foreign currency forwards and options	Other liabilities & accrued expenses		277	n/a		(2)		2		(4)
Embedded derivative financial instruments	*									
Guaranteed accumulation benefits	Contractholder funds		1,085	n/a		(62)				(62)
Guaranteed withdrawal benefits	Contractholder funds		737	n/a		(31)				(31)
Equity-indexed and forward starting options in life and						` ′				` ′
annuity product contracts	Contractholder funds		4,432	n/a		(528)				(528)
Other embedded derivative financial instruments	Contractholder funds		85	n/a		(8)				(8)
Credit default contracts						` ,				` ,
Credit default swaps – buying protection	Other liabilities & accrued expenses		1,020	n/a		(10)		5		(15)
Credit default swaps – selling protection	Other liabilities & accrued expenses		605	n/a		(54)		2		(56)
Total	*	\$	13,379	15,586	\$	(798)	\$	43	\$	(841)
		· -			-		_		-	
Total liability derivatives		\$	13,570	15,586	\$	(827)	\$	43	\$	(870)
•		~=			-		_		_	
Total derivatives			24.724	41 540	\$	(520)				
Iour acritatives		\$	24,724	41,540	Ψ	(536)				

(1) Volume for OTC derivative contracts is represented by their notional amounts. Volume for exchange traded derivatives is represented by the number of contracts, which is the basis on which they are traded. (n/a = not applicable)

29

The following table provides a summary of the volume and fair value positions of derivative instruments as well as their reporting location in the Condensed Consolidated Statement of Financial Position as of December 31, 2010.

(\$ in millions, except number of contracts)	Asset derivatives												
			Volum	e <sup>(1)</sup>									
	Balance sheet location	_	Notional amount	Number of contracts		Fair value, net		Gross asset		Gross liability			
Derivatives designated as accounting hedging instruments					_		_		_				
Interest rate swap agreements	Other investments	\$	156	n/a	\$	(18)	\$		\$	(18)			
Foreign currency swap agreements	Other investments		64	n/a		2		3		(1)			
Total		\$	220	n/a	\$	(16)	\$	3	\$	(19)			
Derivatives not designated as accounting hedging instruments													
Interest rate contracts													
Interest rate swap agreements	Other investments	\$	1,469	n/a	\$	65	\$	81	\$	(16)			
Interest rate swaption agreements	Other investments		4,161	n/a		50		50		`			
Interest rate cap and floor agreements	Other investments		226	n/a		(2)		1		(3)			
Financial futures contracts and options	Other investments		n/a	8,000		`3		3					
Financial futures contracts and options	Other assets		n/a	1,420									
Equity and index contracts													
Options, futures and warrants (2)	Other investments		64	38,451		359		359					
Options, futures and warrants	Other assets		n/a	292									
Foreign currency contracts													
Foreign currency swap agreements	Other investments		90	n/a		6		6					
Foreign currency forwards and options	Other investments		257	n/a		6		7		(1)			
Embedded derivative financial instruments													
Conversion options	Fixed income securities		820	n/a		236		238		(2)			
Equity-indexed call options	Fixed income securities		300	n/a		47		47					
Credit default swaps	Fixed income securities		181	n/a		(88)				(88)			
Other embedded derivative financial instruments	Other investments		1,000	n/a									
Credit default contracts													
Credit default swaps - buying protection	Other investments		299	n/a		(5)		2		(7)			
Credit default swaps - selling protection	Other investments		150	n/a		(8)		2		(10)			
Other contracts													
Other contracts	Other investments		13	n/a									
Other contracts	Other assets		5	n/a		1		1					
Total		\$	9,035	48,163	\$	670	\$	797	\$	(127)			
Total asset derivatives		\$	9,255	48,163	\$	654	\$	800	\$	(146)			

<sup>(1)</sup> Volume for OTC derivative contracts is represented by their notional amounts. Volume for exchange traded derivatives is represented by the number of contracts, which is the basis on which they are traded. (n/a = not applicable)

30

		Lia	ability derivatives						
		Volum	ne (1)						
Desirable desirable and a second second second	Balance sheet location	 Notional amount	Number of contracts	_	Fair value, net	_	Gross asset	_	Gross liability
Derivatives designated as accounting hedging instruments Interest rate swap agreements Interest rate swap agreements Foreign currency swap agreements Foreign currency and interest rate swap agreements Foreign currency and interest rate swap agreements Total	Other liabilities & accrued expenses Contractholder funds Other liabilities & accrued expenses Other liabilities & accrued expenses Contractholder funds	\$  3,345  138 435  3,918	n/a n/a n/a n/a n/a n/a	\$ \$	(181) 2 (20) 34 28 (137)	\$ \$_	20 2  34 28 84	\$ 	(201)  (20)   (221)
Derivatives not designated as accounting hedging instruments Interest rate contracts									
Interest rate swap agreements Interest rate swaption agreements Interest rate cap and floor agreements Financial futures contracts and options	Other liabilities & accrued expenses Other liabilities & accrued expenses Other liabilities & accrued expenses Other liabilities & accrued expenses	\$ 4,543 4,400 3,216 n/a	n/a n/a n/a 15,150	\$	29 18 (22) (1)	\$	97 18 1 	\$	(68)  (23) (1)
Equity and index contracts Options and futures Foreign currency contracts	Other liabilities & accrued expenses	64	21,585		(168)		2		(170)
Foreign currency forwards and options  Embedded derivative financial instruments	Other liabilities & accrued expenses	316	n/a		1		2		(1)
Guaranteed accumulation benefits Guaranteed withdrawal benefits Equity-indexed and forward starting options in life and	Contractholder funds Contractholder funds Contractholder funds	1,067 739 4,694	n/a n/a n/a		(88) (47) (515)		  		(88) (47) (515)

<sup>(2)</sup> In addition to the number of contracts presented in the table, the Company held 2,768 stock rights and 1,379,932 stock warrants. Stock warrants can be converted to cash upon sale of those instruments or exercised for shares of common stock.

annuity product contracts Other embedded derivative financial instruments Credit default contracts	Contractholder funds	85	n/a		(3)			(3)
Credit default swaps – buying protection Credit default swaps – selling protection	Other liabilities & accrued expenses Other liabilities & accrued expenses	1,127 482	n/a		(13) (66)		6	(19)
Total	Other Habilities & accrued expenses	\$ 20,733	n/a 36,735	\$	(875)	\$	127	\$ (1,002)
Total liability derivatives		\$ 24,651	36,735	\$_	(1,012)	\$_	211	\$ (1,223)
Total derivatives		\$ 33,906	84,898	\$	(358)			

(1) Volume for OTC derivative contracts is represented by their notional amounts. Volume for exchange traded derivatives is represented by the number of contracts, which is the basis on which they are traded. (n/a = not applicable)

31

The following table provides a summary of the impacts of the Company's foreign currency contracts in cash flow hedging relationships in the Condensed Consolidated Statements of Operations and the Condensed Consolidated Statements of Financial Position. Amortization of net losses from accumulated other comprehensive income related to cash flow hedges is expected to be \$7 million during the next twelve months.

(\$ in millions)	Three m	onths	ended		Six mo	onths	ended
	Ju	ne 30	,		Jı	ıne 30	),
Effective portion	 2011		2010	_	2011		2010
(Loss) gain recognized in OCI on derivatives during the period	\$ (5)	\$	22	\$	(13)	\$	28
(Loss) gain recognized in OCI on derivatives during the term of the hedging							
relationship	(36)		2		(36)		2
Gain reclassified from AOCI into income (net investment income)	1				1		1
Gain reclassified from AOCI into income (realized capital gains and losses)			2				2
Ineffective portion and amount excluded from effectiveness							
Gain recognized in income on derivatives (realized capital gains and losses)							

The following tables present gains and losses from valuation, settlements and hedge ineffectiveness reported on derivatives used in fair value hedging relationships and derivatives not designated as accounting hedging instruments in the Condensed Consolidated Statements of Operations.

(\$ in millions)					Three months e	nde	d June 30, 2011				
Derivatives in fair value accounting hedging	_	Net investment income	Realized capital gains and losses	_	Life and annuity contract benefits	•	Interest credited to contractholder funds	_	Operating costs and expenses		Total gain (loss) recognized in net income on derivatives
relationships											
Interest rate contracts	\$	(2)	\$ 	\$		\$		\$		\$	(2)
Subtotal	_	(2)								_	(2)
Derivatives not designated as accounting hedging	2										
instruments	,										
Interest rate contracts			(53)								(53)
Equity and index contracts			`				8				8
Embedded derivative financial instruments			(3)		(8)		9				(2) 3
Credit default contracts			3								3
Other contracts							3				3
Subtotal			(53)		(8)	•	20	_		. "	(41)
Total	\$	(2)	\$ (53)	\$	(8)	\$	20	\$		\$	(43)

32

	Six months ended June 30, 2011											
		Net investment income		Realized capital gains and losses		Life and annuity contract benefits		Interest credited to contractholder funds		Operating costs and expenses		Total gain (loss) recognized in net income on derivatives
Derivatives in fair value accounting hedging relationships				_		_	•					
Interest rate contracts Foreign currency and interest rate contracts Subtotal	\$	(1)  (1)	\$	(8)  (8)	\$ -	  	\$	(5) (32) (37)	\$ -	  	\$ _	(14) (32) (46)
Derivatives not designated as accounting hedging instruments												
Interest rate contracts				(104)								(104)
Equity and index contracts				(19)				46		7		34
Embedded derivative financial instruments				5		37		(13)				29
Foreign currency contracts				(5)						2		(3)
Credit default contracts				11								(3) 11 5
Other contracts	_		_	(112)	_			5	_		_	
Subtotal			<sub>e</sub> –	(112)	ф —	37	d.	38	φ -	9	ф —	(28)
Total	\$	(1)	» =	(120)	<b>D</b> =	37	\$	1	Ф =	9	ъ Т	(74)
						Three months e	ndeo	d June 30, 2010				
		Net investment		Realized capital gains and		Life and annuity contract		Interest credited to contractholder		Operating costs and		Total gain (loss) recognized in net income on

2 \$

benefits

\$

13 \$

derivatives

(57)

\$

expenses

income

(72) \$

\$

Derivatives in fair value accounting hedging

relationships

Interest rate contracts

Foreign currency and interest rate contracts Subtotal	(72)	<u>(1)</u> <u>1</u>	<del></del>	(16) (3)		(17) (74)
Derivatives not designated as accounting hedging instruments						
Interest rate contracts		(282)				(282)
Equity and index contracts		82		(70)	(14)	(2)
Embedded derivative financial instruments		(106)	(28)	Ì12	`	(22)
Foreign currency contracts		` 3	`		(1)	2
Credit default contracts		(9)			<u></u>	(9)
Other contracts		1		2		`3´
Subtotal		(311)	(28)	44	(15)	(310)
Total	\$ (72)	\$ (310)	\$ (28)	\$ 41	\$ (15)	\$ (384)

33

	Six months ended June 30, 2010											
Derivatives in fair value accounting hedging	_	Net investment income		Realized capital gains and losses		Life and annuity contract benefits		Interest credited to contractholder funds	_	Operating costs and expenses	_	Total gain (loss) recognized in net income on derivatives
relationships												
Interest rate contracts	\$	(113)	\$	2	\$		\$	12	\$		\$	(99)
Foreign currency and interest rate contracts Subtotal	_			(1)				(40)	_		_	(41)
Suototai	_	(113)		1_				(28)	_		-	(140)
Derivatives not designated as accounting hedging instruments												
Interest rate contracts				(438)								(438)
Equity and index contracts				43				(36)		(8)		(1)
Embedded derivative financial instruments				(119)		(8)		110				(17)
Foreign currency contracts				20						(6)		14
Credit default contracts				(3)								(3)
Other contracts	_			1		<del></del>		2	_		_	3
Subtotal	_			(496)	_	(8)	_	76		(14)		(442)
Total	\$ _	(113)	\$	(495)	\$	(8)	\$	48	\$	(14)	\$	(582)

The following tables provide a summary of the changes in fair value of the Company's fair value hedging relationships in the Condensed Consolidated Statements of Operations.

(\$ in millions)		Three months ended June 30, 2011											
		Gain (loss)	on de	rivatives		Gain (loss) o	n hed	lged risk					
Location of gain or (loss) recognized in net income on derivatives	_	Interest rate contracts		Foreign currency & interest rate contracts		Contractholder funds		Investments					
Net investment income	\$	2	\$		\$		\$	(2)					
Realized capital gains and losses	_						_						
Total	\$	2	\$		\$_	<u>-</u>	\$	(2)					
	_	Gain (loss)	on de		_	June 30, 2011 Gain (loss) o	n hed	lged risk					
Location of gain or (loss) recognized		rate		interest rate		Contractholder							
in net income on derivatives		contracts	_	contracts		funds	_	Investments					
Interest credited to contractholder funds	\$	(7)	\$	(34)	\$	41	\$						
Net investment income		23						(23)					
Realized capital gains and losses Total	<u>e</u>	(8) 8	s -	(34)	s -	41	s -	(23)					
10111	» —		Ť =	(34)				(23)					
		34	1										

	Three months ended June 30, 2010												
		Gain (loss)	on de	rivatives		Gain (loss) o	ged risk						
Location of gain or (loss) recognized in net income on derivatives	_	Interest rate contracts		Foreign currency & interest rate contracts		Contractholder funds		Investments					
Interest credited to contractholder funds	\$	9	\$	(24)	\$	15	\$						
Net investment income		(43)		<del></del>				43					
Realized capital gains and losses	_	2	_	(1)			_	<del></del>					
Total	\$	(32)	٥ =	(25)	» <sub>=</sub>	15	\$ _	43					
	_	Gain (loss)	on de		nded	June 30, 2010 Gain (loss) o	n hed	ged risk					
Location of gain or (loss) recognized in net income on derivatives		Interest rate contracts		Foreign currency & interest rate contracts		Contractholder funds		Investments					
Interest credited to contractholder funds	\$	8	\$	(57)	\$	49	\$						
Net investment income		(56)						56					
Net investment income Realized capital gains and losses	_	<u>2</u>	. –	(1)		 	_						
	s <u> </u>	(56) 2 (46)	\$ <u></u>	(1) (58)	\$ _	  49	\$	56  56					

The Company manages its exposure to credit risk by utilizing highly rated counterparties, establishing risk control limits, executing legally enforceable master netting agreements ("MNAs") and obtaining collateral where appropriate. The Company uses MNAs for OTC derivative transactions, including interest rate swap, foreign currency swap, interest rate cap, interest rate floor, credit default swap, forward and certain option agreements (including swaptions). These agreements permit either party to net payments due for transactions covered by the agreements. Under the provisions of the agreements,

collateral is either pledged or obtained when certain predetermined exposure limits are exceeded. As of June 30, 2011, counterparties pledged \$83 million in cash and securities to the Company, and the Company pledged \$107 million in securities to counterparties which includes \$48 million of collateral posted under MNAs for contracts containing credit-risk-contingent provisions that are in a liability position and \$59 million of collateral posted under MNAs for contracts without credit-risk-contingent liabilities. The Company has not incurred any losses on derivative financial instruments due to counterparty nonperformance. Other derivatives, including futures and certain option contracts, are traded on organized exchanges which require margin deposits and guarantee the execution of trades, thereby mitigating any potential credit risk.

Counterparty credit exposure represents the Company's potential loss if all of the counterparties concurrently fail to perform under the contractual terms of the contracts and all collateral, if any, becomes worthless. This exposure is measured by the fair value of OTC derivative contracts with a positive fair value at the reporting date reduced by the effect, if any, of legally enforceable master netting agreements.

The following table summarizes the counterparty credit exposure by counterparty credit rating as it relates to interest rate swap, foreign currency swap, interest rate cap, interest rate floor, free-standing credit default swap, forward and certain option agreements (including swaptions).

(\$ in millions)			June 3	0, 20	11			December 31, 2010							
	Number						Exposure,	Number						Exposure,	
	of		Notional		Credit		net of	of		Notional		Credit		net of	
Rating (1)	counterparties		amount (2)		exposure (2)		collateral (2)	counterparties		amount (2)		exposure <sup>(2)</sup>	_	collateral <sup>(2)</sup>	
AA-	1	\$	567	\$	16	\$		2	\$	2,322	\$	43	\$	16	
A+	3		5,140		28		8	5		3,189		16		10	
A	4		4,161		50		4	3		3,479		17		17	
A-								1		89		31		31	
BBB+	1		5		35		35								
Total	9	\$	9,873	\$	129	\$	47	11	\$	9,079	\$	107	\$	74	

Market risk is the risk that the Company will incur losses due to adverse changes in market rates and prices. Market risk exists for all of the derivative financial instruments the Company currently holds, as these instruments may become less valuable due to adverse changes in market conditions. To limit this risk, the Company's senior management has established risk control limits. In addition, changes in fair value of the derivative financial

35

instruments that the Company uses for risk management purposes are generally offset by the change in the fair value or cash flows of the hedged risk component of the related assets, liabilities or forecasted transactions.

Certain of the Company's derivative instruments contain credit-risk-contingent termination events, cross-default provisions and credit support annex agreements. Credit-risk-contingent termination events allow the counterparties to terminate the derivative on certain dates if AIC's, ALIC's or Allstate Life Insurance Company of New York's ("ALNY") financial strength credit ratings by Moody's or S&P fall below a certain level or in the event AIC, ALIC or ALNY are no longer rated by both Moody's and S&P. Credit-risk-contingent cross-default provisions allow the counterparties to terminate the derivative instruments if the Company defaults by pre-determined threshold amounts on certain debt instruments. Credit-risk-contingent credit support annex agreements specify the amount of collateral the Company must post to counterparties based on AIC's, ALIC's or ALNY's financial strength credit ratings by Moody's or S&P, or in the event AIC, ALIC or ALNY are no longer rated by both Moody's and S&P.

The following summarizes the fair value of derivative instruments with termination, cross-default or collateral credit-risk-contingent features that are in a liability position, as well as the fair value of assets and collateral that are netted against the liability in accordance with provisions within legally enforceable MNAs.

(\$ in millions)	June 30, 2011	December 31, 2010
Gross liability fair value of contracts containing credit-risk-contingent features	\$ 146	\$ 448
Gross asset fair value of contracts containing credit-risk-contingent features and subject to MNAs	(90)	(255)
Collateral posted under MNAs for contracts containing credit-risk-contingent features	(48)	(171)
Maximum amount of additional exposure for contracts with credit-risk-contingent features if all features were triggered	 	 
concurrently	\$ 8	\$ 22

#### Credit derivatives - selling protection

(\$ in millions)

Free-standing credit default swaps ("CDS") are utilized for selling credit protection against a specified credit event. A credit default swap is a derivative instrument, representing an agreement between two parties to exchange the credit risk of a specified entity (or a group of entities), or an index based on the credit risk of a group of entities (all commonly referred to as the "reference entity" or a portfolio of "reference entities"), in return for a periodic premium. In selling protection, CDS are used to replicate fixed income securities and to complement the cash market when credit exposure to certain issuers is not available or when the derivative alternative is less expensive than the cash market alternative. CDS typically have a five-year term.

Notional amount

The following table shows the CDS notional amounts by credit rating and fair value of protection sold as of June 30, 2011:

(5 III IIIIIIIIIII)	Notional amount											
							BB and			Fair		
	AA		A		BBB		lower		Total		value	
Single name	 	_					<u>.</u>					
Investment grade corporate												
debt	\$ 50	\$	138	\$	130	\$	45	\$	363	\$	(1)	
High yield debt							2		2			
Municipal	135								135		(8)	
Subtotal	 185	_	138		130		47		500		(9)	
Baskets												
Tranche												
Investment grade corporate							65		65		(18)	

 <sup>(1)</sup> Rating is the lower of S&P or Moody's ratings.
 (2) Only OTC derivatives with a net positive fair value are included for each counterparty.

debt						
First-to-default						
Municipal			100			
Subtotal	•		100		65	
Total	\$	185	\$ 238	\$ 130	\$ 112	\$

36

100

165

(28)

(46)

The following table shows the CDS notional amounts by credit rating and fair value of protection sold as of December 31, 2010:

(\$ in millions)	Notional amount											
								BB and				Fair
		AA		A		BBB		lower		Total		value
Single name			_				_		_			
Investment grade corporate												
debt	\$	50	\$	148	\$	103	\$	25	\$	326	\$	(4)
High yield debt								6		6		
Municipal		135								135		(14)
Subtotal		185		148		103	_	31	_	467	_	(18)
Baskets												
Tranche												
Investment grade corporate												
debt								65		65		(19)
First-to-default												
Municipal				100						100		(37)
Subtotal			_	100			_	65	_	165		(56)
Total	\$	185	\$	248	\$	103	\$	96	\$	632	\$	(74)

In selling protection with CDS, the Company sells credit protection on an identified single name, a basket of names in a first-to-default ("FTD") structure or a specific tranche of a basket, or credit derivative index ("CDX") that is generally investment grade, and in return receives periodic premiums through expiration or termination of the agreement. With single name CDS, this premium or credit spread generally corresponds to the difference between the yield on the reference entity's public fixed maturity cash instruments and swap rates at the time the agreement is executed. With a FTD basket or a tranche of a basket, because of the additional credit risk inherent in a basket of named reference entities, the premium generally corresponds to a high proportion of the sum of the credit spreads of the names in the basket and the correlation between the names. CDX index is utilized to take a position on multiple (generally 125) reference entities. Credit events are typically defined as bankruptcy, failure to pay, or restructuring, depending on the nature of the reference entities. If a credit event occurs, the Company settles with the counterparty, either through physical settlement or cash settlement. In a physical settlement, a reference asset is delivered by the buyer of protection to the Company, in exchange for cash payment at par, whereas in a cash settlement, the Company pays the difference between par and the prescribed value of the reference asset. When a credit event occurs in a single name or FTD basket (for FTD, the first credit event occurring for any one name in the basket), the contract terminates at the time of settlement. When a credit event occurs in a tranche of a basket, there is no immediate impact to the Company until cumulative losses in the basket exceed the contractual subordination. To date, realized losses have not exceeded the subordination. For CDX index, the reference entity's name incurring the credit event is removed from the index while the contract continues until expiration. The

The Company monitors risk associated with credit derivatives through individual name credit limits at both a credit derivative and a combined cash instrument/credit derivative level. The ratings of individual names for which protection has been sold are also monitored.

In addition to the CDS described above, the Company's synthetic collateralized debt obligations contain embedded credit default swaps which sell protection on a basket of reference entities. The synthetic collateralized debt obligations are fully funded; therefore, the Company is not obligated to contribute additional funds when credit events occur related to the reference entities named in the embedded credit default swaps. The Company's maximum amount at risk equals the amount of its aggregate initial investment in the synthetic collateralized debt obligations.

### 7. Reserve for Property-Liability Insurance Claims and Claims Expense

The Company establishes reserves for claims and claims expense on reported and unreported claims of insured losses. The Company's reserving process takes into account known facts and interpretations of circumstances and factors including the Company's experience with similar cases, actual claims paid, historical trends involving claim payment patterns and pending levels of unpaid claims, loss management programs, product mix and contractual terms, changes in law and regulation, judicial decisions, and economic conditions. In the normal course of business, the Company may also supplement its claims processes by utilizing third party adjusters, appraisers, engineers, inspectors,

37

and other professionals and information sources to assess and settle catastrophe and non-catastrophe related claims. The effects of inflation are implicitly considered in the reserving process.

Because reserves are estimates of unpaid portions of losses that have occurred, including incurred but not reported losses, the establishment of appropriate reserves, including reserves for catastrophes, is an inherently uncertain and complex process. The ultimate cost of losses may vary materially from recorded amounts, which are based on management's best estimates. The highest degree of uncertainty is associated with reserves for losses incurred in the current reporting period as it contains the greatest proportion of losses that have not been reported or settled. The Company regularly updates its reserve estimates as new information becomes available and as events unfold that may affect the resolution of unsettled claims. Changes in prior year reserve estimates, which may be material, are reported in property-liability insurance claims and claims expense in the Condensed Consolidated Statements of Operations in the period such changes are determined.

Management believes that the reserve for property-liability insurance claims and claims expense, net of reinsurance recoverables, is appropriately established in the aggregate and adequate to cover the ultimate net cost of reported and unreported claims arising from losses which had occurred by the date of the Condensed Consolidated Statements of Financial Position based on available facts, technology, laws and regulations.

#### 8. Reinsurance

Property-liability insurance premiums earned and life and annuity premiums and contract charges have been reduced by the reinsurance ceded amounts shown in the following table:

(\$ in millions)	Three mont June		Six months ended June 30,				
	2011	2010	 2011	2010			
Property-liability insurance premiums earned	\$ 274 \$	273	\$ 544 \$	541			
Life and annuity premiums and contract charges	185	202	378	393			

Property-liability insurance claims and claims expense, life and annuity contract benefits and interest credited to contractholder funds have been reduced by the reinsurance ceded amounts shown in the following table.

(\$ in millions)	Three m Ju	onth ne 30		Six months ended June 30,					
	 2011		2010	_	2011		2010		
Property-liability insurance claims and claims expense	\$ 61	\$	111	\$	198	\$	191		
Life and annuity contract benefits	52		219		136		349		
Interest credited to contractholder funds	6		9		14		16		

#### 9. Company Restructuring

The Company undertakes various programs to reduce expenses. These programs generally involve a reduction in staffing levels, and in certain cases, office closures. Restructuring and related charges include employee termination and relocation benefits, and post-exit rent expenses in connection with these programs, and non-cash charges resulting from pension benefit payments made to agents in connection with the 1999 reorganization of Allstate's multiple agency programs to a single exclusive agency program. In the six months ended June 30, 2011, restructuring programs primarily relate to Allstate Protection's field claim office consolidations and reorganization of technology shared services. The expenses related to these activities are included in the Condensed Consolidated Statements of Operations as restructuring and related charges, and totaled \$11 million and \$13 million during the three months ended June 30, 2011 and 2010, respectively, and \$20 million and \$24 million for the six months ended June 30, 2011 and 2010, respectively.

The following table presents changes in the restructuring liability during the six months ended June 30, 2011.

(\$ in millions)	E	mployee costs	Exit costs	Total liability
Balance as of December 31, 2010	\$	13 \$	\$ 3	\$ 16
Expense incurred		14	5	19
Adjustments to liability		(9)		(9)
Payments applied against liability		(9)	(2)	(11)
Balance as of June 30, 2011	\$	9 \$	6	\$ 15
	38			

The payments applied against the liability for employee costs primarily reflect severance costs, and the payments for exit costs generally consist of post-exit rent expenses and contract termination penalties. As of June 30, 2011, the cumulative amount incurred to date for active programs totaled \$103 million for employee costs and \$45 million for exit costs.

#### 10. Guarantees and Contingent Liabilities

#### State facility assessments

The Company is required to participate in assigned risk plans, reinsurance facilities and joint underwriting associations in various states that provide insurance coverage to individuals or entities that otherwise are unable to purchase such coverage from private insurers. Underwriting results related to these arrangements, which tend to be adverse, have been immaterial to the Company's results of operations. Because of the Company's participation, it may be exposed to losses that surpass the capitalization of these facilities and/or assessments from these facilities.

## **Shared markets**

As a condition of maintaining its licenses to write personal property and casualty insurance in various states, the Company is required to participate in assigned risk plans, reinsurance facilities and joint underwriting associations that provide various types of insurance coverage to individuals or entities that otherwise are unable to purchase such coverage from private insurers. Underwriting results related to these arrangements, which tend to be adverse, have been immaterial to the Company's results of operations.

#### Guarantees

The Company owns certain fixed income securities that obligate the Company to exchange credit risk or to forfeit principal due, depending on the nature or occurrence of specified credit events for the reference entities. In the event all such specified credit events were to occur, the Company's maximum

amount at risk on these fixed income securities, as measured by the amount of the aggregate initial investment, was \$67 million as of June 30, 2011. The obligations associated with these fixed income securities expire at various dates on or before March 11, 2018.

Related to the disposal through reinsurance of substantially all of Allstate Financial's variable annuity business to Prudential in 2006, the Company and its consolidated subsidiaries, ALIC and ALNY, have agreed to indemnify Prudential for certain pre-closing contingent liabilities (including extra-contractual liabilities of ALIC and ALNY and liabilities specifically excluded from the transaction) that ALIC and ALNY have agreed to retain. In addition, the Company, ALIC and ALNY will each indemnify Prudential for certain post-closing liabilities that may arise from the acts of ALIC, ALNY and their agents, including in connection with ALIC's and ALNY's provision of transition services. The reinsurance agreements contain no limitations or indemnifications with regard to insurance risk transfer, and transferred all of the future risks and responsibilities for performance on the underlying variable annuity contracts to Prudential, including those related to benefit guarantees. Management does not believe this agreement will have a material adverse effect on results of operations, cash flows or financial position of the Company.

The Company provides residual value guarantees on Company leased automobiles. If all outstanding leases were terminated effective June 30, 2011, the Company's maximum obligation pursuant to these guarantees, assuming the automobiles have no residual value, would be \$8 million as of June 30, 2011. The remaining term of each residual value guarantee is equal to the term of the underlying lease that ranges from less than one year to three years. Historically, the Company has not made any material payments pursuant to these guarantees.

In the normal course of business, the Company provides standard indemnifications to contractual counterparties in connection with numerous transactions, including acquisitions and divestitures. The types of indemnifications typically provided include indemnifications for breaches of representations and warranties, taxes and certain other liabilities, such as third party lawsuits. The indemnification clauses are often standard contractual terms and are entered into in the normal course of business based on an assessment that the risk of loss would be remote. The terms of the indemnifications vary in duration and nature. In many cases, the maximum obligation is not explicitly stated and the contingencies triggering the obligation to indemnify have not occurred and are not expected to occur. Consequently, the maximum amount of the obligation under such indemnifications is not determinable. Historically, the Company has not made any material payments pursuant to these obligations.

The aggregate liability balance related to all guarantees was not material as of June 30, 2011.

39

#### **Regulation and Compliance**

The Company is subject to changing social, economic and regulatory conditions. From time to time, regulatory authorities or legislative bodies seek to influence and restrict premium rates, require premium refunds to policyholders, require reinstatement of terminated policies, restrict the ability of insurers to cancel or non-renew policies, require insurers to continue to write new policies or limit their ability to write new policies, limit insurers' ability to change coverage terms or to impose underwriting standards, impose additional regulations regarding agent and broker compensation, regulate the nature of and amount of investments, and otherwise expand overall regulation of insurance products and the insurance industry. The Company has established procedures and policies to facilitate compliance with laws and regulations, to foster prudent business operations, and to support financial reporting. The Company routinely reviews its practices to validate compliance with laws and regulations and with internal procedures and policies. As a result of these reviews, from time to time the Company may decide to modify some of its procedures and policies. Such modifications, and the reviews that led to them, may be accompanied by payments being made and costs being incurred. The ultimate changes and eventual effects of these actions on the Company's business, if any, are uncertain.

#### Legal and regulatory proceedings and inquiries

The Company and certain subsidiaries are involved in a number of lawsuits, regulatory inquiries, and other legal proceedings arising out of various aspects of its business.

### **Background**

These matters raise difficult and complicated factual and legal issues and are subject to many uncertainties and complexities, including the underlying facts of each matter; novel legal issues; variations between jurisdictions in which matters are being litigated, heard, or investigated; differences in applicable laws and judicial interpretations; the length of time before many of these matters might be resolved by settlement, through litigation, or otherwise; the fact that some of the lawsuits are putative class actions in which a class has not been certified and in which the purported class may not be clearly defined; the fact that some of the lawsuits involve multi-state class actions in which the applicable law(s) for the claims at issue is in dispute and therefore unclear; and the current challenging legal environment faced by large corporations and insurance companies.

The outcome of these matters may be affected by decisions, verdicts, and settlements, and the timing of such decisions, verdicts, and settlements, in other individual and class action lawsuits that involve the Company, other insurers, or other entities and by other legal, governmental, and regulatory actions that involve the Company, other insurers, or other entities. The outcome may also be affected by future state or federal legislation, the timing or substance of which cannot be predicted.

In the lawsuits, plaintiffs seek a variety of remedies which may include equitable relief in the form of injunctive and other remedies and monetary relief in the form of contractual and extra-contractual damages. In some cases, the monetary damages sought may include punitive or treble damages. Often specific information about the relief sought, such as the amount of damages, is not available because plaintiffs have not requested specific relief in their pleadings. When specific monetary demands are made, they are often set just below a state court jurisdictional limit in order to seek the maximum amount available in state court, regardless of the specifics of the case, while still avoiding the risk of removal to federal court. In Allstate's experience, monetary demands in pleadings bear little relation to the ultimate loss, if any, to the Company.

In connection with regulatory examinations and proceedings, government authorities may seek various forms of relief, including penalties, restitution, and changes in business practices. The Company may not be advised of the nature and extent of relief sought until the final stages of the examination or proceeding.

#### Accrual and disclosure policy

The Company reviews its lawsuits, regulatory inquiries, and other legal proceedings on an ongoing basis and follows appropriate accounting guidance when making accrual and disclosure decisions. The Company establishes accruals for such matters at management's best estimate when the Company assesses that it is probable that a loss has been incurred and the amount of the loss can be reasonably estimated. The Company does not establish accruals for such matters when the Company does not believe both that it is probable that a loss has been incurred and the amount of the loss can be reasonably estimated. The Company's assessment of whether a loss is reasonably possible or probable is based on its assessment of the ultimate outcome of the matter following all appeals.

40

The Company continues to monitor its lawsuits, regulatory inquiries, and other legal proceedings for further developments that would make the loss contingency both probable and estimable, and accordingly accruable, or that could affect the amount of accruals that have been previously established. There may continue to be exposure to loss in excess of any amount accrued. Disclosure of the nature and amount of an accrual is made when there have been sufficient legal and factual developments such that the Company's ability to resolve the matter would not be impaired by the disclosure of the amount of accrual.

When the Company assesses it is reasonably possible or probable that a loss has been incurred, it discloses the matter. When it is possible to estimate the reasonably possible loss or range of loss above the amount accrued, if any, for the matters disclosed, that estimate is aggregated and disclosed. Disclosure is not required when an estimate of the reasonably possible loss or range of loss cannot be made.

For certain of the matters described below in the "Claims related proceedings" and "Other proceedings" subsections, the Company is able to estimate the reasonably possible loss or range of loss above the amount accrued, if any. In determining whether it is possible to estimate the reasonably possible loss or range of loss, the Company reviews and evaluates the disclosed matters, in conjunction with counsel, in light of potentially relevant factual and legal developments.

These developments may include information learned through the discovery process, rulings on dispositive motions, settlement discussions, information obtained from other sources, experience from managing these and other matters, and other rulings by courts, arbitrators or others. When the Company possesses sufficient appropriate information to develop an estimate of the reasonably possible loss or range of loss above the amount accrued, if any, that estimate is aggregated and disclosed below. There may be other disclosed matters for which a loss is probable or reasonably possible but such an estimate is not possible. Disclosure of the estimate of the reasonably possible loss or range of loss above the amount accrued, if any, for any individual matter would only be considered when there have been sufficient legal and factual developments such that the Company's ability to resolve the matter would not be impaired by the disclosure of the individual estimate.

As of June 30, 2011, the Company estimates that the aggregate range of reasonably possible loss in excess of the amount accrued, if any, for the disclosed matters where such an estimate is possible is zero to \$855 million, pre-tax. This disclosure is not an indication of expected loss, if any. Under accounting guidance, an event is "reasonably possible" if "the chance of the future event or events occurring is more than remote but less than likely" and an event is "remote" if "the chance of the future event or events occurring is slight." This estimate is based upon currently available information and is subject to significant judgment and a variety of assumptions, and known and unknown uncertainties. The matters underlying the estimate will change from time to time, and actual results may vary significantly from the current estimate. The estimate does not include matters or losses for which an estimate is not possible. Therefore, this estimate represents an estimate of possible loss only for certain matters meeting these criteria. It does not represent the Company's maximum possible loss exposure. Information is provided below regarding the nature of all of the disclosed matters and, where specified, the amount, if any, of plaintiff claims associated with these loss contingencies.

Due to the complexity and scope of the matters disclosed in the "Claims related proceedings" and "Other proceedings" subsections below and the many uncertainties that exist, the ultimate outcome of these matters cannot be predicted. In the event of an unfavorable outcome in one or more of these matters, the ultimate liability may be in excess of amounts currently accrued, if any, and may be material to the Company's operating results or cash flows for a particular quarterly or annual period. However, based on information currently known to it, management believes that the ultimate outcome of all matters described below, as they are resolved over time, is not likely to have a material effect on the financial position of the Company.

#### Claims related proceedings

The Company is vigorously defending a number of matters in various stages of development filed in the aftermath of Hurricane Katrina, including individual lawsuits and a statewide putative class action in Louisiana. The Louisiana Attorney General filed a putative class action lawsuit in state court against Allstate and every other homeowner insurer doing business in the State of Louisiana, on behalf of the State, as assignee, and on behalf of a class of Road Home fund recipients (the "Road Home Class Action") alleging that the insurers have failed to pay all damages owed under their policies. The insurers removed the matter to federal court. The district court denied plaintiffs' motion to remand the matter to state court and the U.S. Court of Appeals for the Fifth Circuit ("Fifth Circuit") affirmed that ruling. The defendants filed a motion to dismiss and the plaintiffs filed a motion to remand the claims involving a Road Home subrogation agreement. In March 2009, the district court denied the State's request that its claims be remanded to state court. As for the defendant insurers' motion, the judge granted it in part and denied it in

41

part. Dismissal of all of the extra-contractual claims, including the bad faith and breach of fiduciary duty claims, was granted. Dismissal also was granted of all claims based on the Valued Policy Law and all flood loss claims based on the levee breaches finding that the insurers flood exclusions precluded coverage. The remaining claims are for breach of contract and for declaratory relief on the alleged underpayment of claims by the insurers. The judge did not dismiss the class action allegations. The defendants also had moved to dismiss the complaint on grounds that the State had no standing to bring the lawsuit as an assignee of insureds because of anti-assignment language in the insurers' policies. The judge denied the defendants' motion for reconsideration on the assignment issue but found the matter was ripe for consideration by the federal appellate court. The defendants filed a petition for permission to appeal to the Fifth Circuit. The Fifth Circuit accepted review. After the Fifth Circuit accepted review, plaintiffs filed a motion to remand the case to state court, asserting that the class claims on which federal jurisdiction was premised have now effectively been dismissed as a result of a ruling in a related case. The Fifth Circuit denied the motion for remand, without prejudice to plaintiffs' right to refile the motion for remand after the Fifth Circuit disposes of the pending appeal. On July 28, 2010, the Fifth Circuit issued an order stating that since there is no controlling Louisiana Supreme Court precedent on the issue of whether an insurance policy's anti-assignment clause prohibits post-loss assignments, the Fifth Circuit is certifying that issue to the Louisiana Supreme Court. On

May 10, 2011, the Louisiana Supreme Court issued its ruling, holding that the contractual prohibition on post-loss assignments does not violate public policy, and that parties can contract to prohibit post-loss assignments. However, the Court went on to hold that the contract language must clearly and unambiguously express that the non-assignment clause applies to post-loss assignments. The Supreme Court refused to evaluate the language of the various policies before it. Rather, the Court stated that it is necessary for the federal court to evaluate the relevant anti-assignment clauses on a policy-by-policy basis to determine whether the language is sufficient to prohibit post-loss assignment. The case will now be sent back to the Fifth Circuit for a determination of whether each carrier's anti-assignment clause is sufficient to prohibit post-loss assignment.

The Company believes that its adjusting practices and processes in connection with Katrina homeowners claims were sound and in accordance with industry standards and state law. Each of the claims involved is fact-specific and requires independent analysis. There remain significant questions of Louisiana law that have yet to be decided, including the enforceability of the Company's anti-assignment clause and certain statute of limitation (prescription) issues. Based on recent rulings by the Louisiana Supreme Court, which have been construed to extend the time period within which Katrina actions can be filed, new individual cases continue to be filed. In addition, the State has yet to identify the specific claims that it contends are at issue in the Road Home Class Action, or the alleged deficiencies in adjusting those claims. There are many potential individual claims at issue in this litigation, each of which will require individual analysis, and a number of which may be subject to individual defenses, including release, accord and satisfaction, prescription, waiver and estoppel. There has been no factual development or discovery in connection with the Road Home Class Action. The motions to dismiss have been pending since the inception of the case. No answers have been filed, and the case remains stayed until the current appeal is concluded. Moreover, the State has indicated that it intends to drop its class allegations and seek a remand to state court, which is considered to be a less favorable forum, but the dismissal of the class allegations and prosecution by the State solely on its own behalf as assignee, may on the other hand adversely impact the State's ability to recover exemplary damages or penalties that might otherwise be sought on the underlying claim. In the Company's judgment a loss is not probable.

The Company has settled on a 48-state basis a nationwide class action alleging that it failed to properly pay general contractors overhead and profit on many homeowners structural loss claims. General contractors' overhead and profit is an amount that is added to payments on claims where the services of a general contractor are reasonably likely to be required. To a large degree, this lawsuit mirrored similar lawsuits filed against other carriers in the industry, some of which have settled. A class was certified for settlement purposes only, and the settlement received preliminary approval from the court on December 6, 2010 and final approval on May 6, 2011. No appeal was taken from the final approval order and the settlement is now final. Processing of class members' claims made as part of this settlement is underway. The \$75 million settlement was accrued as a prior year reserve reestimate in property-liability insurance claims and claims expense in 2010.

Allstate has been vigorously defending a lawsuit in regards to certain claims employees involving worker classification issues. This lawsuit is a certified class action challenging a state wage and hour law. In this case, plaintiffs sought actual damages in an amount to be proven at trial, liquidated damages in an amount equal to an unspecified percentage of the aggregate underpayment of wages to be proven at trial, as well as attorneys' fees and costs. Plaintiffs have not made a settlement demand nor have they alleged the amount of damages with any specificity.\_The case was bifurcated between liability and damages and is currently focused only on liability issues. No discovery has taken place regarding plaintiffs' alleged damages. In December 2009, the liability phase of the case was tried, and, on July 6, 2010, the court issued its decision finding in favor of Allstate on all claims. The plaintiffs have appealed the

42

decision in favor of Allstate to the first level appellate court. After concluding the current appeal, the parties may seek a subsequent appeal to the Illinois Supreme Court. Only liability issues are being addressed on appeal and no damages may be awarded at this stage of the proceedings. In the event the trial court's order were to be overturned, however, the parties would need to conduct damages discovery, and a trial on damages would have to take place, before any damages could be awarded. In the Company's judgment a loss is not probable.

# Other proceedings

The Company is defending certain matters relating to the Company's agency program reorganization announced in 1999. Although these cases have been pending for many years, they currently are in the early stages of litigation because of appellate court proceedings and threshold procedural issues.

These matters include a lawsuit filed in 2001 by the U.S. Equal Employment Opportunity Commission ("EEOC") alleging retaliation under federal civil rights laws ("EEOC I") and a class action filed in 2001 by former employee agents alleging retaliation and age discrimination under the Age Discrimination in Employment Act ("ADEA"), breach of contract and ERISA violations ("Romero I"). In 2004, in the consolidated EEOC I and Romero I litigation, the trial court issued a memorandum and order that, among other things, certified classes of agents, including a mandatory class of agents who had signed a release, for purposes of effecting the court's declaratory judgment that the release was voidable at the option of the release signer. The court also ordered that an agent who voided the release must return to Allstate "any and all benefits received by the [agent] in exchange for signing the release." The court also stated that, "on the undisputed facts of record, there is no basis for claims of age discrimination." The EEOC and plaintiffs asked the court to clarify and/or reconsider its memorandum and order and in January 2007, the judge denied their request. In June 2007, the court reversed its prior ruling that the release was voidable and granted the Company's motions for summary judgment, ruling that the asserted claims were barred by the release signed by most plaintiffs. Plaintiffs filed a notice of appeal with the U.S. Court of Appeals for the Third Circuit ("Third Circuit"). In July 2009, the Third Circuit vacated the trial court's entry of summary judgment in the Company's favor and remanded the cases to the trial court for additional discovery, including additional discovery related to the validity of the release and waiver. In its opinion, the Third Circuit held that if the release and waiver is held to be valid, then all of the claims in Romero I and EEOC I are barred. Thus, if the waiver and release is upheld, then only the claims in Romero I asserted by the small group of employee agents who did not sign the release and waiver would remain for adjudication. In January 2010, following the remand, the cases were assigned to a new judge for further proceedings in the trial court. Plaintiffs filed their Second Amended Complaint on July 28, 2010. Plaintiffs seek broad but unspecified "make whole relief," including back pay, compensatory and punitive damages, liquidated damages, lost investment capital, attorneys' fees and costs, and equitable relief, including reinstatement to employee agent status with all attendant benefits for up to approximately 6,500 former employee agents. Despite the length of time that these matters have been pending, to date only limited discovery has occurred related to the damages claimed by individual plaintiffs, and no damages discovery has occurred related to the claims of the putative class. Nor have plaintiffs provided any calculations of the putative class's alleged back pay or the alleged liquidated, compensatory or punitive damages, instead asserting that such calculations will be provided at a later stage during expert discovery. Damage claims are subject to reduction by amounts and benefits received by plaintiffs and putative class members subsequent to their employment termination. Little to no discovery has occurred with respect to amounts earned or received by plaintiffs and putative class members in mitigation of their alleged losses. Alleged damage amounts and lost benefits of the approximately 6,500 putative class members also are subject to individual variation and determination dependent upon retirement dates, participation in employee benefit programs, and years of service. Discovery limited to the validity of the waiver and release is in process. At present, no class is certified. Summary judgment proceedings on the validity of the waiver and release are expected to occur in early 2012.

A putative nationwide class action has also been filed by former employee agents alleging various violations of ERISA, including a worker classification issue ("Romero II"). These plaintiffs are challenging certain amendments to the Agents Pension Plan and are seeking to have exclusive agent independent contractors treated as employees for benefit purposes. Romero II was dismissed with prejudice by the trial court, was the subject of further proceedings on appeal, and was reversed and remanded to the trial court in 2005. In June 2007, the court granted the Company's motion to dismiss the case. Plaintiffs filed a notice of appeal with the Third Circuit. In July 2009, the Third Circuit vacated the

district court's dismissal of the case and remanded the case to the trial court for additional discovery, and directed that the case be reassigned to another trial court judge. In its opinion, the Third Circuit held that if the release and waiver is held to be valid, then one of plaintiffs' three claims asserted in Romero II is barred. The Third Circuit directed the district court to consider on remand whether the other two claims asserted in Romero II are barred by the release and waiver. In January 2010, following the remand, the case was assigned to a new judge (the same judge for the Romero I and EEOC I cases) for further proceedings in the trial court. On April 23, 2010, plaintiffs filed their First Amended Complaint. Plaintiffs seek broad but unspecified "make whole" or other equitable relief, including losses of income and benefits as a result of their decision to retire from the Company between November 1, 1999 and December 31, 2000. They also seek repeal of the challenged amendments to the Agents Pension Plan with all attendant benefits revised and recalculated for thousands of former employee agents, and attorney's fees and costs. Despite the length of time that this matter has been pending, to date only limited discovery has occurred related to the damages claimed by individual plaintiffs, and no damages discovery has occurred related to the claims of the putative class. Nor have plaintiffs provided any calculations of the putative class's alleged losses, instead asserting that such calculations will be provided at a later stage during expert discovery. Damage claims are subject to reduction by amounts and benefits received by plaintiffs and putative class members subsequent to their employment termination. Little to no discovery has occurred with respect to amounts earned or received by plaintiffs and putative class members in mitigation of their alleged losses. Alleged damage amounts and lost benefits of the approximately 6,500 putative class members also are subject to individual variation and determination dependent upon retirement dates, participation in employee benefit programs, and years of service. As in Romero I and EEOC I, discovery at this time is limited to issues relating to the validity of the waiver and release. Class certification has not been decided. Summary judgment proceedings on the validity of the waiver and release are expected to occur in early 2012.

In these agency program reorganization matters, the threshold issue of the validity and scope of the waiver and release is yet to be decided and, if decided in favor of the Company, would preclude any damages being awarded in Romero I and EEOC I and may also preclude damages from being awarded in Romero II. In the Company's judgment a loss is not probable. Allstate has been vigorously defending these lawsuits and other matters related to its agency program reorganization.

#### **Other Matters**

Various other legal, governmental, and regulatory actions, including state market conduct exams, and other governmental and regulatory inquiries are currently pending that involve the Company and specific aspects of its conduct of business. Like other members of the insurance industry, the Company is the target of a number of class action lawsuits and other types of proceedings, some of which involve claims for substantial or indeterminate amounts. These actions are based on a variety of issues and target a range of the Company's practices. The outcome of these disputes is currently unpredictable; however it is not likely, they will have a material effect on the consolidated financial statements of the Company.

#### Asbestos and environmental

Allstate's reserves for asbestos claims were \$1.09 billion and \$1.10 billion, net of reinsurance recoverables of \$529 million and \$555 million, as of June 30, 2011 and December 31, 2010, respectively. Reserves for environmental claims were \$192 million and \$201 million, net of reinsurance recoverables of \$44 million and \$47 million, as of June 30, 2011 and December 31, 2010, respectively. Approximately 61% and 60% of the total net asbestos and environmental reserves as of June 30, 2011 and December 31, 2010, respectively, were for incurred but not reported estimated losses.

Management believes its net loss reserves for asbestos, environmental and other discontinued lines exposures are appropriately established based on available facts, technology, laws and regulations. However, establishing net loss reserves for asbestos, environmental and other discontinued lines claims is subject to uncertainties that are much greater than those presented by other types of claims. The ultimate cost of losses may vary materially from recorded amounts, which are based on management's best estimate. Among the complications are lack of historical data, long reporting delays, uncertainty as to the number and identity of insureds with potential exposure and unresolved legal issues regarding policy coverage; unresolved legal issues regarding the determination, availability and timing of exhaustion of policy limits; plaintiffs' evolving and expanding theories of liability; availability and collectability of recoveries from reinsurance; retrospectively determined premiums and other contractual agreements; estimates of

44

the extent and timing of any contractual liability; the impact of bankruptcy protection sought by various asbestos producers and other asbestos defendants; and other uncertainties. There are also complex legal issues concerning the interpretation of various insurance policy provisions and whether those losses are covered, or were ever intended to be covered, and could be recoverable through retrospectively determined premium, reinsurance or other contractual agreements. Courts have reached different and sometimes inconsistent conclusions as to when losses are deemed to have occurred and which policies provide coverage; what types of losses are covered; whether there is an insurer obligation to defend; how policy limits are determined; how policy exclusions and conditions are applied and interpreted; and whether clean-up costs represent insured property damage. Management believes these issues are not likely to be resolved in the near future, and the ultimate costs may vary materially from the amounts currently recorded resulting in material changes in loss reserves. In addition, while the Company believes that improved actuarial techniques and databases have assisted in its ability to estimate asbestos, environmental, and other discontinued lines net loss reserves, these refinements may subsequently prove to be inadequate indicators of the extent of probable losses. Due to the uncertainties and factors described above, management believes it is not practicable to develop a meaningful range for any such additional net loss reserves that may be required.

#### 11. Components of Net Periodic Pension and Postretirement Benefit Costs

The components of net periodic cost for the Company's pension and postretirement benefit plans are as follows:

(\$ in millions)		Three m Ju		Six months ended June 30,				
	_	2011	_	2010	_	2011	_	2010
Pension benefits								
Service cost	\$	38	\$	37	\$	76	\$	75
Interest cost		80		80		161		160
Expected return on plan assets		(92)		(82)		(184)		(165)
Amortization of:								
Prior service credit		(1)				(1)		(1)
Net actuarial loss		39		39		77		79
Settlement loss		8		13		17		26
Net periodic pension cost	\$	72	\$	87	\$	146	\$	174
Postretirement benefits								
Service cost	\$	3	\$	3	\$	6	\$	6
Interest cost		9		10		18		20
Amortization of:								
Prior service credit		(5)		(5)		(11)		(11)
Net actuarial gain		(8)		(6)		(15)		(11)
Net periodic postretirement benefit cost	\$	(1)	\$	2	\$ _	(2)	\$	4
		45						

# 12. Business Segments

Summarized revenue data for each of the Company's business segments are as follows:

(\$ in millions)	Three months ended June 30,					Six months ended June 30,					
		2011		2010	_	2011		2010			
Revenues					_		_	_			
Property-Liability											
Property-liability insurance premiums											
Standard auto	\$	4,093	\$	4,154	\$	8,181	\$	8,291			
Non-standard auto		206		230		417		464			
Total auto		4,299		4,384	_	8,598		8,755			
Homeowners		1,548		1,512		3,087		3,028			
Other personal lines		610		617		1,221		1,233			
Allstate Protection		6,457		6,513	_	12,906		13,016			
Discontinued Lines and Coverages						(1)					
Total property-liability insurance premiums		6,457		6,513		12,905		13,016			
Net investment income		310		310		594		614			
Realized capital gains and losses		(8)		(106)		49		(296)			
Total Property-Liability		6,759		6,717		13,548		13,334			
Allstate Financial											
Life and annuity premiums and contract charges											
Traditional life insurance		109		104		217		210			
Immediate annuities with life contingencies		15		31		58		58			
Accident and health insurance		162		151		323		307			
Total life and annuity premiums		286		286	_	598		575			
Interest-sensitive life insurance		253		249		501		491			
Fixed annuities		8		10		17		23			
Total contract charges		261		259	_	518		514			
Total life and annuity premiums and contract charges		547		545		1,116		1,089			
Net investment income		694		723		1,378		1,454			
Realized capital gains and losses		62		(353)		101		(515)			
Total Allstate Financial		1,303		915		2,595		2,028			
Corporate and Other											
Service fees		2		3		4		6			
Net investment income		16		16		30		31			
Realized capital gains and losses		3		8		3		12			
Total Corporate and Other before reclassification of service fees		21		27	_	37		49			
Reclassification of service fees (1)		(2)		(3)		(4)		(6)			
Total Corporate and Other		19		24		33		43			
Consolidated revenues	\$	8,081	\$	7,656	\$	16,176	\$	15,405			

<sup>(1)</sup> For presentation in the Condensed Consolidated Statements of Operations, service fees of the Corporate and Other segment are reclassified to operating costs and expenses.

 $Summarized\ financial\ performance\ data\ for\ each\ of\ the\ Company's\ reportable\ segments\ are\ as\ follows:$ 

(\$ in millions)		Three m Ju	onths ne 30			Six months ended June 30,					
		2011		2010		2011		2010			
Net income							_				
Property-Liability											
Underwriting (loss) income											
Allstate Protection	\$	(1,498)	\$	209	\$	(1,165)	\$	284			
Discontinued Lines and Coverages		(4)		(2)		(10)		(6)			
Total underwriting (loss) income		(1,502)		207		(1,175)		278			
Net investment income		310		310		594		614			
Income tax benefit (expense) on operations		460		(149)		279		(237)			
Realized capital gains and losses, after-tax		(6)		(69)		32		(192)			
Property-Liability net (loss) income		(738)	<u></u>	299		(270)		463			
Allstate Financial											
Life and annuity premiums and contract charges		547		545		1,116		1,089			
Net investment income		694		723		1,378		1,454			
Periodic settlements and accruals on non-hedge derivative financial											
instruments		19		11		36		28			
Contract benefits and interest credited to contractholder funds		(834)		(935)		(1,713)		(1,840)			
Operating costs and expenses and amortization of deferred policy											
acquisition costs		(213)		(157)		(435)		(335)			
Restructuring and related charges				1		2		1			
Income tax expense on operations		(72)		(63)		(127)		(133)			
Operating income		141		125		257		264			
Realized capital gains and losses, after-tax		40		(230)		65		(335)			
Valuation changes on embedded derivatives that are not hedged, after-tax		(3)				5					
DAC and DSI (amortization) accretion related to realized capital gains and											
losses and valuation changes on embedded derivatives that are not											
hedged, after-tax		(5)		4		(31)		2			
DAC and DSI unlocking related to realized capital gains and losses, after-											
tax						1		(18)			
Reclassification of periodic settlements and accruals on non-hedge											
financial instruments, after-tax		(11)		(7)		(23)		(18)			
Gain (loss) on disposition of operations, after-tax		4		1		(11)		2			
Allstate Financial net income (loss)		166		(107)		263		(103)			
Corporate and Other											
Service fees (1)		2		3		4		6			
Net investment income		16		16		30		31			
Operating costs and expenses (1)		(100)		(104)		(193)		(204)			
Income tax benefit on operations		32		33		63		65			
Operating loss		(50)		(52)		(96)	_	(102)			
Realized capital gains and losses, after-tax		2		5		2		7			
Corporate and Other net loss		(48)	_	(47)		(94)		(95)			
Consolidated net (loss) income	\$	(620)	\$	145	\$	(101)	\$	265			
	~—	()	_		_	\ - /	-				

<sup>(1)</sup> For presentation in the Condensed Consolidated Statements of Operations, service fees of the Corporate and Other segment are reclassified to operating costs and expenses.

47

# 13. Other Comprehensive Income

The components of other comprehensive (loss) income on a pre-tax and after-tax basis are as follows:

(\$ in millions)	Three months ended June 30,												
,	2011						2010						
					After-						After-		
	Pre-tax		Tax	tax Pre-tax		Tax		tax					
Unrealized net holding gains and losses													
arising during the period, net of													
related offsets	\$ 676	\$	(237)	\$	439	\$	492	\$	(172)	\$	320		
Less: reclassification adjustment of													
realized capital gains and losses	111		(39)		72		(142)		50		(92)		
Unrealized net capital gains and losses	 565		(198)		367		634		(222)		412		
Unrealized foreign currency translation													
adjustments	7		(3)		4		(27)		10		(17)		
Unrecognized pension and other	24		(7)		17		31		(10)		21		

Net (loss) income Comprehensive (loss) income		-		<u></u>	(620) (232)			 	•	145 561
Comprehensive (1055) income				Ψ	Six months	ended I	une 30		Ψ	301
			2011		OIX MONTHS	chaca s	une 50,	2010		
					After-					After-
	Pre-tax		Tax		tax		Pre-tax	Tax		tax
Unrealized net holding gains and losses arising during the period, net of										
related offsets	\$ 1,044	\$	(366)	\$	678	\$	1,574	\$ (550)	\$	1,024
Less: reclassification adjustment of										
realized capital gains and losses	257		(90)		167		(268)	 94		(174)
Unrealized net capital gains and losses	787		(276)		511		1,842	(644)		1,198
Unrealized foreign currency translation										
adjustments	22		(8)		14		(5)	2		(3)
Unrecognized pension and other	45		(4.5)		22			(10)		20
postretirement benefit cost	 47		(15)	_	32		57	 (19)	_	38
Other comprehensive income	\$ 856	\$	(299)		557	\$	1,894	\$ (661)		1,233
Net (loss) income					(101)					265
Comprehensive income				\$	456				\$	1,498
				48						

388

416

#### REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors and Shareholders of The Allstate Corporation Northbrook, IL 60062

postretirement benefit cost
Other comprehensive income

We have reviewed the accompanying condensed consolidated statement of financial position of The Allstate Corporation and subsidiaries (the "Company") as of June 30, 2011, and the related condensed consolidated statements of operations for the three-month and six-month periods ended June 30, 2011 and 2010, and the condensed consolidated statements of cash flows for the six-month periods ended June 30, 2011 and 2010. These interim financial statements are the responsibility of the Company's management.

We conducted our reviews in accordance with the standards of the Public Company Accounting Oversight Board (United States). A review of interim financial information consists principally of applying analytical procedures and making inquiries of persons responsible for financial and accounting matters. It is substantially less in scope than an audit conducted in accordance with the standards of the Public Company Accounting Oversight Board (United States), the objective of which is the expression of an opinion regarding the financial statements taken as a whole. Accordingly, we do not express such an opinion.

Based on our reviews, we are not aware of any material modifications that should be made to such condensed consolidated interim financial statements for them to be in conformity with accounting principles generally accepted in the United States of America.

We have previously audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the consolidated statement of financial position of The Allstate Corporation and subsidiaries as of December 31, 2010, and the related consolidated statements of operations, comprehensive income, shareholders' equity, and cash flows for the year then ended (not presented herein); and in our report dated February 24, 2011, which report includes an explanatory paragraph relating to a change in the Company's recognition and presentation for other-than-temporary impairments of debt securities in 2009, we expressed an unqualified opinion on those consolidated financial statements. In our opinion, the information set forth in the accompanying condensed consolidated statement of financial position as of December 31, 2010 is fairly stated, in all material respects, in relation to the consolidated statement of financial position from which it has been derived.

/s/ Deloitte & Touche LLP

Chicago, Illinois July 29, 2011

49

# Item 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS FOR THE THREE-MONTH AND SIX-MONTH PERIODS ENDED JUNE 30, 2011 AND 2010

# **OVERVIEW**

The following discussion highlights significant factors influencing the consolidated financial position and results of operations of The Allstate Corporation (referred to in this document as "we," "our," "us," the "Company" or "Allstate"). It should be read in conjunction with the condensed consolidated financial statements and notes thereto found under Part I. Item 1. contained herein, and with the discussion, analysis, consolidated financial statements and notes thereto in Part I. Item 1. and Part II. Item 7. and Item 8. of The Allstate Corporation Annual Report on Form 10-K for 2010. Further analysis of our insurance segments is provided in the Property-Liability Operations (which includes the Allstate Protection and the Discontinued Lines and Coverages segments) and in the Allstate Financial Segment sections of Management's Discussion and Analysis ("MD&A"). The segments are consistent with the way in which we use financial information to evaluate business performance and to determine the allocation of resources.

Allstate is focused on three priorities:

- · improve our operating results;
- · grow our businesses profitably; and
- differentiate ourselves from the competition by reinventing our business.

#### HIGHLIGHTS

- Consolidated net loss was \$620 million in the second quarter of 2011 compared to net income of \$145 million in the second quarter of 2010, and net loss was \$101 million in the first six months of 2011 compared to net income of \$265 million in the first six months of 2010. Net loss per diluted share was \$1.19 in the second quarter of 2011 compared to net income per diluted share of \$0.27 in the second quarter of 2010, and net loss per diluted share was \$0.19 in the first six months of 2011 compared to net income per diluted share of \$0.49 in the first six months of 2010.
- · Property-Liability net loss was \$738 million in the second quarter of 2011 compared to net income of \$299 million in the second quarter of 2010, and net loss was \$270 million in the first six months of 2011 compared to net income of \$463 million in the first six months of 2010.
- The Property-Liability combined ratio was 123.3 in the second quarter of 2011 compared to 96.8 in the second quarter of 2010, and 109.1 in the first six months of 2011 compared to 97.9 in the first six months of 2010.
- Allstate Financial had net income of \$166 million in the second quarter of 2011 compared to a net loss of \$107 million in the second quarter of 2010, and net income of \$263 million in the first six months of 2011 compared to a net loss of \$103 million in the first six months of 2010.
- Total revenues were \$8.08 billion in the second quarter of 2011 compared to \$7.66 billion in the second quarter of 2010, and \$16.18 billion in the first six months of 2011 compared to \$15.41 billion in the first six months of 2010.
- · Property-Liability premiums earned in the second quarter of 2011 totaled \$6.46 billion, a decrease of 0.9% from \$6.51 billion in the second quarter of 2010, and \$12.91 billion in the first six months of 2011, a decrease of 0.9% from \$13.02 billion in the first six months of 2010.
- Net realized capital gains were \$57 million in the second quarter of 2011 compared to net realized capital losses of \$451 million in the second quarter of 2010, and net realized capital gains were \$153 million in the first six months of 2011 compared to net realized capital losses of \$799 million in the first six months of 2010.
- · Investments as of June 30, 2011 totaled \$99.29 billion, a decrease of 1.2% from \$100.48 billion as of December 31, 2010. Net investment income in the second quarter of 2011 was \$1.02 billion, a decrease of 2.8% from \$1.05 billion in the second quarter of 2010, and \$2.00 billion in the first six months of 2011, a decrease of 4.6% from \$2.10 billion in the first six months of 2010.
- Book value per diluted share (ratio of shareholders' equity to total shares outstanding and dilutive potential shares outstanding) was \$35.95 as of June 30, 2011, an increase of 8.2% from \$33.24 as of June 30, 2010 and an increase of 1.8% from \$35.32 as of December 31, 2010.
- For the twelve months ended June 30, 2011, return on the average of beginning and ending period shareholders' equity was 3.1%, a decrease of 3.0 points from 6.1% for the twelve months ended June 30, 2010.
- · As of June 30, 2011, we had \$18.76 billion in shareholders' equity. This total included \$3.49 billion in deployable invested assets at the parent holding company level.

50

On May 17, 2011, we entered into a definitive agreement with White Mountains Insurance Group, Ltd. to purchase certain entities making up the Esurance and Answer Financial groups of companies for \$700 million plus the tangible book value of the entities acquired at close. The total price is expected to be about \$1 billion. The transaction is expected to close in the fall of 2011. The transaction is subject to regulatory approvals and other customary closing conditions, including review by antitrust authorities and state regulators.

# CONSOLIDATED NET (LOSS) INCOME

Net (loss) income

(\$ in millions)		Three mo	onths ne 30,	ended	Six months ended June 30,					
		2011		2010		2011		2010		
Revenues					_		_			
Property-liability insurance premiums	\$	6,457	\$	6,513	\$	12,905	\$	13,016		
Life and annuity premiums and contract charges		547		545		1,116		1,089		
Net investment income		1,020		1,049		2,002		2,099		
Realized capital gains and losses:										
Total other-than-temporary impairment losses		(82)		(288)		(238)		(538)		
Portion of loss recognized in other comprehensive income		(4)		(18)		(31)		(23)		
Net other-than-temporary impairment losses recognized in earnings		(86)		(306)	_	(269)		(561)		
Sales and other realized capital gains and losses		143		(145)		422		(238)		
Total realized capital gains and losses		57		(451)	_	153	_	(799)		
Total revenues	_	8,081		7,656	_	16,176	_	15,405		
Costs and expenses										
Property-liability insurance claims and claims expense		(6,355)		(4,714)		(10,831)		(9,506)		
Life and annuity contract benefits		(422)		(485)		(876)		(927)		
Interest credited to contractholder funds		(417)		(450)		(835)		(913)		
Amortization of deferred policy acquisition costs		(1,018)		(949)		(2,069)		(1,963)		
Operating costs and expenses		(802)		(789)		(1,640)		(1,618)		
Restructuring and related charges		(11)		(13)		(20)		(24)		
Interest expense		(91)		(92)		(183)		(184)		
Total costs and expenses		(9,116)	_	(7,492)	_	(16,454)	_	(15,135)		
Gain (loss) on disposition of operations		6		2		(17)		3		
Income tax benefit (expense)		409		(21)		194		(8)		

145

(101)

Property-Liability
Allstate Financial
Corporate and Other
Net (loss) income

\$ (738)	\$ 299	\$ (270)	\$ 463
166	(107)	263	(103)
(48)	(47)	(94)	(95)
\$ (620)	\$ 145	\$ (101)	\$ 265

51

#### PROPERTY-LIABILITY HIGHLIGHTS

- Premiums written, an operating measure that is defined and reconciled to premiums earned in the Property-Liability Operations section of the MD&A, decreased 0.4% to \$6.61 billion in the second quarter of 2011 from \$6.64 billion in the second quarter of 2010, and decreased 0.6% to \$12.83 billion in the first six months of 2011 from \$12.90 billion in the first six months of 2010.
  - Allstate brand standard auto premiums written decreased 0.9% to \$3.91 billion in the second quarter of 2011 from \$3.95 billion in the second quarter of 2010, and 1.0% to \$7.90 billion in the first six months of 2011 from \$7.97 billion in the first six months of 2010.
  - Allstate brand homeowners premiums written increased 2.6% to \$1.61 billion in the second quarter of 2011 from \$1.57 billion in the second quarter of 2010, and increased 2.8% to \$2.83 billion in the first six months of 2011 from \$2.75 billion in the first six months of 2010.
  - Encompass brand premiums written decreased 5.2% to \$273 million in the second quarter of 2011 from \$288 million in the second quarter of 2010, and 6.0% to \$518 million in the first six months of 2011 from \$551 million in the first six months of 2010.
- · Premium operating measures and statistics contributing to overall Allstate brand standard auto premiums written decrease were the following:
  - 0.6% decrease in policies in force ("PIF") as of June 30, 2011 compared to June 30, 2010
  - 0.5% decrease in the six month policy term average gross premium before reinsurance to \$442 in the second quarter of 2011 from \$444 in the second quarter of 2010, and 0.7% decrease in the six month policy term average gross premium before reinsurance to \$441 in the first six months of 2011 from \$444 in the first six months of 2010
  - 0.2 point increase in the six month renewal ratio to 89.2% in the second quarter of 2011 compared to 89.0% in the second quarter of 2010, and a 0.1 point increase in the six month renewal ratio to 89.0% in the first six months of 2011 compared to 88.9% in the first six months of 2010
  - 5.2% decrease and 3.0% increase in new issued applications in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
- Premium operating measures and statistics contributing to overall Allstate brand homeowners premiums written increase were the following:
  - 3.9% decrease in PIF as of June 30, 2011 compared to June 30, 2010
  - 6.0% increase in the twelve month policy term average gross premium before reinsurance to \$989 in the second quarter of 2011 from \$933 in the second quarter of 2010, and 6.0% increase in the twelve month policy term average gross premium before reinsurance to \$983 in the first six months of 2011 from \$927 in the first six months of 2010
  - 0.1 point increase in the twelve month renewal ratio to 88.4% in the second quarter of 2011 compared to 88.3% in the second quarter of 2010, and 0.1 increase in the twelve month renewal ratio to 88.3% in the first six months of 2011 compared to 88.2% in the first six months of 2010
  - 18.5% and 12.2% decrease in new issued applications in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
  - \$9 million decrease in catastrophe reinsurance costs to \$127 million in the second quarter of 2011 from \$136 million in the second quarter of 2010, and \$20 million decrease in catastrophe reinsurance costs to \$251 million in the first six months of 2011 from \$271 million in the first six months of 2010
- Factors comprising the Allstate brand standard auto loss ratio increase of 3.1 points to 73.2 in the second quarter of 2011 from 70.1 in the second quarter of 2010, and an increase of 1.9 points to 71.7 in the first six months of 2011 from 69.8 in the first six months of 2010 were the following:
  - 4.7 point and 2.3 point increase in the effect of catastrophe losses in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
  - 3.9% and 1.4% decrease in standard auto claim frequency (rate of claim occurrence per policy in force) for property damage in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
  - 2.3% decrease and 0.3% increase in standard auto claim frequency for bodily injury in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
  - 1.1% and 1.0% increase in auto paid claim severities (average cost per claim) for property damage in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010

52

- 0.4% and 2.0% increase in auto paid claim severities for bodily injury in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
- Factors comprising the Allstate brand homeowners loss ratio, which includes catastrophes, increase of 88.5 points to 171.1 in the second quarter of 2011 from 82.6 in the second quarter of 2010, and a increase of 34.7 points to 119.7 in the first six months of 2011 from 85.0 in the first six months of 2010 were the following:
  - 88.5 point increase in the effect of catastrophe losses to 123.2 points in the second quarter of 2011 compared to 34.7 points in the second quarter of 2010, and 34.7 point increase in the effect of catastrophe losses to 70.6 points in the first six months of 2011 compared to 35.9 points in the first six months of 2010
  - 0.8% decrease and 0.4% increase in homeowner claim frequency, excluding catastrophes, in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
  - 3.4% and 3.5% increase in paid claim severity, excluding catastrophes, in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010
- Factors comprising the \$1.70 billion increase in catastrophe losses to \$2.34 billion in the second quarter of 2011 compared to \$636 million in the second quarter of 2010, and \$1.39 billion increase to \$2.67 billion in the first six months of 2011 compared to \$1.28 billion in the first six months of 2010 were the following:
  - 33 events with losses of \$2.34 billion in the second quarter of 2011 compared to 30 events with losses of \$758 million in the second quarter of 2010, and 49 events with losses of \$2.72 billion in the first six months of 2011 compared to 41 events with losses of \$1.38 billion in the first six months of

- \$17 million favorable prior year reserve reestimates in the second quarter of 2011 compared to \$83 million favorable reserve reestimates in the second quarter of 2010, and \$51 million favorable reserve reestimates in the first six months of 2011 compared to \$98 million favorable reserve reestimates in the first six months of 2010
- Factors comprising prior year reserve reestimates of \$47 million favorable in the second quarter of 2011 compared to \$150 million favorable in the second quarter of 2010, and prior year reserve reestimates of \$88 million favorable in the first six months of 2011 compared to \$173 million favorable in the first six months of 2010 included:
  - prior year reserve reestimates related to auto, homeowners and other personal lines in the second quarter of 2011 contributed \$90 million favorable, \$3 million unfavorable and \$36 million unfavorable, respectively, compared to prior year reserve reestimates in the second quarter of 2010 of \$85 million favorable, \$61 million favorable and \$5 million favorable, respectively, and prior year reserve reestimates related to auto, homeowners and other personal lines in the first six months of 2011 contributed \$109 million favorable, \$35 million favorable and \$49 million unfavorable, respectively, compared to prior year reserve reestimates in the first six months of 2010 of \$80 million favorable, \$69 million favorable and \$27 million favorable, respectively
  - prior year reestimates in the second quarter and first six months of 2011 and 2010 are largely attributable to prior year catastrophes and severity development that was better than expected.
- Property-Liability underwriting loss was \$1.50 billion in the second quarter of 2011 compared to underwriting income of \$207 million in the second quarter of 2010, and Property-Liability underwriting loss was \$1.18 billion in the first six months of 2011 compared to underwriting income of \$278 million in the first six months of 2010. Underwriting income (loss), a measure not based on accounting principles generally accepted in the United States of America ("GAAP"), is defined below.
- · Net realized capital losses were \$8 million in the second quarter of 2011 compared to net realized capital losses of \$106 million in the second quarter of 2010, and net realized capital gains were \$49 million in the first six months of 2011 compared to net realized capital losses of \$296 million in the first six months of 2010.
- Property-Liability investments as of June 30, 2011 were \$36.12 billion, an increase of 3.1% from \$35.05 billion as of December 31, 2010. Net investment income of \$310 million in the second quarter of 2011 was comparable to the second quarter of 2010. Net investment income was \$594 million in the first six months of 2011, a decrease of 3.3% from \$614 million in the first six months of 2010.

53

#### PROPERTY-LIABILITY OPERATIONS

**Overview** Our Property-Liability operations consist of two business segments: Allstate Protection and Discontinued Lines and Coverages. Allstate Protection comprises two brands, the Allstate brand and Encompass® brand. Allstate Protection is principally engaged in the sale of personal property and casualty insurance, primarily private passenger auto and homeowners insurance, to individuals in the United States and Canada. Discontinued Lines and Coverages includes results from insurance coverage that we no longer write and results for certain commercial and other businesses in run-off. These segments are consistent with the groupings of financial information that management uses to evaluate performance and to determine the allocation of resources.

Underwriting income (loss), a measure that is not based on GAAP and is reconciled to net income (loss) below, is calculated as premiums earned, less claims and claims expense ("losses"), amortization of deferred policy acquisition costs ("DAC"), operating costs and expenses and restructuring and related charges, as determined using GAAP. We use this measure in our evaluation of results of operations to analyze the profitability of the Property-Liability insurance operations separately from investment results. It is also an integral component of incentive compensation. It is useful for investors to evaluate the components of income separately and in the aggregate when reviewing performance. Net income (loss) is the GAAP measure most directly comparable to underwriting income (loss). Underwriting income (loss) should not be considered as a substitute for net income and does not reflect the overall profitability of the business.

The table below includes GAAP operating ratios we use to measure our profitability. We believe that they enhance an investor's understanding of our profitability. They are calculated as follows:

- Claims and claims expense ("loss") ratio the ratio of claims and claims expense to premiums earned. Loss ratios include the impact of catastrophe
- · Expense ratio the ratio of amortization of DAC, operating costs and expenses, and restructuring and related charges to premiums earned.
- Combined ratio the ratio of claims and claims expense, amortization of DAC, operating costs and expenses, and restructuring and related charges to premiums earned. The combined ratio is the sum of the loss ratio and the expense ratio. The difference between 100% and the combined ratio represents underwriting income (loss) as a percentage of premiums earned, or underwriting margin.

We have also calculated the following impacts of specific items on the GAAP operating ratios because of the volatility of these items between fiscal periods.

- · Effect of catastrophe losses on combined ratio the percentage of catastrophe losses included in claims and claims expense to premiums earned. This ratio includes prior year reserve reestimates of catastrophe losses.
- Effect of prior year reserve reestimates on combined ratio the percentage of prior year reserve reestimates included in claims and claims expense to premiums earned. This ratio includes prior year reserve reestimates of catastrophe losses.
- · Effect of restructuring and related charges on combined ratio the percentage of restructuring and related charges to premiums earned.
- Effect of Discontinued Lines and Coverages on combined ratio the ratio of claims and claims expense and other costs and expenses in the Discontinued Lines and Coverages segment to Property-Liability premiums earned. The sum of the effect of Discontinued Lines and Coverages on the combined ratio and the Allstate Protection combined ratio is equal to the Property-Liability combined ratio.

Summarized financial data, a reconciliation of underwriting (loss) income to net (loss) income, and GAAP operating ratios for our Property-Liability operations are presented in the following table.

(\$ in millions, except ratios)		Three mo	onths one 30,	ended	Six months ended June 30,						
		2011		2010		2011		2010			
Premiums written	\$	6,611	\$	6,640	\$	12,826	\$	12,898			
Revenues											
Premiums earned	\$	6,457	\$	6,513	\$	12,905	\$	13,016			
Net investment income		310		310		594		614			
Realized capital gains and losses		(8)		(106)	_	49	_	(296)			
Total revenues		6,759		6,717		13,548		13,334			
Costs and expenses											
Claims and claims expense		(6,355)		(4,714)		(10,831)		(9,506)			
Amortization of DAC		(908)		(914)		(1,812)		(1,839)			
Operating costs and expenses		(685)		(664)		(1,415)		(1,368)			
Restructuring and related charges		(11)		(14)		(22)		(25)			
Total costs and expenses		(7,959)		(6,306)		(14,080)		(12,738)			
Income tax benefit (expense)		462		(112)		262		(133)			
Net (loss) income	\$	(738)	\$	299	\$	(270)	\$	463			
Underwriting (loss) income	\$	(1,502)	\$	207	\$	(1,175)	\$	278			
Net investment income		310		310		594		614			
Income tax benefit (expense) on operations		460		(149)		279		(237)			
Realized capital gains and losses, after-tax		(6)		(69)		32		(192)			
Net (loss) income	\$	(738)	\$ <u></u>	299	\$	(270)	\$	463			
Catastrophe losses (1)	\$	2,339	\$	636	\$	2,672	\$	1,284			
GAAP operating ratios											
Claims and claims expense ratio		98.4		72.4		83.9		73.1			
Expense ratio		24.9		24.4		25.2		24.8			
Combined ratio		123.3		96.8		109.1		97.9			
Effect of catastrophe losses on combined ratio (1)		36.2		9.8		20.7		9.9			
Effect of prior year reserve reestimates on combined ratio (1)		(0.7)		(2.3)		(0.7)		(1.3)			
Effect of restructuring and related charges on combined ratio		0.2		0.2		0.2		0.2			
Effect of Discontinued Lines and Coverages on combined ratio	===	0.1			_	0.1	_	0.1			
<del>-</del>			_		_		_				

<sup>(1)</sup> Prior year reserve reestimates included in catastrophe losses totaled \$17 million and \$51 million favorable in the three months and six months ended June 30, 2011, respectively, compared to \$83 million and \$98 million favorable in the three months and six months ended June 30, 2010, respectively.

**Premiums written**, an operating measure, is the amount of premiums charged for policies issued during a fiscal period. Premiums earned is a GAAP measure. Premiums are considered earned and are included in the financial results on a pro-rata basis over the policy period. The portion of premiums written applicable to the unexpired terms of the policies is recorded as unearned premiums on our Condensed Consolidated Statements of Financial Position.

55

A reconciliation of premiums written to premiums earned is shown in the following table.

(\$ in millions)		Three me			Six moi Ju			
	_	2011		2010	_	2011		2010
Premiums written:		,			_			
Allstate Protection	\$	6,611	\$	6,640	\$	12,827	\$	12,898
Discontinued Lines and Coverages						(1)		
Property-Liability premiums written	_	6,611		6,640	_	12,826	_	12,898
(Increase) decrease in unearned premiums		(165)		(110)		69		135
Other		11		(17)		10		(17)
Property-Liability premiums earned	\$	6,457	\$	6,513	\$	12,905	\$	13,016
Premiums earned:								
Allstate Protection	\$	6,457	\$	6,513	\$	12,906	\$	13,016
Discontinued Lines and Coverages						(1)		
Property-Liability	\$	6,457	\$	6,513	\$	12,905	\$	13,016

#### ALLSTATE PROTECTION SEGMENT

(\$ in millions)				Th	iree months	s end	ed June 30,				
	 Allstate brand				Encomp	orand		Allstate Protection			
	 2011		2010	_	2011		2010	_	2011		2010
Standard auto	\$ 3,911	\$	3,948	\$	154	\$	169	\$	4,065	\$	4,117
Non-standard auto	197		220				1		197		221
Homeowners	1,606		1,565		94		94		1,700		1,659
Other personal lines (1)	624		619		25		24		649		643
Total	\$ 6,338	\$	6,352	\$	273	\$	288	\$	6,611	\$	6,640

	_	Allstate brand				Encomp	orand		ection			
	_	2011	2011 2010		_	2011		2010		2011		2010
Standard auto	\$	7,895	\$	7,971	\$	298	\$	329	\$	8,193	\$	8,300
Non-standard auto		407		457		1		4		408		461
Homeowners		2,831		2,754		173		174		3,004		2,928
Other personal lines (1)		1,176		1,165		46		44		1,222		1,209
Total	\$	12,309	\$	12,347	\$	518	\$	551	\$	12,827	\$	12,898

<sup>(1)</sup> Other personal lines include commercial, condominium, renters, involuntary auto and other personal lines.

Allstate brand premiums written by the direct channel, excluding Allstate Canada, increased 10.5% to \$200 million in the second quarter of 2011 from \$181 million in the second quarter of 2010, and 10.9% to \$406 million in the first six months of 2011 from \$366 million in the first six months of 2010, reflecting an impact by profitability management actions taken in New York and Florida. The direct channel includes call centers and the internet.

56

Premiums earned by brand are shown in the following table.

(\$ in millions)	Three months ended June 30,

	_	Allstate brand				Encompass brand				Allstate Protection				
	<u> </u>	2011		2010	_	2011		2010		2011		2010		
Standard auto	\$	3,938	\$	3,969	\$	155	\$	185	\$	4,093	\$	4,154		
Non-standard auto		205		228		1		2		206		230		
Homeowners		1,457		1,416		91		96		1,548		1,512		
Other personal lines		587		592		23		25		610		617		
Total	\$	6,187	\$	6,205	\$	270	\$	308	\$	6,457	\$	6,513		

Six months ended June 30,

	Allstate brand				<b>Encompass brand</b>			Allstate Protection			
	 2011		2010	_	2011		2010		2011		2010
Standard auto	\$ 7,866	\$	7,912	\$	315	\$	379	\$	8,181	\$	8,291
Non-standard auto	415		458		2		6		417		464
Homeowners	2,905		2,832		182		196		3,087		3,028
Other personal lines	1,175		1,184		46		49		1,221		1,233
Total	\$ 12,361	\$	12,386	\$	545	\$	630	\$_	12,906	\$	13,016

Premium operating measures and statistics that are used to analyze the business are calculated and described below. Measures and statistics presented for Allstate brand exclude Allstate Canada, loan protection and specialty auto.

- · PIF: Policy counts are based on items rather than customers. A multi-car customer would generate multiple item (policy) counts, even if all cars were insured under one policy.
- · Average premium-gross written: Gross premiums written divided by issued item count. Gross premiums written include the impacts from discounts and surcharges, and exclude the impacts from mid-term premium adjustments, ceded reinsurance premiums, and premium refund accruals. Allstate brand average gross premiums represent the appropriate policy term for each line, which is 6 months for standard and non-standard auto and 12 months for homeowners. Encompass brand average gross premiums represent the appropriate policy term for each line, which is 12 months for standard auto and homeowners and 6 months for non-standard auto.
- · Renewal ratio: Renewal policies issued during the period, based on contract effective dates, divided by the total policies issued 6 months prior for standard and non-standard auto (12 months prior for Encompass brand standard auto) or 12 months prior for homeowners.
- · New issued applications: Item counts of automobiles or homeowners insurance applications for insurance policies that were issued during the period. Does not include automobiles that are added by existing customers.

57

*Standard auto premiums written* totaled \$4.07 billion in the second quarter of 2011, a decrease of 1.3% from \$4.12 billion in the second quarter of 2010, and \$8.19 billion in the first six months of 2011, a decrease of 1.3% from \$8.30 billion in the first six months of 2010.

	Allstat	te brand	Encompa	ss brand
Standard Auto	2011	2010	2011	2010

Three months ended June 30,						
PIF (thousands)	17,420		17,529		672	750
Average premium-gross written (1)	\$ 442	\$	444	\$	938	\$ 989
Renewal ratio (%) (1)	89.2		89.0		70.2	66.9
Approved rate changes (2):						
# of states	18	(8)	32	(6) (7)	3	10
Countrywide (%) (3)	1.9		0.2		0.3	(0.1)
State specific (%) (4) (5)	5.3		0.5		4.0	(0.5)
Six months ended June 30,						
PIF (thousands)	17,420		17,529		672	750
Average premium-gross written (1)	\$ 441	\$	444	\$	945	\$ 993
Renewal ratio (%) (1)	89.0		88.9		70.7	67.8
Approved rate changes (2):		(0)		_		
# of states (6)	24	(8)	34	(7)	6	14
Countrywide (%) (3)	3.1		0.5		0.9	1.4
State specific (%) (4) (5)	5.8		1.1		4.6	3.2

<sup>(1)</sup> Policy term is six months for Allstate brand and twelve months for Encompass brand.

Allstate brand standard auto premiums written totaled \$3.91 billion in the second quarter of 2011, a decrease of 0.9% from \$3.95 billion in the second quarter of 2010, and \$7.90 billion in the first six months of 2011, a decrease of 1.0% from \$7.97 billion in the first six months of 2010. Contributing to the Allstate brand standard auto premiums written decrease in the second quarter and first six months of 2011 compared to the same periods of 2010 were the following:

- decrease in PIF of 0.6% as of June 30, 2011 compared to June 30, 2010, due to fewer policies available to renew and a slight decrease in net items added to existing policies. Excluding Florida and New York, PIF increased 0.2% as of June 30, 2011 compared to June 30, 2010.
- 5.2% decrease in new issued applications on a countrywide basis to 472 thousand in the second quarter of 2011 from 498 thousand in the second quarter of 2010, and 3.0% increase to 991 thousand in the first six months of 2011 from 962 thousand in the first six months of 2010. Excluding Florida and New York (impacted by actions to improve profitability), new issued applications on a countrywide basis increased 2.4% to 422 thousand in the second quarter of 2011 from 412 thousand in the second quarter of 2010, and increased 9.3% to

58

871 thousand in the first six months of 2011 from 797 thousand in the first six months of 2010. New issued applications have increased in 34 states in 2011 compared to 2010.

- decreased average gross premium in the second quarter and first six months of 2011 compared to the same periods of 2010. The decrease in average gross premium is primarily due to rate decreases taken from the second half of 2010 through the first quarter of 2011 to improve competitive position, as well as customers electing lower coverage levels on their policies. State mix is also having an unfavorable impact on average gross premium, as we have been writing less business in Florida and New York, which have premiums higher than the countrywide average, and are writing more business in states with lower average premiums.
- 0.2 point and 0.1 point increase in the renewal ratio in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010.
   25 states are showing favorable comparisons to same period of prior year.

*Non-standard auto premiums written* totaled \$197 million in the second quarter of 2011, a decrease of 10.9% from \$221 million in the second quarter of 2010, and \$408 million in the first six months of 2011, a decrease of 11.5% from \$461 million in the first six months of 2010.

	Alls	tate bra	ınd		<b>Encompass brand</b>				
Non-Standard Auto	2011		2010		2011		2010		
Three months ended June 30,									
PIF (thousands)	599		706		3		10		
Average premium-gross written (6 months) \$	620	\$	619	\$	418	\$	429		
Renewal ratio (%) (6 months)	70.8		72.5		42.8		40.2		
Approved rate changes:									
# of states	3		5	(2)					
Countrywide (%)	0.4		2.7						
State specific (%) (1)	6.1		10.9						
Six months ended June 30,									
PIF (thousands)	599		706		3		10		
Average premium-gross written (6 months) \$	620	\$	619	\$	407	\$	436		
Renewal ratio (%) (6 months)	70.6		72.1		56.7		45.3		

<sup>(2)</sup> Rate changes that are indicated based on loss trend analysis to achieve a targeted return will continue to be pursued. Rate changes do not include rating plan enhancements, including the introduction of discounts and surcharges, that result in no change in the overall rate level in the state. These rate changes do not reflect initial rates filed for insurance subsidiaries initially writing business in a state.

<sup>(3)</sup> Represents the impact in the states where rate changes were approved during the three months and six months ended June 30, 2011 and 2010, respectively, as a percentage of total countrywide prior year-end premiums written.

<sup>(4)</sup> Represents the impact in the states where rate changes were approved during the three months and six months ended June 30, 2011 and 2010, respectively, as a percentage of its respective total prior year-end premiums written in those states.

<sup>(5)</sup> Based on historical premiums written in those states, rate changes approved for standard auto totaled \$298 million and \$474 million in the three months and six months ended June 30, 2011, respectively, compared to \$23 million and \$82 million in the three months and six months ended June 30, 2010, respectively.

<sup>(6)</sup> Allstate Brand includes targeted rate decreases in certain markets to improve our competitive position for target customers.

<sup>(7)</sup> Includes Washington D.C.

<sup>(6)</sup> Three months ended June 30, 2011 includes the impact of Florida rate increases of 20.0% and 6.0%, and a New York rate increase of 3.7%. Six months ended June 30, 2011 includes the impact of Florida rate increases of 20.9%, 20.0%, 6.0% and 2.3%, and New York rate increases of 12.0% and 3.7%.

Approved rate changes:				
# of states	6	6	(2)	
Countrywide (%)	4.0	3.6		
State specific (%) <sup>(1)</sup>	15.3	12.5		

<sup>(1)</sup> Based on historical premiums written in those states, rate changes approved for non-standard auto totaled \$3 million and \$33 million in the three months and six months ended June 30, 2011, respectively, compared to \$24 million and \$32 million in the three months and six months ended June 30, 2010, respectively.

Allstate brand non-standard auto premiums written totaled \$197 million in the second quarter of 2011, a decrease of 10.5% from \$220 million in the second quarter of 2010, and \$407 million in the first six months of 2011, a decrease of 10.9% from \$457 million in the first six months of 2010. Contributing to the Allstate brand non-standard auto premiums written decrease in the second quarter and first six months of 2011 compared to the same periods of 2010 were the following:

- decrease in PIF as of June 30, 2011 compared to June 30, 2010, due to a decline in the number of policies available to renew, a lower retention rate and fewer new issued applications
- 23.4% decrease in new issued applications to 59 thousand in the second quarter of 2011 from 77 thousand in the second quarter of 2010, and 22.2% decrease to 137 thousand in the first six months of 2011 from 176 thousand in the first six months of 2010
- increase in average gross premium in the second quarter and first six months of 2011 compared to the same periods of 2010
- 1.7 point and 1.5 point decrease in the renewal ratio in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010

59

Homeowners premiums written totaled \$1.70 billion in the second quarter of 2011, an increase of 2.5% from \$1.66 billion in the second quarter of 2010, and \$3.00 billion in the first six months of 2011, an increase of 2.6% from \$2.93 billion in the first six months of 2010. Excluding the cost of catastrophe reinsurance, premiums written increased 1.8% in both the second quarter and first six months of 2011 compared to the same periods of 2010.

	Allstate brand				<b>Encompass brand</b>			
<u>Homeowners</u>	 2011		2010		2011		2010	
Three months ended June 30,								
PIF (thousands)	6,555		6,821		307		336	
Average premium-gross written (12 months)	\$ 989	\$	933	\$	1,298	\$	1,301	
Renewal ratio (%) (12 months)	88.4		88.3		80.9		76.5	
Approved rate changes (1):				(2)		(2)		
# of states	18		14	(3)	11	(3)	7	
Countrywide (%)	1.5		2.0		0.3			
State specific (%) <sup>(2)</sup>	6.0		11.3		2.6		(0.3)	
Six months ended June 30,								
PIF (thousands)	6,555		6,821		307		336	
Average premium-gross written (12 months)	\$ 983	\$	927	\$	1,298	\$	1,300	
Renewal ratio (%) (12 months)	88.3		88.2		81.6		76.9	
Approved rate changes (1):		(2)		(2)		(2)		
# of states	27 (	(3)	19	(3)	16	(3)	11	
Countrywide (%)	3.4		2.9		1.6		0.6	
State specific (%) (2)	8.2		9.7		4.1		2.5	

<sup>(1)</sup> Includes rate changes approved based on our net cost of reinsurance.

Allstate brand homeowners premiums written totaled \$1.61 billion in the second quarter of 2011, an increase of 2.6% from \$1.57 billion in the second quarter of 2010, and \$2.83 billion in the first six months of 2011, an increase of 2.8% from \$2.75 billion in the first six months of 2010. Contributing to the Allstate brand homeowners premiums written increase in the second quarter and first six months of 2011 compared to the same periods of 2010 were the following:

- decrease in PIF of 3.9% as of June 30, 2011 compared to June 30, 2010, due to fewer policies available to renew and fewer new issued applications
- 18.5% decrease in new issued applications to 123 thousand in the second quarter of 2011 from 151 thousand in the second quarter of 2010, and 12.2% decrease to 237 thousand in the first six months of 2011 from 270 thousand in the first six months of 2010. During the second quarter of 2011, our Castle Key Indemnity Company subsidiary completed a 2008 regulatory consent decree to sell 50,000 new homeowners policies in Florida by November 2011.
- increase in average gross premium in the second quarter and first six months of 2011 compared to the same periods of 2010, primarily due to rate changes
- 0.1 point increase in the renewal ratio in both the second quarter and first six months of 2011 compared to the same periods of 2010
- decrease in the cost of our catastrophe reinsurance program in the second quarter and first six months of 2011 compared to the same periods of 2010

<sup>(2)</sup> Includes Washington D.C.

<sup>(2)</sup> Based on historical premiums written in those states, rate changes approved for homeowners totaled \$94 million and \$210 million in the three months and six months ended June 30, 2011, respectively, compared to \$120 million and \$174 million in the three months and six months ended June 30, 2010, respectively.

<sup>(3)</sup> Includes Washington D.C.

**Underwriting results** are shown in the following table.

(\$ in millions)	Three months ended June 30,				Six months ended June 30,			
	_	2011		2010	2011		2010	
Premiums written	\$	6,611	\$	6,640	\$ 12,827	\$	12,898	
Premiums earned	\$	6,457	\$	6,513	\$ 12,906	\$	13,016	
Claims and claims expense		(6,352)		(4,713)	(10,824)		(9,503)	
Amortization of DAC		(908)		(914)	(1,812)		(1,839)	
Other costs and expenses		(684)		(663)	(1,413)		(1,365)	
Restructuring and related charges		(11)		(14)	(22)		(25)	
Underwriting (loss) income	\$	(1,498)	\$	209	\$ (1,165)	\$	284	
Catastrophe losses	\$	2,339	\$	636	\$ 2,672	\$	1,284	
Underwriting (loss) income by line of business								
Standard auto	\$	59	\$	217	\$ 247	\$	430	
Non-standard auto		15		10	40		25	
Homeowners		(1,397)		(57)	(1,270)		(249)	
Other personal lines		(175)		39	(182)		78	
Underwriting (loss) income	\$	(1,498)	\$	209	\$ (1,165)	\$	284	
Underwriting (loss) income by brand								
Allstate brand	\$	(1,444)	\$	201	\$ (1,110)	\$	319	
Encompass brand		(54)		8	(55)		(35)	
Underwriting (loss) income	\$	(1,498)	\$	209	\$ (1,165)	\$	284	

Allstate Protection experienced an underwriting loss of \$1.50 billion in the second quarter of 2011 compared to underwriting income of \$209 million in the same period of 2010, and an underwriting loss of \$1.17 billion in the first six months of 2011 compared to underwriting income of \$284 million in the same period of 2010. The decrease in both periods was primarily due to decreases in homeowners, other personal lines and standard auto underwriting income. Homeowners underwriting income decreased \$1.34 billion to an underwriting loss of \$1.40 billion in the second quarter of 2011 from an underwriting loss of \$57 million in the second quarter of 2010, and decreased \$1.02 billion to an underwriting loss of \$1.27 billion in the first six months of 2011 from an underwriting loss of \$249 million in the first six months of 2010. The decrease in both periods was primarily due to increases in catastrophe losses, including prior year reestimates for catastrophes. Other personal lines underwriting income decreased \$214 million to an underwriting loss of \$175 million in the second quarter of 2011 from an underwriting income of \$39 million in the second quarter of 2010, and decreased \$260 million to an underwriting loss of \$182 million in the first six months of 2011 from an underwriting income of \$78 million in the first six months of 2010. The decrease in both periods was primarily due to increases in catastrophe losses and unfavorable reserve reestimates. Standard auto underwriting income decreased \$158 million to an underwriting income of \$59 million in the second quarter of 2011 from an underwriting income of \$430 million in the first six months of 2010. The decrease in both periods was primarily due to increases in catastrophe losses, partially offset by improved claim frequency.

**Catastrophe losses** in the second quarter and first six months of 2011 were \$2.34 billion and \$2.67 billion, respectively, as detailed in the table below. This compares to catastrophe losses in the second quarter and first six months of 2010 of \$636 million and \$1.28 billion, respectively.

We define a "catastrophe" as an event that produces pre-tax losses before reinsurance in excess of \$1 million and involves multiple first party policyholders, or an event that produces a number of claims in excess of a preset, per-event threshold of average claims in a specific area, occurring within a certain amount of time following the event. Catastrophes are caused by various natural events including high winds, winter storms, tornadoes, hailstorms, wildfires, tropical storms, hurricanes, earthquakes and volcanoes. We are also exposed to man-made catastrophic events, such as

61

certain acts of terrorism or industrial accidents. The nature and level of catastrophes in any future period cannot be reliably predicted.

Catastrophe losses related to events that occurred by the size of the event are shown in the following table.

(\$ in millions) Three months ended
June 30, 2011

		Combined	Average			
	Number of events		and claims expense		ratio impact	catastrophe loss per event
Size of catastrophe	or evenes		сарсизс		Impuct	 1033 per event
Greater than \$250 million	3	9.1%	1,172	50.1%	18.1	\$ 391
\$101 million to \$250 million	3	9.1	470	20.1	7.3	157
\$50 million to \$100 million	6	18.2	443	18.9	6.9	74
Less than \$50 million	21	63.6	250	10.7	3.9	12
Total	33	100.0%	2,335	99.8	36.2	71
Prior year reserve						
reestimates			(17)	(0.7)	(0.3)	
Prior quarter reserve						
reestimates			21	0.9	0.3	
Total catastrophe losses		9	2,339	100.0%	36.2	

#### Six months ended June 30, 2011

				Claims					
				and		Combined		Average	
	Number	claims				ratio		catastrophe	
	of events	expense				impact		loss per event	
Size of catastrophe			_	<u> </u>			_		
Greater than \$250 million	3	6.1%	\$	1,172	43.8%	9.1	\$	391	
\$101 million to \$250 million	3	6.1		470	17.6	3.6		157	
\$50 million to \$100 million	8	16.3		609	22.8	4.7		76	
Less than \$50 million	35	71.5		472	17.7	3.7		13	
Total	49	100.0%		2,723	101.9	21.1		56	
Prior year reserve									
reestimates				(51)	(1.9)	(0.4)			
Total catastrophe losses			\$	2,672	100.0%	20.7			

Catastrophe losses incurred by the type of event are shown in the following table.

(\$ in millions)			Three m Ju	onth ine 3			Six months ended June 30,							
	_	2011	Number of events		2010	Number of events	_	2011	Number of events	_	2010	Number of events		
Tornadoes	\$	1,326	5	\$	141	5	\$	1,340	6	\$	141	5		
Wind/Hail		995	25		616	24		1,192	35		979	30		
Wildfires Other events Prior year reserve		14 	3	ı	1	1		19 172	5 3		 262	6		
reestimates Prior quarter reserve		(17)			(83)			(51)			(98)			
reestimates		21			(39)		_			_				
Total catastrophe losses	\$	2,339	33	\$	636	30	\$	2,672	49	\$_	1,284	41		

62

**Combined ratio** Loss ratios are a measure of profitability. Loss ratios by product, and expense and combined ratios by brand, are shown in the following table. These ratios are defined in the Property-Liability Operations section of the MD&A.

			Three mont						Six month June			
	Loss 1	catio (1)	Effect of catastrophe losses on the loss ratio		Effect of pre-tax reserve reestimates on the combined ratio		Loss ratio (1)		Effect of catastrophe losses on the loss ratio		Effec pre-tax r reestin on t combine	reserve nates he
	2011	2010	2011	2010	2011	2010	2011	2010	2011	2010	2011	2010
Allstate brand loss ratio:												
Standard auto	73.2	70.1	6.7	2.0	(2.2)	(1.9)	71.7	69.8	3.6	1.3	(1.3)	(1.0)
Non-standard auto	69.3	68.9	3.9	0.4	(1.0)	(4.8)	67.0	68.8	1.9	0.4	(2.2)	(3.1)
Homeowners	171.1	82.6	123.2	34.7	0.3	(4.2)	119.7	85.0	70.6	35.9	(1.2)	(2.3)
Other personal lines	100.5	65.7	35.3	8.3	6.1	(0.7)	83.9	64.6	21.1	7.8	4.3	(2.3)
Total Allstate brand loss ratio	98.7	72.5	36.8	10.0	(0.8)	(2.4)	84.0	72.7	21.0	9.8	(0.8)	(1.5)
Allstate brand expense ratio	24.6	24.3	50.0	10.0	(0.0)	(2.4)	25.0	24.7	21.0	5.0	(0.0)	(1.5)
•												
Allstate brand combined ratio	123.3	96.8					109.0	97.4				
Encompass brand loss ratio:												
Standard auto	78.7	73.0	3.2	0.5		1.6	77.1	74.9	1.6	0.8	1.6	3.4
Non-standard auto	100.0	100.0			(100.0)		100.0	100.0			(50.0)	
Homeowners	107.7	64.6	61.5	15.6	(1.1)	(1.0)	86.8	84.2	39.0	31.1		(1.5)
Other personal lines	104.3	64.0	17.4			(4.0)	84.8	77.6	13.0	6.1	(4.3)	
Total Encompass brand loss ratio	90.7	69.8	24.1	5.2	(0.7)	0.3	81.1	78.3	15.0	10.6	0.4	1.6
Encompass brand expense ratio	29.3	27.6			(*)		29.0	27.3			***	
Encompass brand combined ratio	120.0	97.4					110.1	105.6				
Encompass Diana combinea rado	120.0	57.7					110.1	103.0				
Allstate Protection loss ratio	98.4	72.4	36.2	9.8	(0.8)	(2.3)	83.9	73.0	20.7	9.9	(0.7)	(1.4)
Allstate Protection expense ratio	24.8	24.4					25.1	24.8				
Allstate Protection combined ratio	123.2	96.8					109.0	97.8				

 $<sup>^{\</sup>left(1\right)}$  Ratios are calculated using the premiums earned for the respective line of business.

Standard auto loss ratio for the Allstate brand increased 3.1 points and 1.9 points in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010 due to higher catastrophe losses, partially offset by improved claim frequency. Excluding the impact of catastrophe losses, the Allstate brand standard auto loss ratio improved 1.6 points and 0.4 points in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010. Florida and New York continue to have loss ratios higher than the countrywide average in the first six months of 2011

and results in these two key states have improved for three straight quarters, reducing the pressure on countrywide results. We continue to pursue profitability management actions in Florida and New York, including rate increases, underwriting restrictions, increased claims staffing and review, and continued advocacy for legislative reform. In the second quarter of 2011, claim frequencies in the bodily injury and physical damage coverages have decreased compared to the same period of 2010. Through the first six months of 2011, claim frequencies in the physical damage coverages are lower than the same period of 2010. In the first six months of 2011, bodily injury claim frequencies remain slightly above 2010 levels but remain within historical norms. Bodily injury and physical damage coverages severity results increased in line with historical Consumer Price Index ("CPI") trends.

*Homeowners loss ratio* for the Allstate brand increased 88.5 points to 171.1 in the second quarter of 2011 from 82.6 in the second quarter of 2010, and 34.7 points to 119.7 in the first six months of 2011 from 85.0 in the first six months of 2010 due to higher catastrophe losses including prior year reserve reestimates for catastrophes. Excluding

63

the impact of catastrophe losses, homeowners loss ratio for the Allstate brand in the second quarter and first six months of 2011 were comparable to the same periods of 2010.

**Expense ratio** for Allstate Protection increased 0.4 points and 0.3 points in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010. The increase in both periods was driven by marketing costs and lower premiums earned, partially offset by improved operational efficiencies. Restructuring costs in the second quarter and first six months of 2011 were comparable to the same periods of 2010.

The impact of specific costs and expenses on the expense ratio are included in the following table.

			i nree months e	naea June 30,			
	Allstate	brand	Encompas	s brand	Allstate Protection		
	2011	2010	2011	2010	2011	2010	
Amortization of DAC	13.9	13.9	18.1	18.2	14.0	14.0	
Other costs and expenses	10.6	10.2	11.2	9.1	10.6	10.2	
Restructuring and related charges	0.1	0.2		0.3	0.2	0.2	
Total expense ratio	24.6	24.3	29.3	27.6	24.8	24.4	

		Six months ended June 30,										
	Allstate l	brand	Encompas	s brand	Allstate Protection							
	2011	2010	2011	2010	2011	2010						
Amortization of DAC	13.9	13.9	18.0	18.3	14.0	14.1						
Other costs and expenses	10.9	10.6	11.0	8.4	10.9	10.5						
Restructuring and related charges	0.2	0.2		0.6	0.2	0.2						
Total expense ratio	25.0	24.7	29.0	27.3	25.1	24.8						

#### **Allstate Protection Reinsurance**

Our catastrophe reinsurance program was designed, utilizing our risk management methodology, to address our exposure to catastrophes nationwide. Our program provides reinsurance protection for catastrophes including storms named or numbered by the National Weather Service, fires following earthquakes, earthquakes and wildfires including California wildfires. These reinsurance agreements are part of our catastrophe management strategy, which is intended to provide our shareholders an acceptable return on the risks assumed in our property business, and to reduce variability of earnings, while providing protection to our customers.

During the second quarter of 2011, we placed the Florida component of our reinsurance program. The Florida component of our reinsurance program is designed separately from the other components of the program to address the distinct needs of our separately capitalized legal entities in that state. It comprises multiple contracts reinsuring Castle Key Insurance Company and its subsidiaries ("Castle Key Group") for personal property excess catastrophe losses in Florida for multiple perils including hurricanes, windstorms, hail, tornados, earthquakes, fire following earthquakes, riots, freeze and wildfires. The agreement, effective June 1, 2011 for a one year term, incorporates coverage placed with the Florida Hurricane Catastrophe Fund ("FHCF") for hurricane losses including both the mandatory FHCF coverage and the Castle Key Group's elected participation in the optional temporary increase in coverage limit ("TICL"). The FHCF coverage includes an estimated maximum provisional limit of 90% of \$422.4 million or \$380.2 million (comprising 90% of the mandatory FHCF coverage layer of \$312.2 million plus 90% of the TICL layer of \$110.2 million), in excess of a provisional retention of \$121.7 million, and also includes reimbursement of eligible loss adjustment expenses at 5%. The limits and retentions for the FHCF and TICL coverage are subject to remeasurement based on June 30<sup>th</sup> exposure data. The FHCF's retention is subject to adjustment upward or downward to an actual retention based on submitted exposures to the FHCF by all participants. For each of the two largest hurricanes for the season beginning June 1, 2011. The agreements are listed and described below.

• Below FHCF – provides coverage on \$91.7 million of losses in excess of \$30 million and after \$10 million in losses otherwise recoverable and is 100% placed, with one prepaid reinstatement of limit. Losses from multiple qualifying occurrences can apply to the \$10 million threshold.

64

- Third Limit Below FHCF provides coverage on \$91.7 million of losses in excess of \$30 million after the exhaustion of the two limits (\$183.4 million) provided by the Below FHCF contract with no reinstatement of limit.
- · Mandatory FHCF provides 90% of \$312.2 million excess of \$121.7 million with no reinstatement of limit.
- · FHCF Sliver provides coverage on 10% co-participation of the mandatory FHCF coverage payout up to \$31.2 million, and is 100% placed with

- no reinstatement of limit.
- · Optional TICL provides 90% of \$110.2 million excess of \$433.9 million with no reinstatement of limit.
- · TICL Sliver provides coverage on 10% co-participation of the TICL coverage payout up to \$11 million, and is 100% placed with no reinstatement of limit.
- Excess provides coverage of \$372.6 million of losses in excess of \$121.7 million (the FHCF Retention), and in excess of an estimated \$422.4 million equivalent to \$312.2 million (the mandatory FHCF coverage and FHCF Sliver payouts) and \$110.2 million (the TICL and TICL Sliver payouts). This contract is 100% placed with one prepaid reinstatement of limit.

We estimate that the total annualized cost of all catastrophe reinsurance programs for the year beginning June 1, 2011 will be approximately \$564 million or \$141 million per quarter compared to \$560 million annualized cost for the year beginning June 1, 2010. The total cost of our catastrophe reinsurance programs during 2010 was \$150 million in the first quarter, \$151 million in the second quarter, \$141 million in the third quarter and \$151 million in the fourth quarter. The total cost of our property catastrophe reinsurance programs during the first and second quarter of 2011 was \$138 million and \$142 million, respectively. These quarterly costs reflect premium re-measurements recognized in the quarter. We continue to attempt to capture our reinsurance cost in premium rates as allowed by state regulatory authorities.

**Reserve reestimates** The tables below show Allstate Protection net reserves representing the estimated cost of outstanding claims as they were recorded at the beginning of years 2011 and 2010, and the effect of reestimates in each year.

(\$ in millions)		January 1 reserves						
	2011 2010							
Auto	\$	11,034	\$	10,606				
Homeowners		2,442		2,399				
Other personal lines		2,141		2,145				
Total Allstate Protection	\$	15,617	\$	15,150				

(\$ in millions, except ratios)				Three montl June 3					Six months ended June 30,				
	_	Reserve reestimates (1)(2)			Effect combined		_	Reserve reestimates (1)(2)			Effect on combined ratio (2)		
	_	2011		2010	2011	2010		2011		2010	2011	2010	
Auto	\$	(90)	\$	(85)	(1.4)	(1.3)	\$	(109)	\$	(80)	(0.8)	(0.6)	
Homeowners		3		(61)		(0.9)		(35)		(69)	(0.3)	(0.6)	
Other personal lines		36		(5)	0.6	(0.1)		49		(27)	0.4	(0.2)	
Total Allstate Protection (3)	\$	(51)	\$	(151)	(0.8)	(2.3)	\$	(95)	\$	(176)	(0.7)	(1.4)	
Allstate brand	\$	(49)	\$	(152)	(0.8)	(2.3)	\$	(97)	\$	(186)	(0.7)	(1.5)	
Encompass brand		(2)		1				2		10		0.1	
Total Allstate Protection (3)	\$	(51)	\$	(151)	(0.8)	(2.3)	\$	(95)	\$	(176)	(0.7)	(1.4)	

Favorable reserve reestimates are shown in parentheses.

65

#### DISCONTINUED LINES AND COVERAGES SEGMENT

**Overview** The Discontinued Lines and Coverages segment includes results from insurance coverage that we no longer write and results for certain commercial and other businesses in run-off. Our exposure to asbestos, environmental and other discontinued lines claims is reported in this segment. We have assigned management of this segment to a designated group of professionals with expertise in claims handling, policy coverage interpretation, exposure identification and reinsurance collection. As part of its responsibilities, this group is also regularly engaged in policy buybacks, settlements and reinsurance assumed and ceded commutations.

Summarized underwriting results are presented in the following table.

(\$ in millions)			nded				
		2011	2010		2011		2010
Premiums written	\$		\$ 	\$	(1)	\$	
Premiums earned	\$		\$ 	\$	(1)	\$	
Claims and claims expense		(3)	(1)		(7)		(3)
Operating costs and expenses		(1)	(1)		(2)		(3)
Underwriting loss	\$	(4)	\$ (2)	\$	(10)	\$	(6)

#### PROPERTY-LIABILITY INVESTMENT RESULTS

**Net investment income** of \$310 million in the second quarter of 2011 was comparable to the second quarter of 2010 as higher yields were offset by lower average investment balances. Net investment income decreased 3.3% or \$20 million to \$594 million in the first six months of 2011 from \$614 million

Discontinued Lines and Coverages segment reserve reestimates in the three months and six months ended June 30, 2011 totaled \$4 million and \$7 million unfavorable compared to \$1 million and \$3 million unfavorable in the three months and six months ended June 30, 2010. The effect on the combined ratio totaled 0.1 in the three months ended June 30, 2010. There was no effect on the combined ratio in the three months ended June 30, 2010 and the six months ended June 30, 2011.

<sup>(3)</sup> Reserve reestimates included in catastrophe losses totaled \$17 million and \$51 million favorable in the three months and six months ended June 30, 2011, respectively, compared to \$83 million and \$98 million favorable in the three months and six months ended June 30, 2010, respectively.

in the first six months of 2010 primarily due to lower average investment balances. Net investment income was \$284 million, \$291 million and \$284 million in the third quarter of 2010, fourth quarter of 2010 and first quarter of 2011, respectively.

Net realized capital gains and losses are presented in the following table.

(\$ in millions)	_	Three mo	onths ne 30		Six months ended June 30,			
		2011		2010	2011	2010		
Impairment write-downs	\$	(27)	\$	(96) \$	(91) \$	(175)		
Change in intent write-downs		(11)		(10)	(38)	(19)		
Net other-than-temporary impairment	·							
losses recognized in earnings		(38)		(106)	(129)	(194)		
Sales		29		121	201	162		
Valuation of derivative instruments		(12)		(134)	14	(235)		
Settlements of derivative instruments		(7)		3	(102)	(46)		
EMA limited partnership income		20		10	65	17		
Realized capital gains and losses, pre-tax	· <u></u>	(8)		(106)	49	(296)		
Income tax benefit (expense)		2		37	(17)	104		
Realized capital gains and losses, after-tax	\$	(6)	\$	(69) \$	32 \$	(192)		

For a further discussion of net realized capital gains and losses, see the Investments section of the MD&A.

66

#### ALLSTATE FINANCIAL HIGHLIGHTS

- Net income was \$166 million and \$263 million in the second quarter and first six months of 2011, respectively, compared to net losses of \$107 million and \$103 million in the second quarter and first six months of 2010, respectively.
- Premiums and contract charges on underwritten products, including traditional life, interest-sensitive life and accident and health insurance, totaled \$524 million in second quarter 2011, an increase of 4.0% from the prior year period, and \$1.04 billion in the first six months of 2011, an increase of 3.3% from the prior year period.
- Net realized capital gains totaled \$62 million and \$101 million in the second quarter and first six months of 2011, respectively, compared to net realized capital losses of \$353 million and \$515 million in the second quarter and first six months of 2010, respectively.
- Investments as of June 30, 2011 totaled \$59.66 billion, reflecting a decrease in carrying value of \$1.92 billion from \$61.58 billion as of December 31, 2010. Net investment income decreased 4.0% to \$694 million in the second quarter and 5.2% to \$1.38 billion in the first six months of 2011 from \$723 million and \$1.45 billion in the second quarter and first six months of 2010, respectively.
- Contractholder funds as of June 30, 2011 totaled \$45.08 billion, reflecting decreases of \$3.11 billion from \$48.19 billion as of December 31, 2010 and \$4.36 billion from \$49.44 billion as of June 30, 2010.

## ALLSTATE FINANCIAL SEGMENT

**Summary analysis** Summarized financial data is presented in the following table.

(\$ in millions)	Three mo	onths ene 30,	Six mor Jui	nths en ne 30,	ıded	
	 2011		2010	 2011		2010
Revenues						
Life and annuity premiums and contract charges	\$ 547	\$	545	\$ 1,116	\$	1,089
Net investment income	694		723	1,378		1,454
Realized capital gains and losses	62		(353)	101		(515)
Total revenues	 1,303		915	 2,595		2,028
Costs and expenses						
Life and annuity contract benefits	(422)		(485)	(876)		(927)
Interest credited to contractholder funds	(417)		(450)	(835)		(913)
Amortization of DAC	(110)		(35)	(257)		(124)
Operating costs and expenses	(110)		(116)	(219)		(236)
Restructuring and related charges	` <u></u>		1	2		1
Total costs and expenses	 (1,059)		(1,085)	 (2,185)		(2,199)
Gain (loss) on disposition of operations	6		2	(17)		3
Income tax (expense) benefit	(84)		61	(130)		65
Net income (loss)	\$ 166	\$	(107)	\$ 263	\$	(103)
Investments as of June 30				\$ 59,659	\$	61,804

*Net income* in the second quarter of 2011 was \$166 million compared to a net loss of \$107 million in the same period of 2010. The \$273 million improvement was primarily due to net realized capital gains in the current year compared to net realized capital losses in the prior year, lower life and annuity contract benefits and decreased interest credited to contractholder funds, partially offset by higher amortization of DAC and lower net investment income.

Net income in the first six months of 2011 was \$263 million compared to a net loss of \$103 million in the first six months of 2010. The \$366 million improvement was primarily due to net realized capital gains in the current year compared to net realized capital losses in the prior year, decreased interest credited to contractholder funds and lower life and annuity contract benefits, partially offset by higher amortization of DAC.

67

**Analysis of revenues** Total revenues increased 42.4% or \$388 million in the second quarter of 2011 and 28.0% or \$567 million in the first six months of 2011 compared to the same periods of 2010 due to net realized capital gains in the current year compared to net realized capital losses in the prior year and higher premiums and contract charges, partially offset by lower net investment income.

Life and annuity premiums and contract charges Premiums represent revenues generated from traditional life insurance, immediate annuities with life contingencies, and accident and health insurance products that have significant mortality or morbidity risk. Contract charges are revenues generated from interest-sensitive and variable life insurance and fixed annuities for which deposits are classified as contractholder funds or separate account liabilities. Contract charges are assessed against the contractholder account values for maintenance, administration, cost of insurance and surrender prior to contractually specified dates.

The following table summarizes life and annuity premiums and contract charges by product.

(\$ in millions)  Three months ended  June 30,						Six mo Ju	nths er ne 30,	nded
		2011		2010		2011		2010
Underwritten products	-							
Traditional life insurance premiums	\$	109	\$	104	\$	217	\$	210
Accident and health insurance premiums		162		151		323		307
Interest-sensitive life insurance contract charges		253		249		501		491
Subtotal		524		504		1,041		1,008
Annuities								
Immediate annuities with life contingencies premiums		15		31		58		58
Other fixed annuity contract charges		8		10		17		23
Subtotal	_	23	_	41	_	75	_	81
Life and annuity premiums and contract charges (1)	\$	547	\$	545	\$	1,116	\$	1,089

<sup>(1)</sup> Contract charges related to the cost of insurance totaled \$162 million and \$159 million in the second quarter of 2011 and 2010, respectively, and \$324 million and \$315 million in the first six months of 2011 and 2010, respectively.

Total premiums and contract charges increased 0.4% and 2.5% in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010 primarily due to growth in Allstate Benefits's accident and health insurance business in force and higher contract charges on interest-sensitive life insurance products resulting from the aging of our policyholders and a shift in the mix of policies in force to contracts with higher cost of insurance rates, partially offset by lower sales of immediate annuities with life contingencies. Sales of immediate annuities with life contingencies fluctuate with changes in our pricing competitiveness relative to other insurers.

68

Contractholder funds represent interest-bearing liabilities arising from the sale of individual and institutional products, such as interest-sensitive life insurance, fixed annuities, funding agreements and bank deposits. The balance of contractholder funds is equal to the cumulative deposits received and interest credited to the contractholder less cumulative contract maturities, benefits, surrenders, withdrawals and contract charges for mortality or administrative expenses. The following table shows the changes in contractholder funds.

(\$ in millions)		Three m Ju	onths e ne 30,	ended		Six months ended June 30,			
	_	2011		2010		2011		2010	
Contractholder funds, beginning balance	\$	46,834	\$	51,027	\$	48,195	\$	52,582	
Deposits									
Fixed annuities		142		237		306		528	
Interest-sensitive life insurance		316		391		645		786	
Bank and other deposits		97		234		310		486	
Total deposits		555	_	862	_	1,261	_	1,800	
Interest credited		413		448		823		910	
Maturities, benefits, withdrawals and other adjustments									
Maturities and retirements of institutional products		(306)		(827)		(793)		(1,781)	

Benefits		(367)	(395)	(739)		(790)
Surrenders and partial withdrawals		(1,723)	(1,355)	(3,016)		(2,603)
Contract charges		(255)	(243)	(506)		(484)
Net transfers from separate accounts		3	3	6		5
Fair value hedge adjustments for institutional products			(74)	(34)		(197)
Other adjustments (1)		(76)	(3)	(119)		1
Total maturities, benefits, withdrawals and other adjustments	_	(2,724)	(2,894)	(5,201)	_	(5,849)
Contractholder funds, ending balance	\$	45,078	\$ 49,443	\$ 45,078	\$	49,443

<sup>(1)</sup> The table above illustrates the changes in contractholder funds, which are presented gross of reinsurance recoverables on the Condensed Consolidated Statements of Financial Position. The table above is intended to supplement our discussion and analysis of revenues, which are presented net of reinsurance on the Condensed Consolidated Statements of Operations. As a result, the net change in contractholder funds associated with products reinsured to third parties is reflected as a component of the other adjustments line.

Contractholder funds decreased 3.7% and 6.5% in the second quarter and first six months of 2011, respectively, compared to decreases of 3.1% and 6.0% in the second quarter and first six months of 2010, respectively, reflecting our continuing strategy to reduce our concentration in spread-based products. Average contractholder funds decreased 8.5% and 8.6% in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010.

Contractholder deposits decreased 35.6% and 29.9% in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010 primarily due to lower deposits on fixed annuities and Allstate Bank products.

Maturities and retirements of institutional products decreased 63.0% to \$306 million in the second quarter of 2011 and 55.5% to \$793 million in the first six months of 2011 from \$827 million and \$1.78 billion in the same periods of 2010, reflecting the continuing decline in these obligations over the past three years.

Surrenders and partial withdrawals on deferred fixed annuities, interest-sensitive life insurance products and Allstate Bank products (including maturities of certificates of deposit) increased 27.2% to \$1.72 billion in the second quarter of 2011 and 15.9% to \$3.02 billion in the first six months of 2011 from \$1.36 billion and \$2.60 billion in the second quarter and first six months of 2010, respectively, primarily due to higher surrenders and partial withdrawals on fixed annuities, partially offset by lower surrenders and partial withdrawals on interest-sensitive life insurance products and Allstate Bank products. The increase for fixed annuities resulted from an increased number of contracts reaching the 30-45 day period (typically at their 5 or 6 year anniversary) during which there is no surrender charge as well as crediting rate actions taken by management. The annualized surrender and partial withdrawal rate on deferred fixed annuities, interest-sensitive life insurance products and Allstate Bank products,

69

based on the beginning of year contractholder funds, was 15.0% in the first six months of 2011 compared to 12.2% in the first six months of 2010.

*Net investment income* decreased 4.0% or \$29 million to \$694 million in the second quarter of 2011 and 5.2% or \$76 million to \$1.38 billion in the first six months of 2011 from \$723 million and \$1.45 billion in the second quarter and first six months of 2010, respectively, primarily due to reduced average investment balances which were partially offset by higher yields. Net investment income was \$707 million, \$692 million and \$684 million in the third quarter of 2010, fourth quarter of 2010 and first quarter of 2011, respectively.

Net realized capital gains and losses are presented in the following table.

(\$ in millions)		Three m Ju	onths e ne 30,	ended	Six months ended June 30,						
	_	2011		2010	 2011	_	2010				
Impairment write-downs	\$	(43)	\$	(143)	\$ (93)	\$	(287)				
Change in intent write-downs		(5)		(57)	(47)		(80)				
Net other-than-temporary impairment losses											
recognized in earnings		(48)		(200)	(140)		(367)				
Sales		112		18	223		62				
Valuation of derivative instruments		(38)		(149)	(42)		(203)				
Settlements of derivative instruments		4		(30)	10		(11)				
EMA limited partnership income		32		8	50		4				
Realized capital gains and losses, pre-tax		62	-	(353)	101		(515)				
Income tax (expense) benefit		(22)		123	(36)		180				
Realized capital gains and losses, after-tax	\$	40	\$	(230)	\$ 65	\$	(335)				

For further discussion of realized capital gains and losses, see the Investments section of the MD&A.

**Analysis of costs and expenses** Total costs and expenses decreased 2.4% or \$26 million in the second quarter of 2011 and 0.6% or \$14 million in the first six months of 2011 compared to the same periods of 2010 primarily due to lower contract benefits and interest credited to contractholder funds, partially offset by higher amortization of DAC.

*Life and annuity contract benefits* decreased 13.0% or \$63 million in the second quarter of 2011 and 5.5% or \$51 million in the first six months of 2011 compared to the same periods of 2010 primarily due to reserve reestimations recorded in second quarter 2010 that did not recur in 2011.

The reserve reestimations in the second quarter of 2010 utilized more refined policy level information and assumptions. The increase in reserves for certain secondary guarantees on universal life insurance policies resulted in a charge to contract benefits of \$68 million and a related reduction in amortization of DAC of \$50 million. The decrease in reserves for immediate annuities resulted in a credit to contract benefits of \$26 million. The net impact was an increase to income of \$8 million, pre-tax.

We analyze our mortality and morbidity results using the difference between premiums and contract charges earned for the cost of insurance and life and annuity contract benefits excluding the portion related to the implied interest on immediate annuities with life contingencies ("benefit spread"). This implied interest totaled \$135 million and \$270 million in the second quarter and first six months of 2011, respectively, compared to \$139 million and \$278 million in the second quarter and first six months of 2010, respectively.

The benefit spread by product group is disclosed in the following table.

(\$ in millions)	Three m Ju	onths ne 30,	ended	Six mor Ju	nths ei ne 30,	nded
	 2011		2010	2011		2010
Life insurance	\$ 98	\$	23	\$ 191	\$	111
Accident and health insurance	71		60	145		124
Annuities	(8)		16	(20)		6
Total benefit spread	\$ 161	\$	99	\$ 316	\$	241

Benefit spread increased 62.6% or \$62 million in the second quarter of 2011 and 31.1% or \$75 million in the first six months of 2011 compared to the same periods of 2010. The increase in both periods was primarily due to reestimations of reserves that increased contract benefits for interest-sensitive life insurance and decreased contract benefits for immediate annuities with life contingencies in 2010, and lower morbidity experience on certain accident and health products and growth at Allstate Benefits.

Interest credited to contractholder funds decreased 7.3% or \$33 million in the second quarter of 2011 and 8.5% or \$78 million in the first six months of 2011 compared to the same periods of 2010 primarily due to lower average contractholder funds and lower interest crediting rates on deferred fixed annuities, interest-sensitive life insurance and immediate fixed annuities. Additionally, valuation changes on derivatives embedded in equity-indexed annuity contracts that are not hedged increased interest credited to contractholder funds by \$4 million in the second quarter of 2011 and decreased interest credited to contractholder funds by \$8 million in first six months of 2011.

Amortization of deferred sales inducement costs in the second quarter and first six months of 2011 was \$5 million and \$15 million, respectively, compared to \$6 million and \$11 million in the second quarter and first six months of 2010, respectively.

In order to analyze the impact of net investment income and interest credited to contractholders on net income, we monitor the difference between net investment income and the sum of interest credited to contractholder funds and the implied interest on immediate annuities with life contingencies, which is included as a component of life and annuity contract benefits on the Condensed Consolidated Statements of Operations ("investment spread").

The investment spread by product group is shown in the following table.

(\$ in millions)	June 30,						Six months ended June 30,				
		2011		2010		2011		2010			
Annuities and institutional products	\$	51	\$	54	\$	99	\$	104			
Life insurance		14		6		25		13			
Allstate Bank products		6		8		14		16			
Accident and health insurance		5		4		10		8			
Net investment income on investments supporting capital		66		62		125		122			
Total investment spread	\$	142	\$	134	\$	273	\$	263			

Investment spread increased 6.0% or \$8 million in the second quarter of 2011 and 3.8% or \$10 million in the first six months of 2011 compared to the same periods of 2010 as actions to improve investment portfolio yields and lower crediting rates more than offset the effect of the continuing decline in our spread-based business in force.

71

To further analyze investment spreads, the following table summarizes the weighted average investment yield on assets supporting product liabilities and capital, interest crediting rates and investment spreads.

			11	iree month	s ena	ea June 3	50,			
Weighted average investment yield			U		U		Weighted average investment spreads			
2011		2010		2011		2010		2011	2010	
5.5	%	5.5	%	4.2	%	4.4	%	1.3 %	1.1	%
4.6		4.5		3.3		3.2		1.3	1.3	
6.4		6.5		6.3		6.4		0.1	0.1	
3.8		3.7		n/a		n/a		n/a	n/a	
			5	Six months	ende	d June 30	),			
	2011 5.5 4.6 6.4	investment 2011 5.5 % 4.6 6.4	investment yield           2011         2010           5.5         %         5.5           4.6         4.5           6.4         6.5	Weighted average investment yield       2011     2010       5.5     %     5.5     %       4.6     4.5       6.4     6.5       3.8     3.7	Weighted average investment yield         Weight interest or interest	Weighted average investment yield         Weighted average interest credit           2011         2010           5.5         %           4.6         4.5           6.4         6.5           3.8         3.7           n/a	Weighted average investment yield         Weighted average interest crediting rate           2011         2010         2011         2010           5.5         %         5.5         %         4.2         %         4.4           4.6         4.5         3.3         3.2           6.4         6.5         6.3         6.4           3.8         3.7         n/a         n/a	investment yield         interest crediting rate           2011         2010         2011         2010           5.5         %         5.5         %         4.2         %         4.4         %           4.6         4.5         3.3         3.2         6.4         6.4         6.5         6.3         6.4         6.4	Weighted average investment yield         Weighted average interest crediting rate         Weighted average investment         Average investm	Weighted average investment yield         Weighted average interest crediting rate         Weighted average investment spreads           2011         2010         2011         2010         2011         2010           5.5         %         5.5         %         4.2         %         4.4         %         1.3         %         1.1           4.6         4.5         3.3         3.2         1.3         1.3           6.4         6.5         6.3         6.4         0.1         0.1           3.8         3.7         n/a         n/a         n/a         n/a

	Weighted average investment yield			Weighted average interest crediting rate				Weighted average investment spreads				
	2011		2010		2011		2010		2011		2010	-
Interest-sensitive life insurance	5.5	%	5.5	%	4.2	%	4.4	%	1.3	%	1.1	%
Deferred fixed annuities and institutional products Immediate fixed annuities with and without life	4.6		4.5		3.3		3.2		1.3		1.3	
contingencies Investments supporting capital, traditional life and	6.3		6.5		6.3		6.4				0.1	
other products	3.7		3.7		n/a		n/a		n/a		n/a	

The following table summarizes our product liabilities and indicates the account value of those contracts and policies in which an investment spread is generated.

(\$ in millions)	<u></u>	Ju	ıne 30,	
		2011		2010
Immediate fixed annuities with life contingencies	\$	8,768	\$	8,572
Other life contingent contracts and other		5,019		4,911
Reserve for life-contingent contract benefits	\$	13,787	\$	13,483
			_	
Interest-sensitive life insurance	\$	10,728	\$	10,525
Deferred fixed annuities		27,263		30,709
Immediate fixed annuities without life contingencies		3,782		3,840
Institutional products		1,915		2,650
Allstate Bank products		923		1,092
Market value adjustments related to fair value hedges and other	<u></u>	467		627
Contractholder funds	\$	45,078	\$	49,443

72

*Amortization of DAC* increased \$75 million in the second quarter of 2011 and \$133 million in the first six months of 2011 compared to the same periods of 2010. The components of amortization of DAC are summarized in the following table.

(\$ in millions)	Three m	onths ine 30,		Six months ended June 30,			
	 2011		2010	 2011		2010	
Amortization of DAC before amortization relating to realized capital gains and losses and valuation changes on embedded derivatives that are not hedged and changes in assumptions (Amortization) accretion relating to realized capital gains and losses <sup>(1)</sup> and valuation changes on embedded derivatives that are	\$ (103)	\$	(41)	\$ (203)	\$	(139)	
not hedged Amortization (acceleration) deceleration for changes in	(7)		6	(42)		3	
assumptions ("DAC unlocking")				(12)		12	
Total amortization of DAC	\$ (110)	\$	(35)	\$ (257)	\$	(124)	

<sup>(1)</sup> The impact of realized capital gains and losses on amortization of DAC is dependent upon the relationship between the assets that give rise to the gain or loss and the product liability supported by the assets. Fluctuations result from changes in the impact of realized capital gains and losses on actual and expected gross profits.

The increases of \$75 million and \$133 million in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010 were primarily due to lower amortization in the second quarter of 2010 resulting from decreased benefit spread on interest-sensitive life insurance due to the reestimation of reserves, increased amortization relating to realized capital gains and losses and, for the first six months of 2011, an unfavorable change in amortization acceleration/deceleration for changes in assumptions. DAC amortization relating to realized capital gains and losses primarily resulted from realized capital gains on sales of fixed income securities in 2011.

During the first quarter of 2011, we completed our annual comprehensive review of the profitability of our products to determine DAC balances for our interest-sensitive life, fixed annuities and other investment contracts which covers assumptions for investment returns, including capital gains and losses, interest crediting rates to policyholders, the effect of any hedges, persistency, mortality and expenses in all product lines. The review resulted in an acceleration of DAC amortization (charge to income) of \$12 million in the first quarter of 2011. Amortization acceleration of \$17 million related to interest-sensitive life insurance and was primarily due to an increase in projected expenses. Amortization deceleration of \$5 million related to equity-indexed annuities and was primarily due to an increase in projected investment margins.

In the first quarter of 2010, the review resulted in a deceleration of DAC amortization (credit to income) of \$12 million. Amortization deceleration of \$45 million related to variable life insurance and was primarily due to appreciation in the underlying separate account valuations. Amortization acceleration of \$32 million related to interest-sensitive life insurance and was primarily due to an increase in projected realized capital losses and lower projected renewal premium (which is also expected to reduce persistency), partially offset by lower expenses.

*Operating costs and expenses* decreased 5.2% and 7.2% in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010. The following table summarizes operating costs and expenses.

(\$ in millions)	Three mo	nths ended	Six mont	ths ended	
	Jun	e 30,	Jun	me 30,	
	2011	2010	2011	2010	

Non-deferrable acquisition costs	\$ 40	\$ 41	\$ 82	\$ 85
Other operating costs and expenses	70	75	137	151
Total operating costs and expenses	\$ 110	\$ 116	\$ 219	\$ 236
Restructuring and related charges	\$ 	\$ (1)	\$ (2)	\$ (1)

Non-deferrable acquisition costs decreased 2.4% or \$1 million and 3.5% or \$3 million in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010 primarily due to lower premium tax

73

expenses. Other operating costs and expenses decreased 6.7% or \$5 million and 9.3% or \$14 million in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010. The declines were primarily due to lower net Allstate agencies distribution channel expenses reflecting increased fees from sales of third party financial products, lower employee and professional service costs and lower occupancy costs due to consolidation of office buildings, partially offset by lower reinsurance expense allowances.

Loss on disposition of \$17 million in the first six months of 2011 includes \$21 million related to the planned dissolution of Allstate Bank. The previously announced agreement to sell substantially all of the deposits of Allstate Bank to Discover Bank has been terminated. We are continuing with plans to wind down the Allstate Bank's operations and anticipate obtaining regulatory approval to cancel its banking charter by year end 2011.

*Income tax expense* of \$84 million and \$130 million was recognized for the second quarter and first six months of 2011, respectively, compared to an income tax benefit of \$61 million and \$65 million, respectively, in the same periods of 2010. This change was due to the proportionate change in the income on which the income tax expense was determined.

#### **INVESTMENTS HIGHLIGHTS**

- · Investments as of June 30, 2011 totaled \$99.29 billion, a decrease of 1.2% from \$100.48 billion as of December 31, 2010.
- · Unrealized net capital gains totaled \$2.51 billion as of June 30, 2011, improving from \$1.39 billion as of December 31, 2010.
- As of June 30, 2011, the fair value for our below investment grade fixed income securities with gross unrealized losses totaled \$3.91 billion compared to \$3.29 billion as of December 31, 2010. The gross unrealized losses for these securities totaled \$831 million as of June 30, 2011, an improvement of 23.1% from \$1.08 billion as of December 31, 2010.
- Net investment income was \$1.02 billion in the second quarter of 2011, a decrease of 2.8% from \$1.05 billion in the second quarter of 2010, and \$2.00 billion in the first six months of 2011, a decrease of 4.6% from \$2.10 billion in the first six months of 2010.
- · Net realized capital gains were \$57 million in the second quarter of 2011 compared to net realized capital losses of \$451 million in the second quarter of 2010. Net realized capital gains were \$153 million in the first six months of 2011 compared to net realized capital losses of \$799 million in the first six months of 2010.
- Derivative net realized capital losses totaled \$53 million in the second quarter of 2011 compared to net realized capital losses of \$310 million in the second quarter of 2010, and net realized capital losses of \$120 million in the first six months of 2011 compared to net realized capital losses of \$495 million in the first six months of 2010.

74

#### INVESTMENTS

Improved markets in combination with our risk mitigation and return optimization strategies have strengthened our capital position, enabling us to execute yield and return enhancement strategies, while continuing to manage market risks. We modified the maturity profile of our fixed income portfolio through shifts out of longer term fixed rate and shorter term lower yielding securities into intermediate term maturity securities. Additionally, we increased our exposure to below investment grade corporate fixed income securities through a higher targeted allocation and reinvestment of proceeds from the sale of lower rated structured securities.

The composition of the investment portfolios as of June 30, 2011 is presented in the table below.

(\$ in millions)	Property-L	iability (5)	Allstate Fi	nancial (5)		Corpo and Otl		Tot	al
		Percent to total		Percent to total	·		Percent to total		Percent to total
Fixed income securities (1)	\$ 27,504	76.2%	\$ 47,861	80.2%	\$	3,049	86.8%	\$ 78,414	79.0%
Equity securities (2)	4,748	13.1	206	0.4				4,954	5.0
Mortgage loans	132	0.4	6,695	11.2				6,827	6.9
Limited partnership interests (3)	2,913	8.1	1,449	2.4		38	1.1	4,400	4.4
Short-term (4)	770	2.1	1,342	2.3		424	12.1	2,536	2.5
Other	52	0.1	2,106	3.5				2,158	2.2
Total	\$ 36,119	100.0%	\$ 59,659	100.0%	\$	3,511	100.0%	\$ 99,289	100.0%

<sup>(1)</sup> Fixed income securities are carried at fair value. Amortized cost basis for these securities was \$27.10 billion, \$46.42 billion and \$2.98 billion for Property-Liability, Allstate Financial and Corporate and Other, respectively.

Total investments decreased to \$99.29 billion as of June 30, 2011, from \$100.48 billion as of December 31, 2010, primarily due to net reductions in contractholder obligations, partially offset by higher valuations of fixed income securities and an increase of \$442 million in collateral from securities lending

<sup>&</sup>lt;sup>(2)</sup> Equity securities are carried at fair value. Cost basis for these securities was \$4.17 billion and \$159 million for Property-Liability and Allstate Financial, respectively.

<sup>(3)</sup> We have commitments to invest in additional limited partnership interests totaling \$1.01 billion and \$705 million for Property-Liability and Allstate Financial, respectively.

<sup>(4)</sup> Short-term investments are carried at fair value. Amortized cost basis for these investments was \$770 million, \$1.34 billion and \$424 million for Property-Liability, Allstate Financial and Corporate and Other, respectively.

Balances reflect the elimination of related party investments between segments.

activities. Valuations of fixed income securities are typically driven by a combination of changes in relevant risk-free interest rates and credit spreads over the period. Risk-free interest rates are typically defined as the yield on U.S. Treasury securities, whereas credit spread is the additional yield on fixed income securities above the risk-free rate that market participants require to compensate them for assuming credit, liquidity and/or prepayment risks. The increase in valuation of fixed income securities for the six months ended June 30, 2011 was due to a combination of declining risk-free interest rates and narrowing credit spreads.

The Property-Liability investment portfolio increased to \$36.12 billion as of June 30, 2011, from \$35.05 billion as of December 31, 2010, primarily due to positive operating cash flows, higher valuations of fixed income securities and increased collateral from securities lending activities, partially offset by dividends paid by Allstate Insurance Company ("AIC") to its parent, The Allstate Corporation (the "Corporation").

The Allstate Financial investment portfolio decreased to \$59.66 billion as of June 30, 2011, from \$61.58 billion as of December 31, 2010, primarily due to net reductions in contractholder obligations of \$3.11 billion, partially offset by higher valuations of fixed income securities.

The Corporate and Other investment portfolio decreased to \$3.51 billion as of June 30, 2011, from \$3.85 billion as of December 31, 2010, primarily due to share repurchases, dividends paid to shareholders and interest paid on debt, partially offset by dividends of \$438 million paid by AIC to the Corporation.

75

**Fixed income securities** by type are listed in the table below.

(\$ in millions)			Percent to			Percent to
	$\mathbf{F}_{i}$	air value as of	total	Fa	ir value as of	total
	J	June 30, 2011	investments	Dece	ember 31, 2010	investments
U.S. government and agencies	\$	6,187	6.2%	\$	8,596	8.6%
Municipal		14,673	14.8		15,934	15.9
Corporate		42,369	42.7		37,655	37.5
Foreign government		3,043	3.1		3,158	3.1
Residential mortgage-backed securities ("RMBS")		5,990	6.0		7,993	7.9
Commercial mortgage-backed securities ("CMBS")		1,986	2.0		1,994	2.0
Asset-backed securities ("ABS")		4,142	4.2		4,244	4.2
Redeemable preferred stock		24			38	
Total fixed income securities	\$	78,414	79.0%	\$	79,612	79.2%

As of June 30, 2011, 90.8% of the consolidated fixed income securities portfolio was rated investment grade, which is defined as a security having a rating of Aaa, Aa, A or Baa from Moody's, a rating of AAA, AA, A or BBB from Standard & Poor's ("S&P"), Fitch, Dominion, or Realpoint, a rating of aaa, aa, a or bbb from A.M. Best, or a comparable internal rating if an externally provided rating is not available. All of our fixed income securities are rated by third party credit rating agencies, the National Association of Insurance Commissioners ("NAIC"), and/or internally rated. Our initial investment decisions and ongoing monitoring procedures for fixed income securities are based on a thorough due diligence process which includes, but is not limited to, an assessment of the credit quality, sector, structure, and liquidity risks of each issue.

76

The following table summarizes the fair value and unrealized net capital gains and losses for fixed income securities by credit rating as of June 30, 2011.

(\$ in millions)		Aa	a	_		Aa			Α	
	Fair value		Unrealized gain/(loss)		Fair value		Unrealized gain/(loss)	Fair value		Unrealized gain/(loss)
U.S. government and agencies	\$ 6,187	\$	315	\$	value 	\$		\$ value 	\$	
Municipal Tax exempt Taxable Auction rate securities ("ARS")	1,367 198 683		82 5 (40)		4,063 2,477 59		126 83 (6)	2,383 1,047 83		40 (5) (11)
Corporate Public Privately placed	1,188 1,115		32 37		2,853 1,853		117 58	10,119 4,184		463 206
Foreign government	1,602		227		567		26	524		41
RMBS U.S. government sponsored entities ("U.S. Agency") Prime residential mortgage-backed securities ("Prime") Alt-A residential mortgage-backed securities ("Alt-A") Subprime residential mortgage-backed securities ("Subprime")	3,374 271 3 		158 5  		40 43 66		(1) (1) (19)	197 72 43		2 (1) (6)
CMBS	1,053		42		278		(6)	153		(14)
ABS Collateralized debt obligations ("CDO") Consumer and other asset-backed securities ("Consumer and other ABS")	12 1,371		 37		705 387		(4) 4	377 328		(31)
Redeemable preferred stock					1					
Total fixed income securities	\$ 18,424	\$	900	\$	13,392	\$	377	\$ 19,510	\$	687
		Ba	a		Ва	or l	ower		Tota	I

Fair

Unrealized

Fair

Unrealized

Fair

Unrealized

U.S. government and agencies	\$ \$		\$ \$	:	\$ 6,187 \$	315
Municipal Tax exempt Taxable ARS	1,202 424 70	(37) (37) (8)	501 116 	(54) (22) 	9,516 4,262 895	157 24 (65)
Corporate Public Privately placed	10,110 6,705	518 263	2,716 1,526	36 29	26,986 15,383	1,166 593
Foreign government	350	29			3,043	323
RMBS U.S. Agency Prime Alt-A Subprime	39 47 86	 1 1 (27)	541 398 770	(12) (77) (389)	3,374 1,088 563 965	158 (5) (78) (441)
CMBS	331	(58)	171	(61)	1,986	(97)
ABS CDO Consumer and other ABS	285 219	(60) (1)	411 47	(84) (3)	1,790 2,352	(179) 40
Redeemable preferred stock	23	1			24	1
Total fixed income securities	\$ 19,891 \$	585	\$ 7,197 \$	(637)	\$ 78,414 \$	1,912

*Municipal bonds*, including tax exempt, taxable and ARS securities, totaled \$14.67 billion as of June 30, 2011 with an unrealized net capital gain of \$116 million. The municipal bond portfolio includes general obligations of state and local issuers, revenue bonds and pre-refunded bonds, which are bonds for which an irrevocable trust has been established to fund the remaining payments of principal and interest.

77

The following table summarizes by state the fair value, amortized cost and credit rating of our municipal bonds, excluding \$1.66 billion of pre-refunded bonds, as of June 30, 2011.

(\$ in millions)	State general	Local general		- 0			Amortized	Average credit
State	 obligation	obligation	_	Revenue (1)	_	Fair value	 cost	rating (2)
California	\$ 73 \$	752	\$	685	\$	1,510	\$ 1,569	A
Texas	17	435		614		1,066	1,051	Aa
Florida	54	207		549		810	799	A
New York	33	26		525		584	573	Aa
Ohio	39	205		292		536	544	A
Illinois		166		347		513	508	A
Missouri	32	157		279		468	468	A
Michigan	42	138		280		460	460	Aa
Pennsylvania	94	116		212		422	420	Aa
Delaware				412		412	438	Aa
All others	878	1,404		3,945		6,227	6,186	A
Total	\$ 1,262 \$	3,606	\$	8,140	\$	13,008	\$ 13,016	A

<sup>(1)</sup> The nature of the activities supporting revenue municipals is highly diversified and includes transportation, health care, industrial development, housing, higher education, utilities, recreation/convention centers and other activities.

Our practice for acquiring and monitoring municipal bonds is predominantly based on the underlying credit quality of the primary obligor. We currently rely on the primary obligor to pay all contractual cash flows and are not relying on bond insurers for payments. As a result of downgrades in the insurers' credit ratings, the ratings of the insured municipal bonds generally reflect the underlying ratings of the primary obligor. As of June 30, 2011, 99.4% of our insured municipal bond portfolio is rated investment grade. Given the effects of the economic crisis on bond insurers, the value inherent in the insurance has declined. Further, we believe the fair value of our insured municipal bond portfolio substantially reflects the decline in the value of the insurance. We believe that the loss of the benefit of insurance would not result in a material adverse impact on our results of operations, financial position or liquidity.

Included in our municipal bond holdings as of June 30, 2011 are \$869 million of municipal securities which are not rated by third party credit rating agencies, but are rated by the NAIC and are also internally rated. These holdings include \$424 million of below investment grade municipal bonds, most of which were purchased to provide the opportunity to achieve incremental returns.

*Corporate bonds*, including publicly traded and privately placed, totaled \$42.37 billion as of June 30, 2011 with an unrealized net capital gain of \$1.76 billion. Privately placed securities primarily consist of corporate issued senior debt securities that are in unregistered form or are directly negotiated with the borrower.

RMBS, CMBS and ABS are structured securities that are primarily collateralized by residential and commercial real estate loans and other consumer or corporate borrowings. The cash flows from the underlying collateral paid to the securitization trust are generally applied in a pre-determined order and are designed so that each security issued by the trust, typically referred to as a "class", qualifies for a specific original rating. For example, the "senior" portion or "top" of the capital structure, or rating class, which would originally qualify for a rating of Aaa typically has priority in receiving principal repayments on the underlying collateral and retains this priority until the class is paid in full. In a sequential structure, underlying collateral principal repayments are directed to the most senior rated Aaa class in the structure until paid in full, after which principal repayments are directed to the next most senior Aaa class in the structure until it is paid in full. Senior Aaa classes generally share any losses from the underlying collateral on a pro-rata basis after losses are absorbed by classes with lower original ratings. The payment priority and class subordination included in these securities serves as credit enhancement for holders of the

<sup>(2)</sup> The municipal bonds are rated by third party credit rating agencies, the NAIC and/or internally rated.

in the securitization trust, such as overcollateralization, excess spread and bond insurance. The underlying collateral can have fixed interest rates, variable interest rates (such as adjustable rate mortgages ("ARM")) or may contain features of both fixed and variable rate mortgages.

*RMBS*, including U.S. Agency, Prime, Alt-A and Subprime, totaled \$5.99 billion, with 71.5% rated investment grade, as of June 30, 2011. The RMBS portfolio is subject to interest rate risk, but unlike other fixed income securities, is additionally subject to significant prepayment risk from the underlying residential mortgage loans. The credit risk associated with our RMBS portfolio is partially mitigated due to the fact that 56.3% of the portfolio consists of securities that were issued by or have underlying collateral guaranteed by U.S. government agencies. The unrealized net capital loss of \$366 million as of June 30, 2011 was the result of wider credit spreads than at initial purchase on the non-U.S. Agency portion of our RMBS portfolio, largely due to higher risk premiums caused by macroeconomic conditions and credit market deterioration, including the impact of lower real estate valuations, which show signs of stabilization or recovery in certain geographic areas but remain under stress in other geographic areas. The following table shows our RMBS portfolio as of June 30, 2011 based upon vintage year of the issuance of the securities.

(\$ in millions)		U.	S. Ag	gency		Pri	ne	A	dt-A	Subp	rime	Total RMBS		RMBS
	_	Fair value		Unrealized gain/(loss)	Fair value		Unrealized gain/(loss)	Fair value	Unrealized gain/(loss)	Fair value	Unrealized gain/(loss)	Fair value		Unrealized gain/(loss)
2010	\$	169	\$	2	\$ 187	\$	2	\$ 56 \$		\$ \$		\$ 412	\$	4
2009		651		20	66		1	8				725		21
2008		561		21								561		21
2007		222		8	243		5	72	(32)	231	(140)	768		(159)
2006		201		10	193			164	(16)	242	(122)	800		(128)
2005		429		22	175		(13)	117	(15)	263	(110)	984		(116)
Pre-2005		1,141		75	224			146	(15)	229	(69)	1,740		(9)
Total	\$	3,374	\$	158	\$ 1,088	\$	(5)	\$ 563 \$	(78)	\$ 965 \$	(441)	\$ 5,990	\$	(366)

Prime are collateralized by residential mortgage loans issued to prime borrowers. As of June 30, 2011, \$815 million of the Prime had fixed rate underlying collateral and \$273 million had variable rate underlying collateral.

Alt-A includes securities collateralized by residential mortgage loans issued to borrowers who do not qualify for prime financing terms due to high loan-to-value ratios or limited supporting documentation, but have stronger credit profiles than subprime borrowers. As of June 30, 2011, \$423 million of the Alt-A had fixed rate underlying collateral and \$140 million had variable rate underlying collateral.

Subprime includes securities collateralized by residential mortgage loans issued to borrowers that cannot qualify for Prime or Alt-A financing terms due in part to weak or limited credit history. It also includes securities that are collateralized by certain second lien mortgages regardless of the borrower's credit history. The Subprime portfolio consisted of \$713 million and \$252 million of first lien and second lien securities, respectively. As of June 30, 2011, \$532 million of the Subprime had fixed rate underlying collateral and \$433 million had variable rate underlying collateral.

CMBS totaled \$1.99 billion, with 91.4% rated investment grade, as of June 30, 2011. The CMBS portfolio is subject to credit risk, but unlike certain other structured securities, is generally not subject to prepayment risk due to protections within the underlying commercial mortgage loans. Of the CMBS investments, 94.6% are traditional conduit transactions collateralized by commercial mortgage loans, broadly diversified across property types and geographical area. The remainder consists of non-traditional CMBS such as small balance transactions, large loan pools and single borrower transactions.

79

The following table shows our CMBS portfolio as of June 30, 2011 based upon vintage year of the underlying collateral.

(\$ in millions)	Fair value	Unrealized gain/(loss)
2011	\$ 1	\$ 
2010	24	1
2007	280	(13)
2006	584	(81)
2005	326	(23)
Pre-2005	771	19
Total CMBS	\$ 1,986	\$ (97)

The unrealized net capital loss of \$97 million as of June 30, 2011 on our CMBS portfolio was the result of wider credit spreads than at initial purchase, largely due to the macroeconomic conditions and credit market deterioration, including the impact of lower real estate valuations, which show signs of stabilization or recovery in certain geographic areas but remain under stress in other geographic areas. CMBS credit spreads in most rating classes remain wider than at initial purchase, which is particularly evident in our 2005-2007 vintage year CMBS.

ABS, including CDO and Consumer and other ABS, totaled \$4.14 billion, with 88.9% rated investment grade, as of June 30, 2011. Credit risk is managed by monitoring the performance of the underlying collateral. Many of the securities in the ABS portfolio have credit enhancement with features such as overcollateralization, subordinated structures, reserve funds, guarantees and/or insurance. The unrealized net capital loss of \$139 million as of June 30, 2011 on our ABS portfolio was the result of wider credit spreads than at initial purchase.

CDO totaled \$1.79 billion, with 77.0% rated investment grade, as of June 30, 2011. CDO consist primarily of obligations collateralized by high yield and investment grade corporate credits including \$1.40 billion of cash flow collateralized loan obligations ("CLO") with unrealized losses of \$88 million.

The remaining \$395 million of securities consisted of synthetic CDO, trust preferred CDO, project finance CDO, market value CDO, collateralized bond obligations and other CLO with unrealized losses of \$91 million.

Consumer and other ABS totaled \$2.35 billion, with 98.0% rated investment grade, as of June 30, 2011. Consumer and other ABS consists of \$878 million of consumer auto and \$1.47 billion of other ABS with unrealized gains of \$6 million and \$34 million, respectively.

**Mortgage loans** Our mortgage loan portfolio, which is primarily held in the Allstate Financial portfolio, totaled \$6.83 billion as of June 30, 2011, compared to \$6.68 billion as of December 31, 2010, and primarily comprises loans secured by first mortgages on developed commercial real estate. Key considerations used to manage our exposure include property type and geographic diversification.

We recognized \$7 million and \$13 million of realized capital losses related to net increases in the valuation allowance on impaired mortgage loans for the three months and six months ended June 30, 2011, respectively, primarily due to the risk associated with refinancing near-term maturities, and decreases in occupancy which resulted in deteriorating debt service coverage and declines in property valuations. While property valuations show signs of stabilization or recovery in many larger, primary markets, valuations in many smaller cities remain under stress.

For further detail on our mortgage loan portfolio, see Note 4 of the condensed consolidated financial statements.

80

**Limited partnership interests** consist of investments in private equity/debt funds, real estate funds, hedge funds and tax credit funds. The limited partnership interests portfolio is well diversified across a number of characteristics including fund sponsors, vintage years, strategies, geography (including international), and company/property types. The following table presents information about our limited partnership interests as of June 30, 2011.

(\$ in millions)	Private equity/debt funds	Real estate funds	Hedge funds	Tax credit funds	Total
Cost method of accounting ("Cost")	\$ 977	\$ 345	\$ 79	\$ 7	\$ 1,408
Equity method of accounting ("EMA")	715	559	1,346	372	2,992
Total	\$ 1,692	\$ 904	\$ 1,425	\$ 379	\$ 4,400
Number of sponsors	92	44	12	7	
Number of individual funds	148	89	92	11	
Largest exposure to single fund	\$ 37	\$ 135	\$ 81	\$ 60	

Our aggregate limited partnership exposure represented 4.4% and 3.8% of total invested assets as of June 30, 2011 and December 31, 2010, respectively.

The following table shows the results from our limited partnership interests by fund type and accounting classification.

(\$ in millions)							Three mo	onth ne 3						
					201	1						201	10	<u> </u>
		Cost		EMA		Total income	Impairment write-downs (1)		Cost		EMA		Total income	Impairment write- downs (1)
Private equity/debt funds	\$	18	\$	33	\$	51	\$ (1)	\$	8	\$	20	\$	28	\$ 
Real estate funds				19		19			1		(8)		(7)	(8)
Hedge funds				5		5					8		8	
Tax credit funds	_		_	(2)	_	(2)	 		(2)	_		_	(2)	
Total	\$ _	18	\$	55	\$ _	73	\$ (1)	\$	7	\$	20	\$	27	\$ (8)

						Ju	ne 3						
	_			20	11					20	10		
	_	Cost	EMA		Total income	Impairment write-downs (1)	_	Cost	EMA		Total income		Impairment write-downs (1)
Private equity/debt funds	\$	27	\$ 52	\$	79	\$ (2)	\$	14	\$ 35	\$	49	\$	(2)
Real estate funds		1	27		28			1	(36)		(35)		(29)
Hedge funds			41		41				25		25		(1)
Tax credit funds	_		(2)		(2)	 		(2)		_	(2)	_	
Total	\$ _	28	\$ 118	\$	146	\$ (2)	\$	13	\$ 24	\$	37	\$	(32)

Six months ended

Limited partnership interests, excluding impairment write-downs, produced income of \$73 million and \$146 million in the three months and six months ended June 30, 2011, respectively, compared to \$27 million and \$37 million in the three months and six months ended June 30, 2010, respectively. Income on EMA limited partnerships is recognized on a delay due to the availability of the related financial statements. The recognition of income on hedge funds is primarily on a one-month delay and the income recognition on private equity/debt funds, real estate funds and tax credit funds are generally on a three-month delay. Income on Cost limited partnerships is recognized only upon receipt of amounts distributed by the partnerships.

<sup>(1)</sup> Impairment write-downs related to Cost limited partnerships were \$1 million and \$2 million in the three months and six months ended June 30, 2011, respectively, compared to \$7 million and \$31 million in the three months and six months ended June 30, 2010, respectively. There were no impairment write-downs related to EMA limited partnerships in the three months and six months ended June 30, 2011, compared to \$1 million in both the three months and six months ended June 30, 2010.

credit spreads. The improvement since December 31, 2010 for equity securities was primarily due to improved equity valuations. The following table presents unrealized net capital gains and losses, pre-tax.

(\$ in millions)	June 30, 2011	March 31, 2011	December 31, 2010
U.S. government and agencies	\$ 315	\$ 257	\$ 276
Municipal	116	(254)	(267)
Corporate	1,759	1,300	1,395
Foreign government	323	295	337
RMBS	(366)	(377)	(516)
CMBS	(97)	(103)	(219)
ABS	(139)	(169)	(181)
Redeemable preferred stock	1	1	1
Fixed income securities (1)	 1,912	950	826
Equity securities	625	645	583
EMA limited partnership interests	7	7	
Derivatives	(36)	(30)	(22)
Unrealized net capital gains and losses, pre-tax	\$ 2,508	\$ 1,572	\$ 1,387

<sup>(1)</sup> Unrealized net capital gains and losses for fixed income securities as of June 30, 2011, March 31, 2011 and December 31, 2010 comprise \$(240) million, \$(257) million and \$(293) million, respectively, related to unrealized net capital losses on fixed income securities with other-than-temporary impairment and \$2.15 billion, \$1.21 billion and \$1.12 billion, respectively, related to other unrealized net capital gains and losses.

The unrealized net capital gains for the fixed income portfolio totaled \$1.91 billion and comprised \$3.50 billion of gross unrealized gains and \$1.59 billion of gross unrealized losses as of June 30, 2011. This is compared to unrealized net capital gains for the fixed income portfolio totaling \$826 million, comprised of \$3.26 billion of gross unrealized gains and \$2.43 billion of gross unrealized losses as of December 31, 2010.

82

Gross unrealized gains and losses as of June 30, 2011 on fixed income securities by type and sector are provided in the table below.

(\$ in millions)			_	Gross 1	ınrea	lized			Amortized cost as a		Fair value as a			
		Par value (1)	_	Amortized cost	_	Gains	_	Losses	_	Fair value	percent of par value (2)	_	percent of par value <sup>(2)</sup>	
Corporate:														
Banking	\$	4,414	\$	4,330	\$	144	\$	(90)	\$	4,384	98.1	%	99.3 %	%
Utilities		6,732		6,754		475		(45)		7,184	100.3		106.7	
Consumer goods (cyclical and non-														
cyclical)		7,296		7,407		329		(31)		7,705	101.5		105.6	
Capital goods		4,751		4,765		275		(22)		5,018	100.3		105.6	
Financial services		3,899		3,849		149		(21)		3,977	98.7		102.0	
Transportation		1,883		1,901		110		(14)		1,997	101.0		106.1	
Basic industry		2,019		2,036		104		(9)		2,131	100.8		105.5	
Communications		2,752		2,757		121		(8)		2,870	100.2		104.3	
Technology		1,751		1,769		67		(6)		1,830	101.0		104.5	
Energy		3,047		3,089		163		(3)		3,249	101.4		106.6	
FDIC guaranteed		157		158		2				160	100.6		101.9	
Other	_	1,890	_	1,795	_	75	_	(6)	_	1,864	95.0		98.6	
Total corporate fixed income portfolio	_	40,591	-	40,610	_	2,014	_	(255)	_	42,369	100.0		104.4	
U.S. government and agencies		6,413		5,872		318		(3)		6,187	91.6		96.5	
Municipal		16,746		14,557		491		(375)		14,673	86.9		87.6	
Foreign government		3,079		2,720		327		(4)		3,043	88.3		98.8	
RMBS		7,058		6,356		203		(569)		5,990	90.1		84.9	
CMBS		2,102		2,083		57		(154)		1,986	99.1		94.5	
ABS		4,521		4,281		95		(234)		4,142	94.7		91.6	
Redeemable preferred stock	_	21	_	23	_	1	_		_	24	109.5		114.3	
Total fixed income securities	\$ _	80,531	\$ _	76,502	\$ _	3,506	\$_	(1,594)	\$ _	78,414	95.0		97.4	

<sup>(1)</sup> Included in par value are zero-coupon securities that are generally purchased at a deep discount to the par value that is received at maturity. These primarily included corporate, U.S. government and agencies, municipal and foreign government zero-coupon securities with par value of \$534 million, \$1.43 billion, \$3.63 billion and \$1.12 billion, respectively.

The banking, utilities, consumer goods, capital goods and financial services sectors had the highest concentration of gross unrealized losses in our corporate fixed income securities portfolio as of June 30, 2011. In general, credit spreads remain wider than at initial purchase for most of the securities with gross unrealized losses in these categories.

The unrealized net capital gain for the equity portfolio totaled \$625 million and comprised \$705 million of gross unrealized gains and \$80 million of gross unrealized losses as of June 30, 2011. This is compared to an unrealized net capital gain for the equity portfolio totaling \$583 million, comprised of \$646 million of gross unrealized gains and \$63 million of gross unrealized losses as of December 31, 2010.

We have a comprehensive portfolio monitoring process to identify and evaluate each fixed income and equity security that may be other-than-temporarily impaired. The process includes a quarterly review of all securities to identify instances where the fair value of a security compared to its amortized cost (for fixed income securities) or cost (for equity securities) is below established thresholds. The process also includes the monitoring of other impairment indicators such as ratings, ratings downgrades and payment defaults. The securities identified, in addition to other securities for which we may have a concern, are evaluated based on facts and circumstances for inclusion on our watch-list. All investments in an unrealized loss position as of June 30, 2011 were included in our portfolio monitoring process for determining whether declines in value were other than temporary.

Excluding the impact of zero-coupon securities, the percentage of amortized cost to par value would be 100.4% for corporates, 101.2% for U.S. government and agencies, 99.9% for municipals and 103.6% for foreign governments. Similarly, excluding the impact of zero-coupon securities, the percentage of fair value to par value would be 104.7% for corporates, 104.6% for U.S. government and agencies, 101.1% for municipals and 110.1% for foreign governments.

The extent and duration of a decline in fair value for fixed income securities have become less indicative of actual credit deterioration with respect to an issue or issuer. While we continue to use declines in fair value and the length of time a security is in an unrealized loss position as indicators of potential credit deterioration, our determination of whether a security's decline in fair value is other than temporary has placed greater emphasis on our analysis of the underlying credit and collateral and related estimates of future cash flows.

The following table summarizes the fair value and gross unrealized losses of fixed income securities by type and investment grade classification as of June 30, 2011.

(\$ in millions)	Investr	nent	grade	Below invo	estm	ent grade		Tota	i
	Fair		Unrealized	Fair		Unrealized	 Fair		Unrealized
	value		losses	value		losses	value		losses
U.S. government and agencies	\$ 198	\$	(3)	\$ 	\$		\$ 198	\$	(3)
Municipal	3,729		(286)	478		(89)	4,207		(375)
Corporate	4,879		(192)	1,653		(63)	6,532		(255)
Foreign government	114		(4)				114		(4)
RMBS	487		(59)	1,219		(510)	1,706		(569)
CMBS	763		(91)	160		(63)	923		(154)
ABS	1,281		(128)	395		(106)	1,676		(234)
Total	\$ 11,451	\$	(763)	\$ 3,905	\$	(831)	\$ 15,356	\$	(1,594)

We have experienced declines in the fair values of fixed income securities primarily due to wider credit spreads resulting from higher risk premiums since the time of initial purchase, largely due to macroeconomic conditions and credit market deterioration, including the impact of lower real estate valuations, which show signs of stabilization or recovery in certain geographic areas but remain under stress in other geographic areas. Consistent with their ratings, our portfolio monitoring process indicates that investment grade securities have a low risk of default. Securities rated below investment grade, comprising securities with a rating of Ba, B and Caa or lower, have a higher risk of default.

As of June 30, 2011, 58% of our below investment grade gross unrealized losses were concentrated in RMBS, specifically Alt-A and Subprime. The fair value of these securities totaled \$1.02 billion, a decrease of 5.7%, compared to \$1.08 billion as of December 31, 2010. Gross unrealized losses on these securities totaled \$478 million as of June 30, 2011, an improvement of 13.7%, compared to \$554 million as of December 31, 2010, due to impairment write-downs, principal collections, sales and improved valuations, partially offset by the downgrade of certain securities to below investment grade. In addition, as of June 30, 2011, the fair value of our below investment grade CMBS with gross unrealized losses totaled \$160 million compared to \$135 million as of December 31, 2010. As of June 30, 2011, gross unrealized losses for our below investment grade CMBS portfolio totaled \$63 million, an improvement of 53.0% from \$134 million as of December 31, 2010, due to sales, improved valuations and impairment write-downs, partially offset by the downgrade of certain securities to below investment grade.

Fair values for our structured securities are obtained from third-party valuation service providers and are subject to review as disclosed in our Application of Critical Accounting Estimates. In accordance with GAAP, when fair value is less than the amortized cost of a security and we have not made the decision to sell the security and it is not more likely than not we will be required to sell the security before recovery of its amortized cost basis, we evaluate if we expect to receive cash flows sufficient to recover the entire amortized cost basis of the security. We calculate the estimated recovery value by discounting our best estimate of future cash flows at the security's original or current effective rate, as appropriate, and compare this to the amortized cost of the security. If we do not expect to receive cash flows sufficient to recover the entire amortized cost basis of the security, the credit loss component of the impairment is recorded in earnings, with the remaining amount of the unrealized loss related to other factors ("non-credit-related") recognized in other comprehensive income.

The non-credit-related unrealized losses for our structured securities, including our below investment grade Alt-A and Subprime, are heavily influenced by risk factors other than those related to our best estimate of future cash flows. The difference between these securities' original or current effective rates and the yields implied by their fair value indicates that a higher risk premium is included in the valuation of these securities than existed at initial issue or purchase. This risk premium represents the return that a market participant requires as compensation to assume the risk associated with the uncertainties regarding the future performance of the underlying collateral. The risk premium is comprised of: default risk, which reflects the probability of default and the uncertainty related to collection of contractual principal and interest; liquidity risk, which reflects the risk associated with exiting the

84

investment in an illiquid market, both in terms of timeliness and cost; and volatility risk, which reflects the potential valuation volatility during an investor's holding period. Other factors reflected in the risk premium include the costs associated with underwriting, monitoring and holding these types of complex securities. Certain aspects of the default risk are included in the development of our best estimate of future cash flows, as appropriate. Other aspects of the risk premium are considered to be temporary in nature and are expected to reverse over the remaining lives of the securities as future cash flows are received.

Other-than-temporary impairment assessment for below investment grade Alt-A and Subprime RMBS

As of June 30, 2011, the fair value of our below investment grade Alt-A securities with gross unrealized losses totaled \$318 million, an increase of 10.4% compared to \$288 million as of December 31, 2010. As of June 30, 2011, gross unrealized losses for our below investment grade Alt-A portfolio totaled \$87 million, an improvement of 25.0% compared to \$116 million as of December 31, 2010, due to impairment write-downs, improved valuations, principal collections and sales, partially offset by downgrade of certain securities to below investment grade. For our below investment grade Alt-A securities with gross unrealized gains of \$11 million, we have recognized cumulative write-downs in earnings totaling \$52 million as of June 30, 2011.

As of June 30, 2011, the fair value of our below investment grade Subprime securities with gross unrealized losses totaled \$704 million, a decrease of 11.6% compared to \$796 million as of December 31, 2010. As of June 30, 2011, gross unrealized losses for our below investment grade Subprime portfolio

totaled \$391 million, an improvement of 10.7% compared to \$438 million as of December 31, 2010, due to impairment write-downs, principal collections, sales and improved valuations, partially offset by downgrade of certain securities to below investment grade. For our below investment grade Subprime with gross unrealized gains totaling \$3 million, we have recognized cumulative write-downs in earnings totaling \$84 million as of June 30, 2011.

The credit loss evaluation for Alt-A and Subprime securities with gross unrealized losses is performed in two phases. The first phase estimates the future cash flows of the entire securitization trust from which our security was issued. A critical part of this estimate involves forecasting default rates and loss severities of the residential mortgage loans that collateralize the securitization trust. The factors that affect the default rates and loss severities include, but are not limited to, historical collateral performance, collateral type, transaction vintage year, geographic concentrations, borrower credit quality, origination practices of the transaction sponsor, and practices of the mortgage loan servicers. Current loan-to-value ratios of underlying collateral are not consistently available and accordingly they are not a primary factor in our impairment evaluation. While our projections are developed internally and customized to our specific holdings, they are informed by and benchmarked against credit opinions obtained from third parties, such as industry analysts, nationally recognized credit rating agencies and an RMBS loss modeling advisory service. The default rate and loss severity forecasts result in an estimate of trust-level projected additional collateral loss.

We then analyze the actual cumulative collateral losses incurred to date by the securitization trust, our projected additional collateral losses expected to be incurred and the position of the class of securities we own in the securitization trust relative to the trust's other classes to determine whether any of the collateral losses will be applied to our class. If our class has remaining credit enhancement sufficient to withstand the projected additional collateral losses, no collateral losses will be realized by our class and we expect to collect all contractual principal and interest of the security we own. Remaining credit enhancement is measured in terms of (i) subordination from other classes of securities in the trust that are contractually obligated to absorb losses before the class of security we own and (ii) the expected impact of other structural features embedded in the securitization trust beneficial to our class, such as overcollateralization and excess spread.

For securities where there is insufficient remaining credit enhancement for the class of securities we own, a recovery value is calculated based on our best estimate of future cash flows specific to that security. This estimate is based on the contractual principal payments and current interest payments of the securities we own, adjusted for actual cumulative collateral losses incurred to date and the projected additional collateral losses expected to be incurred. This estimate also takes into consideration additional secondary sources of credit support, such as reliable bond insurance. For securities without secondary sources of credit support or for which the secondary sources do not fully offset the actual and projected additional collateral losses applied to them, a credit loss is recorded in earnings to the extent amortized cost exceeds recovery value.

96.4% and 3.6% of the fair value of our below investment grade Alt-A securities with gross unrealized losses were issued with Aaa and Aa original ratings and capital structure classifications, respectively.74.6%, 21.7% and

85

3.7% of the fair value of our below investment grade Subprime securities with gross unrealized losses were issued with Aaa, Aa and A original ratings and capital structure classifications, respectively. As described previously, Alt-A and Subprime securities with higher original ratings typically have priority in receiving the principal repayments on the underlying collateral compared to those with lower original ratings. While the projected cash flow assumptions for our below investment grade Alt-A and Subprime securities with gross unrealized losses have deteriorated since the securities were originated, as reflected by their current credit ratings, these securities continue to retain the payment priority features that existed at the origination of the securitization trust.

The following tables show trust-level, class-level and security-specific detailed information for our below investment grade Alt-A securities with gross unrealized losses, by credit rating.

(\$ in millions)						June	30,	2011						
	impai	ner-than-te rments rec in earnings	orde							an-temporai ded in earni			_	
	В	Caa or lower		Total		Ba		В		Caa or lower		Total		Total
Trust-level			_		_		_		_				_	
Actual cumulative collateral losses incurred to date	1.3 %	9.3	%	9.1	%	1.6	%	3.2	%	4.5	%	2.8	%	n/a
Projected additional collateral losses to be incurred	12.3 %	24.2	%	24.0	%	10.3	%	12.4	%	23.7	%	14.9	%	n/a
Class-level														
Average remaining credit enhancement (3)	12.0 %	5.6	%	5.7	%	14.6	%	17.6	%	25.4	%	18.6	%	n/a
Security-specific														
Number of positions	1	31		32		6		5		5		16		48
Par value	\$ 9 \$	427	\$	436	\$	60	\$	20	\$	38	\$	118	\$	554
Amortized cost	\$ 9 \$	284	\$	293	\$	60	\$	20	\$	32	\$	112	\$	405
Fair value	\$ 9 \$	214	\$	223	\$	57	\$	14	\$	24	\$	95	\$	318
Gross unrealized losses														
Total	\$ \$	(70)	\$	(70)	\$	(3)	\$	(6)	\$	(8)	\$	(17)	\$	(87)
12-24 months (4)	\$ \$		\$		\$		\$		\$		\$		\$	
Over 24 months (5)	\$ \$	(67)	\$	(67)	\$	(3)	\$	(5)	\$	(6)	\$	(14)	\$	(81)
Cumulative write-downs recognized (6)	\$ \$	(110)	\$	(110)	\$		\$		\$		\$	`	\$	(110)
Principal payments received during the period (7)													\$	32

		im	With other-t pairments re		-temporary led in earnings						an-temporary ded in earnings			
					Caa or						Caa or			
	Ba		В		lower	Total		Ba	В		lower	Total		Total
Trust-level				_						_				
Actual cumulative collateral losses incurred to date	(	).5 %	0.7	%	8.1 %	8.0	%	0.1 %	3.1	%	3.7 %	3.0	%	n/a
Projected additional collateral losses to be incurred		9.9 %	22.5	%	24.6 %	24.5	%	4.8 %	16.6	%	17.8 %	15.4	%	n/a
Class-level														
Average remaining credit enhancement (3)	9	9.9 %	19.0	%	6.8 %	6.9	%	5.3 %	27.1	%	23.9 %	20.7	%	n/a
Security-specific														
Number of positions		1	1		27	29		2	2		8	12		41
Par value	\$	4 \$	3		439 \$	446	\$	16 \$	4	\$	68 \$	88	\$	534
Amortized cost	\$	4 \$	2	\$	316 \$	322	\$	16 \$	4	\$	62 \$	82	\$	404
Fair value	\$	1 \$	1	\$	220 \$	222	\$	13 \$	2	\$	51 \$	66	\$	288
Gross unrealized losses	¢.	(D) (f)	(1)	σ	(OC) ¢	(100)	d.	(D) d	(2)	ф	(11) f	(10)	d.	(116)
Total	\$	(3) \$	(1)	Ф	(96) \$	(100)	Э	(3) \$	(2)	<b>Þ</b>	(11) \$	(16)	Ъ	(116)

December 31, 2010

12-24 months <sup>(4)</sup> Over 24 months <sup>(5)</sup> Cumulative write-downs recognized <sup>(6)</sup> Principal payments received during the period <sup>(7)</sup>	\$ \$ \$	\$ (3) \$ \$	\$ (1) \$ (1) \$	\$ (90) \$ (92) \$	\$ (94) \$ (93) \$	\$ (3) \$ \$	\$ (2) \$ \$	\$ (10) \$ \$	\$ (15) \$ \$	(109 (93
Principal payments received during the period (7)									\$	67

(1) Weighted average actual cumulative collateral losses incurred to date as of period end are based on the actual principal losses incurred as a percentage of the remaining principal amount of the loans in the trust. The weighting calculation is based on the par value of each security. Actual losses on the securities we hold are less than the losses on the underlying collateral as presented in this table. Actual cumulative realized principal losses on the below investment grade Alt-A securities we own, as reported by the trust servicers, were \$9 million as of June 30. 2011.

86

- Weighted average projected additional collateral losses to be incurred as of period end are based on our projections of future losses to be incurred by the trust, taking into consideration the actual cumulative collateral losses incurred to date, as a percentage of the remaining principal amount of the loans in the trust. Our projections are developed internally and customized to our specific holdings and are informed by and benchmarked against credit opinions obtained from third parties, such as industry analysts, nationally recognized credit rating agencies and an RMBS loss modeling advisory service. Projected additional collateral losses to be incurred are compared to average remaining credit enhancement for each security. For securities where the projected additional collateral losses exceed remaining credit enhancement, a recovery value is calculated to determine whether impairment losses should be recorded in earnings. The weighting calculation is based on the par value of each security.
- (3) Weighted average remaining credit enhancement as of period end is based on structural subordination and the expected impact of other structural features existing in the securitization trust beneficial to our class and reflects our projection of future principal losses that can occur as a percentage of the remaining principal amount of the loans in the trust before the class of the security we own will incur its first dollar of principal loss. The weighting calculation is based on the par value of each security.
- (4) Includes total gross unrealized losses on securities in an unrealized loss position for a period of 12 to 24 consecutive months.
- Includes total gross unrealized losses on securities in an unrealized loss position for a period more than 24 consecutive months. As of June 30, 2011, \$43 million of unrealized losses on securities with other-than-temporary impairments recognized in earnings and \$11 million of unrealized losses on securities without other-than-temporary impairments recognized in earnings have been greater than or equal to 20% of those securities' amortized cost for a period of more than 24 consecutive months. As of December 31, 2010, \$70 million of unrealized losses on securities with other-than-temporary impairments recognized in earnings and \$11 million of unrealized losses on securities without other-than-temporary impairments recognized in earnings have been greater than or equal to 20% of those securities' amortized cost for a period of more than 24 consecutive months.
- (6) Includes cumulative write-downs recorded in accordance with GAAP.
- (7) Reflects principal payments for the six months ended June 30, 2011 or the year ended December 31, 2010, respectively.

The above tables include information about our below investment grade Alt-A securities with gross unrealized losses as of each period presented. The par value and composition of securities included can vary significantly from period to period due to changes in variables such as credit ratings, principal payments, sales, purchases and realized principal losses.

As of June 30, 2011, our below investment grade Alt-A securities with gross unrealized losses and without other-than-temporary impairments recorded in earnings had incurred actual cumulative collateral losses of 2.8%. Our impairment evaluation forecasts more severe assumptions than the trusts are actually experiencing, including a projected weighted average underlying default rate of 28.1% and a projected weighted average loss severity of 51.2%, which resulted in projected additional collateral losses of 14.9%. As the average remaining credit enhancement for these securities of 18.6% exceeds the projected additional collateral losses of 14.9%, these securities have not been impaired.

As of June 30, 2011, our below investment grade Alt-A securities with gross unrealized losses and with other-than-temporary impairments recorded in earnings had incurred actual cumulative collateral losses of 9.1%. Our impairment evaluation forecasts more severe assumptions than the trusts are actually experiencing, including a projected weighted average underlying default rate of 43.4% and a projected weighted average loss severity of 54.6%, which resulted in projected additional collateral losses of 24.0%. As the average remaining credit enhancement for these securities of 5.7% is insufficient to withstand the projected additional collateral losses, we have recognized cumulative write-downs in earnings on these securities as reflected in the table above using our calculated recovery value at the time of impairment. The current average recovery value of these securities as a percentage of par was 68.7% and exceeded these securities' current average amortized cost as a percentage of par of 67.3%, which demonstrates our conclusion that the nature of the remaining unrealized loss on these securities is temporary and will reverse over time. The comparison indicates that recovery value exceeds amortized cost based on a comprehensive evaluation of financial, economic and capital markets assumptions developed for this reporting period.

We believe the unrealized losses on our Alt-A securities, including those over 24 months, result from the current risk premium on these securities, which should continue to reverse over the securities' remaining lives, as demonstrated by improved valuations in 2010 and 2011. We expect to receive our estimated share of contractual principal and interest collections used to determine the securities' recovery value. As of June 30, 2011, we do not have the intent to sell and it is not more likely than not we will be required to sell these securities before the recovery of their amortized cost basis. We believe that our valuation and impairment processes are comprehensive, employ the most current views about collateral and securitization trust financial positions, and demonstrate our recorded impairments and that the remaining unrealized losses on these positions are temporary.

87

The following tables show trust-level, class-level and security-specific detailed information for our below investment grade Subprime securities with gross unrealized losses that are not reliably insured, by credit rating.

(\$ in millions)								June 30	), 201	1						
·				r-than-temp recorded in								n-temporar ed in earnir				
_		В		Caa or lower		Total	_	Ba	-	В		Caa or lower		Total	_	Total
Trust-level			_								_		_		_	
Actual cumulative collateral losses incurred to date		14.5	%	17.9	%	17.8	%	3.1	%	6.4	%	11.1	%	8.1	%	n/a
Projected additional collateral losses to be incurred Class-level		38.6	%	42.1	%	42.0	%	32.3	%	31.6	%	35.9	%	34.1	%	n/a
Average remaining credit enhancement Security-specific		26.7	%	20.0	%	20.3	%	47.8	%	44.2	%	43.2	%	44.5	%	n/a
Number of positions		5		72		77		9		13		32		54		131
Par value	\$	40	\$	865	\$	905	\$	85	\$	72	\$	183	\$	340	\$	1,245
Amortized cost	\$	33	\$	589	\$	622	\$	85	\$	72	\$	183	\$	340	\$	962
Fair value	\$	23	\$	382	\$	405	\$	63	\$	47	\$	95	\$	205	\$	610
Gross unrealized losses																
Total	\$	(10)	\$	(207)	\$	(217)	\$	(22)	\$	(25)	\$	(88)	\$	(135)	\$	(352)
12-24 months	\$	`	\$	`	\$	`	\$	`	\$	`´	\$	`	\$	`′	\$	`′
Over 24 months (2)	<b>©</b>	(10)	¢	(207)	¢	(217)	¢	(22)	Φ.	(25)	Φ	(88)	Φ.	(135)	¢	(352)

December 31, 2010

<del>-</del>	¥ . 79 . 1						December			-1					
			r-than-temp						ithout other						
<u>-</u>	ımpaırn	ients	recorded in	earı	nngs			ımp	airments re	cord		ıgs		_	
			Caa or								Caa or				
	В		lower		Total		Ba		В		lower		Total		Total
Trust-level		-				_		_		_		_		_	
Actual cumulative collateral losses incurred to date	12.0	%	16.1	%	16.0	%	13.2	%	12.5	%	12.6	%	12.7	%	n/a
Projected additional collateral losses to be incurred	38.2	%	43.2	%	43.0	%	46.5	%	42.7	%	40.8	%	42.1	%	n/a
Class-level															
Average remaining credit enhancement	26.0	%	22.6	%	22.8	%	72.7	%	63.6	%	50.5	%	56.7	%	n/a
Security-specific															
Number of positions	5		81		86		11		10		35		56		142
Par value	\$ 42	\$	952	\$	994	\$	73	\$	69	\$	265	\$	407	\$	1,401
Amortized cost	\$ 33	\$	650	\$	683	\$	73	\$	69	\$	265	\$	407	\$	1,090
Fair value	\$ 21	\$	425	\$	446	\$	62	\$	54	\$	158	\$	274	\$	720
Gross unrealized losses															
Total	\$ (12)	\$	(225)	\$	(237)	\$	(11)	\$	(15)	\$	(107)	\$	(133)	\$	(370)
12-24 months	\$ `	\$	(9)	\$	(9)	\$	`´	\$	`´	\$	`	\$	`	\$	(9)
Over 24 months (2)	\$ (12)	\$	(216)	\$	(228)	\$	(11)	\$	(15)	\$	(107)	\$	(133)	\$	(361)
Cumulative write-downs recognized	\$ `(9)	\$	(293)	\$	(302)	\$	`′	\$	`′	\$	`′	\$	'	\$	(302)
Principal payments received during the period	( )		, ,		` /									\$	` 99´

<sup>(1)</sup> Actual cumulative realized principal losses on the below investment grade Subprime securities we own, as reported by the trust servicers, were \$17 million as of June 30, 2011.

The above tables include information only about below investment grade Subprime securities with gross unrealized losses that are not reliably insured as of each period presented. As such, the par value and composition of securities included can vary significantly from period to period due to changes in variables such as credit ratings, principal payments, sales, purchases and realized principal losses.

88

As of June 30, 2011, our Subprime securities that are reliably insured include 10 below investment grade Subprime securities with a total fair value of \$94 million and aggregate gross unrealized losses of \$39 million, all of which are rated B. These securities are insured by one bond insurer rated B that we estimate has sufficient claims paying capacity to service its obligations on these securities. The securitization trusts from which our securities were issued are currently receiving contractual payments from the bond insurer and considering the combination of expected future payments from the bond insurer and cash flows available from the underlying collateral, we expect the trust to have adequate cash flows to make all contractual payments due to the class of securities we own. As a result, our security-specific estimates of future cash flows indicate that these securities' estimated recovery values equal or exceed their amortized cost. Accordingly, no other-than-temporary impairments have been recognized on these securities. As of December 31, 2010, our Subprime securities that are reliably insured included 10 below investment grade Subprime securities with a total fair value of \$76 million and aggregate gross unrealized losses of \$68 million.

As of June 30, 2011, our below investment grade Subprime securities with gross unrealized losses that are not reliably insured and without other-than-temporary impairments recorded in earnings had incurred actual cumulative collateral losses of 8.1%. Our impairment evaluation forecasts more severe assumptions than the trusts are actually experiencing, including a projected weighted average underlying default rate of 49.5% and a projected weighted average loss severity of 69.4%, which resulted in projected additional collateral losses of 34.1%. As the average remaining credit enhancement for these securities of 44.5% exceeds the projected additional collateral losses of 34.1%, these securities have not been impaired.

As of June 30, 2011, our below investment grade Subprime securities with gross unrealized losses that are not reliably insured and with other-than-temporary impairments recorded in earnings had incurred actual cumulative collateral losses of 17.8%. Our impairment evaluation forecasts more severe assumptions than the trusts are actually experiencing, including a projected weighted average underlying default rate of 56.0% and a projected weighted average loss severity of 76.5%, which resulted in projected additional collateral losses of 42.0%. As the average remaining credit enhancement for these securities of 20.3% is insufficient to withstand the projected additional collateral losses, we have recognized cumulative write-downs in earnings on the securities as reflected in the table above using our calculated recovery value at the time of impairment. The current average recovery value of these securities as a percentage of par was 71.7% and exceeded these securities' current average amortized cost as a percentage of par of 68.8%, which demonstrates our conclusion that the nature of the remaining unrealized loss on these securities is temporary and will reverse over time. The comparison indicates that recovery value exceeds amortized cost based on a comprehensive evaluation of financial, economic and capital markets assumptions developed for this reporting period.

We believe the unrealized losses on our Subprime securities, including those over 24 months, result from the current risk premium on these securities, which should continue to reverse over the securities' remaining lives, as demonstrated by improved valuations in 2010 and 2011. We expect to receive our estimated share of contractual principal and interest collections used to determine the securities' recovery value. As of June 30, 2011, we do not have the intent to sell and it is not more likely than not we will be required to sell these securities before the recovery of their amortized cost basis. We believe that our valuation and impairment processes are comprehensive, employ the most current views about collateral and securitization trust financial positions, and demonstrate our recorded impairments and that the remaining unrealized losses on these positions are temporary.

#### Problem, restructured, or potential problem securities

We also monitor the quality of our fixed income and bank loan portfolios by categorizing certain investments as "problem," "restructured" or "potential problem." Problem fixed income securities and bank loans are in default with respect to principal or interest and/or are investments issued by companies that have gone into bankruptcy subsequent to our acquisition or loan. Fixed income and bank loan investments are categorized as restructured when the debtor is experiencing financial difficulty and we grant a concession. Potential problem fixed income or bank loan investments are current with respect to contractual principal and/or interest, but because of other facts and circumstances, we have concerns regarding the borrower's ability to pay future principal and interest according to the original terms, which causes us to believe these investments may be classified as problem or restructured in the future.

<sup>(2)</sup> As of June 30, 2011, \$170 million of unrealized losses on securities with other-than-temporary impairments recognized in earnings and \$120 million of unrealized losses on securities without other-than-temporary impairments recognized in earnings have been greater than or equal to 20% of those securities' amortized cost for a period of more than 24 consecutive months. As of December 31, 2010, \$188 million of unrealized losses on securities with other-than-temporary impairments recognized in earnings and \$108 million of unrealized losses on securities without other-than-temporary impairments recognized in earnings have been greater than or equal to 20% of those securities' amortized cost for a period of more than 24 consecutive months.

The following table summarizes problem, restructured and potential problem fixed income securities and bank loans, which are reported in other investments.

(\$ in millions) June 30, 2011											
	·					Percent of					
			Amortized			total fixed					
			cost as a		Fair value as a	income and					
	Par	Amortized	percent of	Fair	percent of	bank loan					
	value (1)	cost (1)	par value	value (2)	par value	portfolios					

Restructured \$ 99 83 83.8% 81 81.8% 0.1% Problem 542 200 36.9 191 35.2 0.3 Potential problem 2,579 1,327 51.5 1,039 40.3 1.3 3,220 1,610 1,311 40.7 1.7% 50.0 Cumulative write-downs recognized (3) 868

December 31, 2010 Percent of Amortized total fixed cost as a Fair value as a income and Par Amortized percent of Fair percent of bank loan cost (1) value (2) portfolios value (1) par value par value Restructured 99 79 83 83.8% 79.8% 0.1% 665 Problem 214 32.2 188 28.3 0.2 3,441 1,485 34.0 1.5 Potential problem 43.2 1.171

1,782

1,005

42.4

1,438

1.8%

34.2

(1) The difference between par value and amortized cost of \$1.61 billion and \$2.42 billion as of June 30, 2011 and December 31, 2010, respectively, is primarily attributable to write-

Cumulative write-downs recognized (3)

4,205

As of June 30, 2011, amortized cost for the problem category was \$200 million and comprised \$87 million of Subprime, \$55 million of municipal bonds, \$44 million of Alt-A, \$6 million of corporates (primarily privately placed), \$5 million of CDO and \$3 million of Consumer and other ABS.

As of June 30, 2011, amortized cost for the potential problem category was \$1.33 billion and comprised \$558 million of Subprime, \$297 million of Alt-A, \$246 million of Prime, \$111 million of municipal bonds, \$56 million of CDO, \$34 million of corporates (primarily public), \$18 million of CMBS, \$4 million of bank loans and \$3 million of Consumer and other ABS.

90

**Net investment income** The following table presents net investment income.

(\$ in millions)		Three m Ju	ionth ine 30		Six mo Ju	nths ne 30	
		2011		2010	 2011		2010
Fixed income securities	\$	899	\$	955	\$ 1,799	\$	1,914
Equity securities		34		25	53		46
Mortgage loans		87		99	176		203
Limited partnership interests		18		7	28		13
Short-term investments		1		2	3		4
Other		26	_	6	 37	_	7
Investment income, before expense	_	1,065		1,094	 2,096		2,187
Investment expense		(45)		(45)	(94)		(88)
Net investment income	\$	1,020	\$	1,049	\$ 2,002	\$	2,099

Net investment income decreased 2.8% or \$29 million in the second quarter of 2011 and 4.6% or \$97 million in the first six months of 2011 compared to the same periods of 2010. These declines were primarily due to lower average investment balances due to decreased Allstate Financial contractholder funds, partially offset by higher yields. Net investment income was \$1.01 billion, \$998 million and \$982 million in the third quarter of 2010, fourth quarter of 2010 and first quarter of 2011, respectively.

**Realized capital gains and losses** The following table presents the components of realized capital gains and losses and the related tax effect.

(\$ in millions)	Three m Ju	onths ne 30			Six mo Ju	nths ne 30	
	 2011		2010	_	2011		2010
Impairment write-downs	\$ (70)	\$	(239)	\$	(184)	\$	(462)
Change in intent write-downs	(16)		(67)		(85)		(99)
Net other-than-temporary impairment losses recognized in earnings	 (86)	_	(306)	_	(269)		(561)
Sales	141		145		424		233
Valuation of derivative instruments	(50)		(283)		(28)		(438)
Settlements of derivative instruments	(3)		(27)		(92)		(57)
EMA limited partnership income	55		20		118		24
Realized capital gains and losses, pre-tax	 57		(451)	_	153	_	(799)
Income tax (expense) benefit	 (21)		157	_	(54)	_	279

downs and a zero-coupon security.

(2) Bank loans are reflected at amortized cost.

<sup>(3)</sup> Cumulative write-downs recognized only reflect impairment write-downs related to investments within the problem, potential problem and restructured categories.

Realized capital gains and losses, after-tax

\$ 36 \$ (294) \$ 99 \$ (520)

Impairment write-downs are presented in the following table.

(\$ in millions)	Three m Ju	onths ne 30		Six mo Ju	nths one 30	
	2011		2010	 2011		2010
Fixed income securities	\$ (48)	\$	(172)	\$ (134)	\$	(352)
Equity securities	(13)		(31)	(33)		(37)
Mortgage loans	(7)		(28)	(13)		(41)
Limited partnership interests	(1)		(8)	(2)		(32)
Other investments	(1)			(2)		
Impairment write-downs	\$ (70)	\$	(239)	\$ (184)	\$	(462)

Impairment write-downs for the three months and six months ended June 30, 2011 were primarily driven by RMBS, which experienced deterioration in expected cash flows, and investments with commercial real estate exposure, including CMBS, equity securities, mortgage loans and certain real estate related municipal bonds, which were impacted by lower real estate valuations or experienced deterioration in expected cash flows. Equity securities were also written down due to the length of time and extent to which fair value was below cost, considering our assessment of the financial condition and near-term and long-term prospects of the issuer, including relevant industry conditions and trends. Impairment write-downs on below investment grade RMBS and CMBS were \$30

91

million and \$13 million for the three months ended June 30, 2011, respectively, and \$76 million and \$32 million for the six months ended June 30, 2011, respectively.

*Change in intent write-downs* are presented in the following table.

(\$ in millions)		Three m Ju	onth ne 30			Six mo Ju	nths ne 3	
	_	2011		2010	_	2011		2010
Fixed income securities	\$	(16)	\$	(67)	\$	(85)	\$	(93)
Mortgage loans								(6)
Change in intent write-downs	\$	(16)	\$	(67)	\$	(85)	\$	(99)

The change in intent write-downs in the three months and six months ended June 30, 2011 were primarily a result of ongoing comprehensive reviews of our portfolios resulting in write-downs of individually identified investments, primarily lower yielding, floating rate RMBS and municipal bonds.

Sales generated \$141 million and \$424 million of net realized gains for the three months and six months ended June 30, 2011, respectively. Net realized gains for the three months ended June 30, 2011 primarily related to \$145 million of net gains on sales of corporate, U.S. government, foreign government and ABS securities and \$28 million of net gains on sales of equity securities, partially offset by \$40 million of net losses on sales of municipal fixed income securities. Net realized gains for the six months ended June 30, 2011 primarily related to \$266 million of net gains on sales of corporate, U.S. government, foreign government and ABS securities and \$170 million of net gains on sales of equity securities, partially offset by \$35 million of net losses on sales of municipal fixed income securities.

Valuation and settlements of derivative instruments net realized capital losses totaling \$53 million for the three months ended June 30, 2011 included \$50 million of losses on the valuation of derivative instruments and \$3 million of losses on the settlement of derivative instruments. Valuation and settlements of derivative instruments net realized capital losses totaling \$120 million for the six months ended June 30, 2011 included \$28 million of losses on the valuation of derivative instruments and \$92 million of losses on the settlement of derivative instruments. The net realized capital losses on derivative instruments for the three months ended June 30, 2011 primarily included losses on interest rate risk management due to decreases in interest rates and volatility and losses on equity exposure risk management due to an increase in equity indices and a decrease in volatility. As a component of our approach to managing interest rate risk, realized gains and losses on certain derivative instruments are most appropriately considered in conjunction with the unrealized gains and losses on the fixed income portfolio. This approach mitigates the impacts of general interest rate changes to our overall financial condition.

#### CAPITAL RESOURCES AND LIQUIDITY HIGHLIGHTS

- · Shareholders' equity as of June 30, 2011 was \$18.76 billion, a decrease of 1.3% from \$19.02 billion as of December 31, 2010.
- · On January 3, 2011, April 1, 2011 and July 1, 2011, we paid a quarterly shareholder dividend of \$0.20, \$0.21 and \$0.21, respectively. On July 12, 2011, we declared a quarterly shareholder dividend of \$0.21 to be payable on October 3, 2011.
- During the first six months of 2011, we repurchased 16.9 million common shares for \$532 million. As of June 30, 2011, our current \$1.00 billion share repurchase program had \$308 million remaining and is expected to be completed by March 31, 2012.

92

#### **CAPITAL RESOURCES AND LIQUIDITY**

**Capital resources** consist of shareholders' equity and debt, representing funds deployed or available to be deployed to support business operations or for general corporate purposes. The following table summarizes our capital resources.

(\$ in millions)

June 30, 2011 December 31, 2010

Common stock, retained income and other shareholders' equity items	\$	18,391	\$ 19,200
Accumulated other comprehensive income (loss)		373	(184)
Total shareholders' equity		18,764	 19,016
Debt		5,907	5,908
Total capital resources	\$	24,671	\$ 24,924
	-		 
Ratio of debt to shareholders' equity		31.5%	31.1%
Ratio of debt to capital resources		23.9%	23.7%

Shareholders' equity decreased in the first six months of 2011, primarily due to share repurchases, dividends paid to shareholders and a net loss, partially offset by increased unrealized net capital gains on investments.

*Debt* Except for \$42 million in long-term debt related to the synthetic leases scheduled to mature in December 2011, we do not have any required principal payments until 2012 when \$350 million of 6.125% Senior Notes are due.

*Share repurchases* During the first six months of 2011, we repurchased 16.9 million common shares for \$532 million. As of June 30, 2011, \$308 million remained on our current \$1.00 billion share repurchase program that we commenced in November 2010, and is expected to be completed by March 31, 2012.

**Financial ratings and strength** Our ratings are influenced by many factors including our operating and financial performance, asset quality, liquidity, asset/liability management, overall portfolio mix, financial leverage (i.e., debt), exposure to risks such as catastrophes and the current level of operating leverage. There have been no changes to our debt, commercial paper and insurance financial strength ratings from Moody's, S&P and A.M. Best since December 31, 2010.

Allstate Life Insurance Company ("ALIC"), AIC and The Allstate Corporation are party to the Amended and Restated Intercompany Liquidity Agreement ("Liquidity Agreement") which allows for short-term advances of funds to be made between parties for liquidity and other general corporate purposes. The Liquidity Agreement does not establish a commitment to advance funds on the part of any party. ALIC and AIC each serve as a lender and borrower and the Corporation serves only as a lender. AIC also has a capital support agreement with ALIC. Under the capital support agreement, AIC is committed to provide capital to ALIC to maintain an adequate capital level. The maximum amount of potential funding under each of these agreements is \$1.00 billion.

In addition to the Liquidity Agreement, the Corporation also has an intercompany loan agreement with certain of its subsidiaries, which include, but are not limited to, AIC and ALIC. The amount of intercompany loans available to the Corporation's subsidiaries is at the discretion of the Corporation. The maximum amount of loans the Corporation will have outstanding to all its eligible subsidiaries at any given point in time is limited to \$1.00 billion. The Corporation may use commercial paper borrowings, bank lines of credit and repurchase agreements to fund intercompany borrowings.

**Liquidity sources and uses** We actively manage our financial position and liquidity levels in light of changing market, economic, and business conditions. Liquidity is managed at both the entity and enterprise level across the Company, and is assessed on both base and stressed level liquidity needs. We believe we have sufficient liquidity to meet these needs. Additionally, we have existing intercompany agreements in place that facilitate liquidity management across the Company to enhance flexibility.

*Parent company capital capacity* At the parent holding company level, we have deployable invested assets totaling \$3.49 billion as of June 30, 2011. These assets include investments that are generally saleable within one quarter totaling \$3.02 billion. This provides funds for the parent company's relatively low fixed charges.

In the first six months of 2011, dividends totaling \$438 million were paid by AIC to its parent, the Corporation.

93

The Corporation has access to additional borrowing to support liquidity as follows:

- A commercial paper facility with a borrowing limit of \$1.00 billion to cover short-term cash needs. As of June 30, 2011, there were no balances outstanding and therefore the remaining borrowing capacity was \$1.00 billion; however, the outstanding balance can fluctuate daily.
- Our primary credit facility is available for short-term liquidity requirements and backs our commercial paper facility. Our \$1.00 billion unsecured revolving credit facility has an initial term of five years expiring in 2012 with two optional one-year extensions that can be exercised at the end of any of the remaining anniversary years of the facility upon approval of existing or replacement lenders providing more than two-thirds of the commitments to lend. The program is fully subscribed among 11 lenders with the largest commitment being \$185 million. The commitments of the lenders are several and no lender is responsible for any other lender's commitment if such lender fails to make a loan under the facility. This facility contains an increase provision that would allow up to an additional \$500 million of borrowing provided the increased portion could be fully syndicated at a later date among existing or new lenders. This facility has a financial covenant requiring that we not exceed a 37.5% debt to capital resources ratio as defined in the agreement. This ratio as of June 30, 2011 was 20.0%. Although the right to borrow under the facility is not subject to a minimum rating requirement, the costs of maintaining the facility and borrowing under it are based on the ratings of our senior, unsecured, nonguaranteed long-term debt. There were no borrowings under the credit facility during the second quarter and first six months of 2011. The total amount outstanding at any point in time under the combination of the commercial paper program and the credit facility cannot exceed the amount that can be borrowed under the credit facility.
- A universal shelf registration statement was filed with the Securities and Exchange Commission on May 8, 2009. We can use this shelf registration to issue an unspecified amount of debt securities, common stock (including 383 million shares of treasury stock as of June 30, 2011), preferred stock, depositary shares, warrants, stock purchase contracts, stock purchase units and securities of trust subsidiaries. The specific terms of any securities we issue under this registration statement will be provided in the applicable prospectus supplements.

*Liquidity exposure* Contractholder funds as of June 30, 2011 were \$45.08 billion. The following table summarizes contractholder funds by their contractual withdrawal provisions as of June 30, 2011.

(\$ in millions)		Percent
		to total
Not subject to discretionary withdrawal	\$ 6,166	13.7 %

Subject to discretionary withdrawal with adjustments:		
Specified surrender charges (1)	17,903	39.7
Market value adjustments (2)	7,045	15.6
Subject to discretionary withdrawal without adjustments (3)	13,964	31.0
Total contractholder funds (4)	\$ 45,078	100.0 %

<sup>(1)</sup> Includes \$9.16 billion of liabilities with a contractual surrender charge of less than 5% of the account balance.

While we are able to quantify remaining scheduled maturities for our institutional products, anticipating retail product surrenders is less precise. Retail life and annuity products may be surrendered by customers for a variety of reasons. Reasons unique to individual customers include a current or unexpected need for cash or a change in life insurance coverage needs. Other key factors that may impact the likelihood of customer surrender include the level of the contract surrender charge, the length of time the contract has been in force, distribution channel, market interest rates, equity market conditions and potential tax implications. In addition, the propensity for retail life insurance policies to lapse is lower than it is for fixed annuities because of the need for the insured to be re-underwritten upon policy replacement. Surrenders and partial withdrawals for our retail annuities increased 42.7% and 23.7% in the second quarter and first six months of 2011, respectively, compared to the same periods of 2010. The annualized surrender and partial withdrawal rate on deferred annuities, interest-sensitive life insurance and

94

Allstate Bank products, based on the beginning of year contractholder funds, was 15.0% and 12.2% for the first six months of 2011 and 2010, respectively. Allstate Financial strives to promptly pay customers who request cash surrenders; however, statutory regulations generally provide up to six months in most states to fulfill surrender requests.

Our institutional products are primarily funding agreements sold to unaffiliated trusts used to back medium-term notes. As of June 30, 2011, total institutional products outstanding were \$1.90 billion. The following table presents the remaining scheduled maturities for our institutional products outstanding as of June 30, 2011.

(\$ in millions)	
2011	\$ 25
2012	40
2013	1,750
2016	85
	\$ 1,900

Our asset-liability management practices limit the differences between the cash flows generated by our investment portfolio and the expected cash flow requirements of our life insurance, annuity and institutional product obligations.

The following table summarizes consolidated cash flow activities by business segment for the six months ended June 30.

(\$ in millions)		Property	. Tiabi	liter (1)		Allstate	Finan	cial (1)			rporat Other			Co	nsolid	lated
	_		-LIAUI		_		Filldii		_	2011	Other		-		isona	
NT ( 1 11 11 ( 11 )	_	2011		2010	_	2011	_	2010		2011		2010		2011	_	2010
Net cash provided by (used in):												(00)				
Operating activities	\$	641	\$	736	\$	731	\$	1,427	\$	(112)	\$	(69)	\$	1,260	\$	2,094
Investing activities		(220)		(493)		2,847		2,193		388		53		3,015		1,753
Financing activities		(3)		(4)		(3,396)		(3,549)		(745)		(195)		(4,144)		(3,748)
Net increase in consolidated cash		` ,		. ,		, ,		. ,		, ,		, ,	\$	131	\$	99

<sup>(1)</sup> Business unit cash flows reflect the elimination of intersegment dividends, contributions and borrowings.

*Property-Liability* Lower cash provided by operating activities in the first six months of 2011 compared to the first six months of 2010 was primarily due to higher claim payments, partially offset by lower income tax payments.

Lower cash used in investing activities in the first six months of 2011 compared to the first six months of 2010 was primarily due to higher net sales of fixed income and equity securities, partially offset by higher net purchases of fixed income and equity securities.

Allstate Financial Lower cash provided by operating cash flows in the first six months of 2011 compared to the first six months of 2010 was primarily due to income tax refunds in the first six months of 2010.

Higher cash provided by investing activities in the first six months of 2011 compared to the first six months of 2010 were impacted by higher net sales of fixed income securities used to fund reductions in contractholder fund liabilities.

Lower cash used in financing activities in the first six months of 2011 compared to the first six months of 2010 was primarily due to decreased maturities and retirements of institutional products, partially offset by higher surrenders and partial withdrawals on fixed annuities and lower deposits on fixed annuities and Allstate Bank products.

*Corporate and Other* Fluctuations in the Corporate and Other operating cash flows were primarily due to the timing of intercompany settlements. Investing activities primarily relate to investments in the portfolios of Kennett Capital Inc. Financing cash flows of the Corporate and Other segment reflect

<sup>(2) \$5.84</sup> billion of the contracts with market value adjusted surrenders have a 30-45 day period at the end of their initial and subsequent interest rate guarantee periods (which are typically 5 or 6 years) during which there is no surrender charge or market value adjustment.

<sup>(3) 68%</sup> of these contracts have a minimum interest crediting rate guarantee of 3% or higher.

<sup>(4)</sup> Includes \$1.09 billion of contractholder funds on variable annuities reinsured to The Prudential Insurance Company of America, a subsidiary of Prudential Financial Inc., in 2006.

#### RECENT DEVELOPMENTS

Management is focused on the following to improve our return on equity (excluding primarily the effects of realized and unrealized capital gains and losses) to approach our historic level by 2014 and maintain this level on average over time:

- · Maintain historically strong combined ratio in Allstate Brand standard auto in the mid-90's.
- · Target Allstate Brand homeowners combined ratio excluding catastrophes in the low-60's. Adequate returns require homeowners combined ratios including catastrophes in the mid-80's over time.
- Move to low single-digit growth in Allstate Protection auto policies in 2013; anticipate continued decline in auto policies in 2011.
- · Improve Allstate Financial operating return on equity to reach 9-10% based on Allstate Financial segment operating income measure, excluding approximately \$1.0 billion of capital in excess of management requirements expected to be produced during the same period.
- · Maintain investment portfolio yield and optimize interest rate risk in 2011.

96

#### **Item 4. Controls and Procedures**

Evaluation of Disclosure Controls and Procedures. We maintain disclosure controls and procedures as defined in Rules 13a-15(e) and 15d-15(e) under the Securities Exchange Act of 1934. Under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, we conducted an evaluation of the effectiveness of our disclosure controls and procedures as of the end of the period covered by this report. Based upon this evaluation, the principal executive officer and the principal financial officer concluded that our disclosure controls and procedures are effective in providing reasonable assurance that material information required to be disclosed in our reports filed with or submitted to the Securities and Exchange Commission under the Securities Exchange Act is made known to management, including the principal executive officer and the principal financial officer, as appropriate to allow timely decisions regarding required disclosure.

Changes in Internal Control over Financial Reporting. During the fiscal quarter ended June 30, 2011, there have been no changes in our internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

97

#### PART II. OTHER INFORMATION

#### **Item 1. Legal Proceedings**

Information required for Part II, Item 1 is incorporated by reference to the discussion under the heading "Regulation and Compliance" and under the heading "Legal and regulatory proceedings and inquiries" in Note 10 of the condensed consolidated financial statements in Part I, Item 1 of this Form 10-Q.

#### Item 1A. Risk Factors

This document contains "forward-looking statements" that anticipate results based on our estimates, assumptions and plans that are subject to uncertainty. These statements are made subject to the safe-harbor provisions of the Private Securities Litigation Reform Act of 1995. We assume no obligation to update any forward-looking statements as a result of new information or future events or developments.

These forward-looking statements do not relate strictly to historical or current facts and may be identified by their use of words like "plans," "seeks," "expects," "will," "should," "anticipates," "estimates," "intends," "believes," "likely," "targets" and other words with similar meanings. These statements may address, among other things, our strategy for growth, catastrophe exposure management, product development, investment results, regulatory approvals, market position, expenses, financial results, litigation and reserves. We believe that these statements are based on reasonable estimates, assumptions and plans. However, if the estimates, assumptions or plans underlying the forward-looking statements prove inaccurate or if other risks or uncertainties arise, actual results could differ materially from those communicated in these forward-looking statements. Risk factors which could cause actual results to differ materially from those suggested by such forward-looking statements include but are not limited to those discussed or identified in this document, in our public filings with the Securities and Exchange Commission, and those incorporated by reference in Part I, Item 1A of The Allstate Corporation Annual Report on Form 10-K for 2010.

#### Item 2. Unregistered Sales of Equity Securities and Use of Proceeds

Issuer Purchases of Equity Securities

<u>Period</u> Total number of Average price Total number Maximum number

	shares (or units) purchased <sup>(1)</sup>	1	paid per share (or unit)	of shares (or units) purchased as part of publicly announced plans or programs <sup>(2)</sup>	(or approximate dollar value) of shares (or units) that may yet be purchased under the plans or programs <sup>(3)</sup>
April 1, 2011 - April 30, 2011	2,311,900	\$	31.7557	2,311,900	\$ 466 million
May 1, 2011 - May 31, 2011	2,620,462	\$	32.4648	2,620,462	\$ 381 million
June 1, 2011 - June 30, 2011	2,423,096	\$	30.1562	2,423,078	\$ 308 million
Total	7,355,458	\$	31.4814	7,355,440	

<sup>(1)</sup> In accordance with the terms of its equity compensation plans, Allstate acquired the following shares in connection with stock option exercises by employees and/or directors. The stock was received in payment of the exercise price of the options and in satisfaction of withholding taxes due upon exercise or vesting.

April: none May: none June: 18

98

#### Item 6. Exhibits

(a) Exhibits

An Exhibit Index has been filed as part of this report on page E-1.

99

#### **SIGNATURE**

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

The Allstate Corporation (Registrant)

July 29, 2011

10.3

15

By /s/ Samuel H. Pilch
Samuel H. Pilch
(chief accounting officer and du

(chief accounting officer and duly authorized officer of Registrant)

100

Exhibit No.	<u>Description</u>
3(i)	Amended and Restated Certificate of Incorporation filed with the Secretary of Delaware on May 18, 2011, incorporated herein by reference to Exhibit 3(i) to The Allstate Corporation current report on Form 8-K filed May 18, 2011.
3(ii)	Amended and Restated Bylaws of The Allstate Corporation effective May 18, 2011, incorporated herein by reference to Exhibit 3(ii) to The Allstate Corporation current report on Form 8-K filed May 18, 2011.
4	Registrant hereby agrees to furnish the Commission, upon request, with the instruments defining the rights of holders of each issue of long-term debt of the Registrant and its consolidated subsidiaries.
10.1	Stock Purchase Agreement, dated as of May 17, 2011, between White Mountains Holdings (Luxembourg) S.à r.l. and The Allstate Corporation, incorporated herein by reference to Exhibit 10.1 to The Allstate Corporation current report on Form 8-K filed May 23, 2011. (Certain schedules and exhibits to the Stock Purchase Agreement are omitted pursuant to Item 601(b)(2) of Regulation S-K. The Registrant agrees to furnish to the Securities and Exchange Commission, upon request, a copy of any omitted schedule or exhibit.)
10.2	Guaranty Agreement, dated as of May 17, 2011, by White Mountains Insurance Group, Ltd. in favor of The Allstate Corporation, incorporated herein by reference to Exhibit 10.2 to The Allstate Corporation current report on Form 8-K filed May 23, 2011.

Form of Tier One Change of Control Employment Agreement for Agreements entered into on or after February 22, 2011.

Acknowledgment of awareness from Deloitte & Touche LLP, dated July 29, 2011, concerning unaudited interim financial information.

<sup>(2)</sup> Repurchases under our programs are, from time to time, executed under the terms of a pre-set trading plan meeting the requirements of Rule 10b5-1(c) of the Securities Exchange Act of 1934.

<sup>(3)</sup> On November 9, 2010, we announced the approval of a new share repurchase program for \$1.00 billion. This program is expected to be completed by March 31, 2012.

31 (i)	Rule 13a-14(a) Certification of Principal Executive Officer
31 (i)	Rule 13a-14(a) Certification of Principal Financial Officer
32	Section 1350 Certifications
101.INS	XBRL Instance Document
101.SCH	XBRL Taxonomy Extension Schema
101.CAL	XBRL Taxonomy Extension Calculation Linkbase
101.DEF	XBRL Taxonomy Extension Definition Linkbase
101.LAB	XBRL Taxonomy Extension Label Linkbase
101.PRE	XBRL Taxonomy Extension Presentation Linkbase

# CHANGE OF CONTROL EMPLOYMENT AGREEMENT

# **AMONG**

# THE ALLSTATE CORPORATION, ALLSTATE INSURANCE COMPANY

# AND

# [INSERT NAME OF EXECUTIVE]

(Tier One)

For Agreements entered into on or after February 22, 2011

## **TABLE OF CONTENTS**

		<u>Page</u>
ARTICLE I.	CERTAIN DEFINITIONS	1
ARTICLE II.	POST-CHANGE PERIOD	8
2.1	Position and Duties	8
2.2	Compensation	8
2.3	Stock Incentive Awards	10
2.4	Unfunded Deferred Compensation	11
ARTICLE III.	TERMINATION OF EMPLOYMENT	11
3.1	Disability	11
3.2	Death	12
3.3	Cause	12
3.4	Good Reason	13
ARTICLE IV.	COMPANY'S OBLIGATIONS UPON A TERMINATION OF EMPLOYMENT	15
4.1	If by Executive for Good Reason or by the Company Other Than for Cause or Disability	15
4.2	If by the Company for Cause	17
4.3	If by Executive Other Than for Good Reason	17
4.4	If by the Company for Disability	18
4.5	If Upon Death	18
4.6	Amount Contested	18
ARTICLE V.	REDUCTIONS IN PAYMENTS BY THE COMPANY	20
5.1	Reduction of Potential Parachute Payments	20
5.2	Determination by Auditors	20
ARTICLE VI.	EXPENSES AND INTEREST	21
6.1	Legal and Other Expenses	21
6.2	Interest	21
ARTICLE VII.	NO SET-OFF OR MITIGATION	21
7.1	No Set-off by Company	21
7.2	No Mitigation	21
ARTICLE VIII.	RESTRICTIVE COVENANTS	22
8.1	Non-Competition	22
8.2	Non-Solicitation	22

	-i-	
ARTICLE IX.	NON-EXCLUSIVITY OF RIGHTS	24
9.1	Waiver of Certain Other Rights	24
9.2	Other Rights	24
ARTICLE X.	MISCELLANEOUS	24
10.1	No Assignability	24
10.2	Successors	24
10.3	Payments to Beneficiary	25
10.4	Non-Alienation of Benefits	25
10.5	No Deference	25
10.6	Severability	25
10.7	Amendments	25
10.8	Notices	25
10.9	Counterparts	26
10.10	Governing Law	26
10.11	Captions	26
10.12	Number and Gender	26
10.13	Tax Withholding	26
10.14	No Waiver	26
10.15	Joint and Several Liability	26
10.16	No Rights Prior to Effective Date	26
10.17	Six-month Delay	26
10.18	Interpretation to Avoid 409A Penalties	27
10.19	Entire Agreement	27
	-ii-	

23

23

23

Reasonableness of Restrictive Covenants

Non-Disparagement

Right to Injunction; Survival of Undertakings

8.3

8.4 8.5

## THE ALLSTATE CORPORATION

## CHANGE OF CONTROL EMPLOYMENT AGREEMENT

THIS AGREEMENT dated as of \_\_\_\_\_\_, 201\_ (the "<u>Agreement Date</u>") is made by and among The Allstate Corporation, a Delaware corporation ("<u>Allstate</u>"), the Allstate Insurance Company, an Illinois insurance corporation ("<u>AIC</u>"), and \_\_\_\_\_ ("<u>Executive</u>").

#### **RECITALS**

Allstate has determined that it is in the best interests of Allstate and its stockholders to assure that the Company will have the continued service of Executive. Allstate also believes it is imperative to reduce the distraction of Executive that would result from the personal uncertainties caused by a pending or threatened change of control of Allstate, to encourage Executive's full attention and dedication to the Company, and to provide Executive with compensation and benefits arrangements upon a change of control that will satisfy the expectations of Executive and be competitive with those of similarly situated corporations. This Agreement is intended to accomplish these objectives.

# ARTICLE I. CERTAIN DEFINITIONS

As used in this Agreement, the terms specified below shall have the following meanings:

- 1.1 "<u>Accrued Annual Bonus</u>" means the amount of any Annual Bonus earned and due to be paid but not yet paid to Executive as of the Executive's Termination Date, other than amounts that Executive has elected to defer.
- 1.2 "<u>Accrued Base Salary</u>" means the amount of Executive's Base Salary that is accrued but unpaid as of the Executive's Termination Date, other than amounts that Executive has elected to defer.
- 1.3 "<u>Accrued LTIP Bonus</u>" means the amount of any LTIP Bonus earned and due to be paid but not yet paid to Executive as of the Executive's Termination Date, other than amounts that Executive has elected to defer.

- 1.4 "Accrued Obligations" means, as of any date, the sum of Executive's Accrued Base Salary, Accrued Annual Bonus, Accrued LTIP Bonus, any accrued but unpaid vacation pay, and any other amounts and benefits that are then due to be paid or provided to Executive by the Company (other than pursuant to Sections 2.4 or 4.1(b) or any defined benefit or defined contribution plan of the Company, whether or not qualified under Section 401(a) of the Code), but have not yet been paid or provided (as applicable).
  - 1.5 "<u>Agreement Date</u>" -- see the introductory paragraph of this Agreement.
- 1.6 "Agreement Term" means the period commencing on the Agreement Date and ending on the third anniversary of the Agreement Date or, if later, such later date to which the Agreement Term is extended pursuant to the following sentence. Commencing on the second anniversary of the Agreement Date, the Agreement Term shall automatically be extended each day by one day to create a new one-year term until, at any time after the second anniversary of the Agreement Date, the Company delivers written notice (an "Expiration Notice") to Executive that the Agreement shall expire on a date specified in the Expiration Notice (the "Expiration Date") that is not less than 12 months after the date the Expiration Notice is delivered to Executive; provided, however, that if an Effective Date or an Imminent Control Change Date occurs before the Expiration Date specified in the Expiration Notice, then such Expiration Notice shall be void and of no further effect. "Imminent Control Change Date" means (i) any date on which a proposal or offer for a Change of Control is presented to Allstate's stockholders generally or to any of Allstate's directors or executive officers or is publicly announced (whether by advertisement, press release, press interview, public statement, SEC filing or otherwise) or (ii) any subsequent date as of which such proposal or offer for a Change of Control remains effective and has not expired or been revoked.
  - 1.7 "AIC" -- see the introductory paragraph of this Agreement.
  - 1.8 "Allstate" -- see the introductory paragraph of this Agreement.
  - 1.9 "Annual Bonus" -- see Section 2.2(b).
  - 1.10 "Annual Performance Period" -- see Section 2.2(b).
  - 1.11 "Article" means an article of this Agreement.
  - 1.12 "Base Salary" -- see Section 2.2(a).
  - 1.13 "Beneficiary" -- see Section 10.3.
- 1.14 "Board" means the Board of Directors of Allstate or, from and after the Effective Date of a Change of Control that gives rise to a Surviving Corporation, the Board of Directors of such Surviving Corporation.
  - 1.15 "Bonus Plan" -- see Section 2.2(b).
  - 1.16 "Cause" -- see Section 3.3(b).
  - 1.17 "CEO" means Chief Executive Officer.
- 1.18 "<u>Change of Control</u>" means, except as otherwise provided at the end of this Section, the occurrence of any one or more of the following:
  - (a) (*Voting Power*) any Person or group (as such term is defined in Treasury Regulation Section 1.409A-3(i)(5) (v)(B)), other than a Subsidiary or any employee benefit plan (or any related trust) of Allstate or any of its Subsidiaries, acquires or has acquired during the 12-month period ending on the date of the most recent acquisition by such Person of

-2-

Persons, ownership of stock of Allstate possessing 30% or more of the combined voting power of all Voting Securities of Allstate (such a Person or group that is not a Similarly Owned Company (as defined below), a "More than 30% Owner"), except that no Change of Control shall be deemed to have occurred solely by reason of such ownership by a corporation with respect to which both more than 70% of the common stock of such corporation and Voting Securities representing more than 70% of the combined voting power of the Voting Securities of such corporation are then owned, directly or indirectly, by the Persons who were the direct or indirect owners of the common stock and Voting Securities of Allstate immediately before such acquisition in substantially the same proportions as their ownership, immediately before such acquisition, of the common stock and Voting Securities of Allstate, as the case may be (a "Similarly Owned Company"); or

- (b) (Majority Ownership) any Person or group (as such term is defined in Treasury Regulation Section 1.409A-3(i) (5)(v)(B)), other than a Subsidiary or any employee benefit plan (or any related trust) of Allstate or any of its Subsidiaries, acquires ownership of more than 50% of the voting power of all Voting Securities of Allstate or of the total fair market value of the stock of Allstate (such a Person or group that is not a Similarly Owned Company, a "Majority Owner"), except that no Change of Control shall be deemed to have occurred solely by reason of such ownership by a Similarly Owned Company; or
- (c) (Board Composition) a majority of the members of the Board is replaced during any 12-month period by directors whose appointment or election is not endorsed by a majority of the members of the Board before the date of the appointment or election ("Board Turnover"); or
- (d) (*Reorganization*) the consummation of a merger, reorganization, consolidation, or similar transaction, or of a plan or agreement for the sale or other disposition of all or substantially all of the consolidated assets of Allstate, or a plan of liquidation of Allstate (any of the foregoing, a "<u>Reorganization Transaction</u>") that, does not qualify as an Exempt Reorganization Transaction.

Notwithstanding anything contained herein to the contrary: (i) no transaction or event shall constitute a Change of Control for purposes of this Agreement unless the transaction or event constituting the Change of Control also constitutes a change in the ownership of a corporation (as defined in Treasury Regulation Section 1.409A-3(i)(5)(v)), a change in effective control of a corporation (as defined in Treasury Regulation Section 1.409A-3(i)(5)(vi)) or a change in the ownership of a substantial portion of the assets of a corporation (as defined in Treasury Regulation Section 1.409A-3(i)(5)(vii)); and (ii) no sale or disposition of one or more Subsidiaries ("Sale Subsidiary") or the assets thereof shall constitute a Change of Control for purposes of this Agreement if the investments in and advances by Allstate and its Subsidiaries (other than the Sale Subsidiaries) to such Sale Subsidiary as of immediately prior to the sale or disposition determined in accordance with Generally Accepted Accounting Principles ("GAAP") (but after intercompany eliminations and net of the effect of intercompany reinsurance) are less than 51% of the Consolidated Total Shareholders' Equity of Allstate as of immediately prior to the sale or disposition. Consolidated Total Shareholders' Equity means, at any date, the total shareholders' equity of Allstate and its Subsidiaries at such date, as reported in the consolidated financial statements prepared in accordance with GAAP.

-3-

- 1.19 "Code" means the Internal Revenue Code of 1986, as amended. Any reference to any section of the Code shall also refer to any successor provision.
  - 1.20 "Company" means Allstate, AIC and each of Allstate's other Subsidiaries.
  - 1.21 "Company Certificate" -- see Section 5.2.
- 1.22 "<u>Competitive Business</u>" means as of any date (including during the one-year period commencing on the Termination Date) any corporation or other Person (and any branch, office or operation thereof) that engages in, or proposes to engage in:
  - (a) the underwriting, reinsurance, marketing or sale of (i) any form of insurance of any kind that the Company as of such date does, or proposes to, underwrite, reinsure, market or sell (any such form of insurance, an "Allstate Insurance Product") or (ii) any other form of insurance that is marketed or sold in competition with any Allstate Insurance Product, or
    - (b) any other business that as of such date is a direct and material competitor of the Company;

and that is located (i) anywhere in the United States, or (ii) anywhere outside of the United States where the Company is then engaged in, or proposes to engage in, any of such activities.

- 1.23 "Consummation Date" means the date on which a Reorganization Transaction is consummated.
- 1.24 "Disability" -- see Section 3.1(b).
- 1.25 "<u>Disability Effective Date</u>" -- see Section 3.1.
- 1.26 "Effective Date" means the date on which a Change of Control first occurs during the Agreement Term.
- 1.27 "Exchange Act" means the Securities Exchange Act of 1934.
- 1.28 "Excise Taxes" -- see Section 5.1.
- 1.29 "Exempt Reorganization Transaction" means a Reorganization Transaction that fails to result in (a) any Person or group (as such term is defined in Treasury Regulation Section 1.409A-3(i)(5)(v)(B)) becoming a More than 30% Owner or a Majority Owner, (b) Board Turnover, or (c) a sale or disposition to any Person or group (as such term is defined in Treasury Regulation Section 1.409A-3(i)(5)(v)(B)) of the assets of Allstate that have a total Gross Fair Market Value (as defined below)

equal to at least forty percent (40%) of the total Gross Fair Market Value of all of the assets of Allstate immediately before such transaction. "Gross Fair Market Value" means the value of the assets of Allstate, or the value of the assets being disposed of, determined without regard to any liabilities associated with such assets.

1.30 "Good Reason" -- see Section 3.4(b).

-4-

- 1.31 "<u>including</u>" means including without limitation.
- 1.32 "IRS" means the Internal Revenue Service.
- 1.33 "Legal and Other Expenses" -- see Section 6.1(a).
- 1.34 "LTIP" means the Allstate Long-Term Executive Incentive Compensation Plan (or any successor plan).
- 1.35 "LTIP Award" means an incentive compensation opportunity granted under the LTIP.
- 1.36 "LTIP Bonus" means the amount paid or earned in respect of an LTIP Award.
- 1.37 "<u>LTIP Performance Period</u>" means any performance period designated in accordance with any LTIP approved by the Board or any committee of the Board.
- 1.38 "<u>LTIP Target Award</u>" means, in respect of any LTIP Award, the amount that Executive would have been entitled to receive for the LTIP Performance Period corresponding to such LTIP Award if the performance goals established pursuant to such LTIP Award were achieved at the 100% level as of the end of the LTIP Performance Period.
- 1.39 "Lump Sum Value" of an annuity payable pursuant to a defined benefit plan means, as of a specified date, the present value of such annuity, as determined, as of such date, under generally accepted actuarial principles using (i) the applicable interest rate, mortality tables and other methods and assumptions under Code Section 417(e) as published by the IRS and used for determining the value of an immediate annuity on the Termination Date or (ii) if such interest rate and mortality assumptions are no longer published by the IRS, the interest rate and mortality assumptions determined in a manner as similar as practicable to the manner by which the Code Section 417(e) interest rate and mortality assumptions were determined immediately prior to the IRS's cessation of publication of such assumptions; provided, however, that if such defined benefit plan provides for a lump sum distribution and such lump-sum distribution either (x) is the only payment method available under such plan or (y) provides for a greater amount than the Lump Sum Value of the Maximum Annuity available under such plan, then "Lump Sum Value" shall mean such lump sum amount.
- 1.40 "<u>Maximum Annuity</u>" means, in respect of a defined benefit plan (whether or not qualified under Section 401(a) of the Code), an annuity computed in whatever manner permitted under such plan (including frequency of annuity payments, attained age (whether determined as of a current date or as of a future date upon the commencement of annuity payments), and nature of surviving spouse benefits, if any) that yields the greatest Lump Sum Value.
  - 1.41 "More than 30% Owner" -- see paragraph (a) of the definition of "Change of Control."
  - 1.42 "Notice of Consideration" -- see Section 3.3(c).
  - 1.43 "Non-Qualified Plan" -- see Section 2.4.

-5-

- 1.44 "Notice of Termination" means a written notice given in accordance with Section 10.8 that sets forth (i) the specific termination provision in this Agreement relied on by the party giving such notice, (ii) in reasonable detail the specific facts and circumstances claimed to provide a basis for such Termination of Employment, and (iii) if the Termination Date is other than the date of receipt of such Notice of Termination, the Termination Date.
- 1.45 "Person" means any individual, sole proprietorship, partnership, joint venture, limited liability company, trust, unincorporated organization, association, corporation, institution, public benefit corporation, entity or government instrumentality, division, agency, body or department.
  - 1.46 "Plans" means plans, programs, or Policies of the Company.
  - 1.47 "<u>Policies</u>" means policies, practices or procedures of the Company.

- 1.48 "<u>Post-Change Period</u>" means the period commencing on the Effective Date and ending on the second anniversary of the Effective Date.
  - 1.49 "Potential Parachute Payments" -- see Section 5.1.
- 1.50 "<u>Pro-rata Annual Bonus</u>" means, in respect of the Company's fiscal year during which the Termination Date occurs, an amount equal to the product of Executive's Target Annual Bonus (determined as of the Termination Date) multiplied by a fraction, the numerator of which equals the number of days from and including the first day of such fiscal year through and including the Termination Date, and the denominator of which equals 365.
- 1.51 "Pro-rata LTIP Bonus" means an amount equal to the sum of each of the following amounts: for each LTIP Performance Period that is in effect as of a Termination Date, Executive's LTIP Target Award for such LTIP Performance Period multiplied by a fraction, the numerator of which equals the number of days from and including the beginning of such LTIP Performance Period through and including the Termination Date, and the denominator of which equals the aggregate number of days in such LTIP Performance Period.
  - 1.52 "Reorganization Transaction" -- see clause (d) of the definition of "Change of Control."
  - 1.53 "Restricted Shares" means shares of restricted stock, restricted stock units or similar awards.
  - 1.54 "<u>SEC</u>" means the Securities and Exchange Commission.
  - 1.55 "Section" means, unless the context otherwise requires, a section of this Agreement.
  - 1.56 "<u>SERP</u>" means a supplemental executive retirement Plan that is a Non-Qualified Plan.

-6

- 1.57 "Stock Options" means stock options, stock appreciation rights (including limited stock appreciation rights), or similar awards.
- 1.58 "<u>Subsidiary</u>" means any corporation, business trust, limited liability company or partnership with respect to which Allstate owns, directly or indirectly, Voting Securities representing more than 50% of the aggregate voting power of the thenoutstanding Voting Securities.
- 1.59 "<u>Surviving Corporation</u>" means the corporation resulting from a Reorganization Transaction or, if securities representing at least 50% of the aggregate Voting Power of such resulting corporation are directly or indirectly owned by another corporation, such other corporation.
- 1.60 "<u>Target Annual Bonus</u>" as of any date means the amount equal to the product of Base Salary determined as of such date multiplied by the percentage of such Base Salary to which Executive would have been entitled immediately prior to such date under any Bonus Plan for the Annual Performance Period for which the Annual Bonus is awarded if the performance goals established pursuant to such Bonus Plan were achieved at the 100% level as of the end of the Annual Performance Period.
  - 1.61 "<u>Taxes</u>" means federal, state, local and other income, employment and other taxes.
- 1.62 "<u>Termination Date</u>" means the date of the receipt of the Notice of Termination by Executive (if such Notice is given by the Company) or by the Company (if such Notice is given by Executive), or any later date, not more than 15 days after the giving of such Notice, specified in such Notice; provided, however, that:
  - (a) if Executive's employment is terminated by reason of death or Disability, the Termination Date shall be the date of Executive's death or the Disability Effective Date (as defined in Section 3.1(a)), as applicable; and
  - (b) if no Notice of Termination is given, the Termination Date shall be the last date on which Executive is employed by the Company.
- 1.63 "<u>Termination of Employment</u>" means any termination of Executive's employment with the Company, whether such occurs by reason of (a) the initiative of any Company or Executive or (b) the death of Executive; provided that such termination is also a "separation from service" within the meaning of Treasury Regulation 1.409A-1(h).
- 1.64 "<u>Voting Securities</u>" of a corporation means securities of such corporation that are entitled to vote generally in the election of directors of such corporation.

## ARTICLE II. POST-CHANGE PERIOD

#### 2.1 <u>Position and Duties</u>.

- (a) During the Post-Change Period, (x) Executive's position (including offices, titles, reporting requirements and responsibilities), authority and duties shall be at least commensurate in all material respects with the most significant of those held, exercised and assigned at any time during the 90-day period immediately before the Effective Date and (y) Executive's services shall be performed at the location where Executive was employed immediately before the Effective Date or any other location which does not constitute a material geographic change from the former location.
- (b) During the Post-Change Period (except during any periods of vacation to which Executive is entitled and any authorized sick, disability or other leave of absence), Executive shall devote Executive's full attention and time to the business and affairs of the Company and, to the extent necessary to discharge the duties assigned to Executive in accordance with this Agreement, to use Executive's best efforts to perform such duties. During the Post-Change Period, Executive may (i) serve on corporate, civic or charitable boards or committees, (ii) deliver lectures, fulfill speaking engagements or teach at educational institutions and (iii) manage personal investments, so long as such activities are consistent with the Policies of the Company at the Effective Date and do not significantly interfere with the performance of Executive's duties under this Agreement. To the extent that any such activities have been conducted by Executive immediately prior to the Effective Date and were consistent with the Policies of the Company at the Effective Date, the continued conduct of such activities (or activities similar in nature and scope) after the Effective Date shall not be deemed to interfere with the performance of Executive's duties under this Agreement.

#### 2.2 <u>Compensation</u>.

- (a) <u>Base Salary</u>. During the Post-Change Period, the Company shall pay or cause to be paid to Executive an annual base salary in cash, which shall be paid in a manner consistent with the Company's payroll practices in effect immediately before the Effective Date, at an annual rate not less than 12 times the highest monthly base salary paid or payable to Executive by the Company in respect of the 12-month period immediately before the Effective Date (such annual rate salary, the "<u>Base Salary</u>"). During the Post-Change Period, the Base Salary shall be reviewed at least annually and shall be increased at any time and from time to time as shall be substantially consistent with increases in base salary awarded to other peer executives of the Company. Any increase in Base Salary shall not limit or reduce any other obligation of the Company to Executive under this Agreement. After any such increase, the Base Salary shall not be reduced and "Base Salary" shall thereafter refer to the increased amount.
- (b) <u>Annual Bonus</u>. The Company shall also pay or cause to be paid to Executive a bonus (the "<u>Annual Bonus</u>"), which shall be not less than the Target Annual

-8-

Bonus determined as of the Effective Date, for each Annual Performance Period that ends during the Post-Change Period. "<u>Annual Performance Period</u>" means each period designated in accordance with any annual bonus arrangement or Plan (a "<u>Bonus Plan</u>") that is based on performance and approved by the Board or any committee of the Board, or in the absence of any Bonus Plan or any such designated period of time, each calendar year.

### (c) <u>LTIP Bonus</u>. The Company shall also:

- (i) pay or cause to be paid to Executive an LTIP Bonus equal to the LTIP Target Award for each LTIP Award for which an LTIP Performance Period is in effect as of the Effective Date; and
  - (ii) throughout the Post-Change Period, grant LTIP Awards to Executive as follows:
  - (1) LTIP Awards shall be granted no less frequently than is contemplated by the terms of the LTIP and the Company's practices thereunder, as such terms and practices are in effect immediately prior to the Effective Date:
  - (2) each such LTIP Award shall provide for the payment of a percentage of Executive's Base Salary in effect at the beginning of the Performance Period applicable to such LTIP Award that is no less than the average of the Target LTIP Percentages (as defined below) for all of Executive's LTIP Awards outstanding immediately prior to the Effective Date; and
  - (3) the target performance goals established for each such LTIP Award shall be substantially comparable to the target performance goals under Executive's LTIP Awards outstanding on the Effective Date;

"<u>Target LTIP Percentage</u>" means, in respect of any LTIP Award, the percentage of Executive's Base Salary (determined as of the beginning of the applicable LTIP Performance Period) that Executive would be entitled to receive after the completion of the applicable LTIP Performance Period if the performance goals applicable to such LTIP Award as of the date immediately prior to the Effective Date were achieved at the 100% level.

(d) <u>Incentive, Savings and Retirement Plans</u>. Executive shall also be entitled to participate during the Post-Change Period in all cash and equity incentive (including long-term incentives), savings and retirement Plans applicable to other peer executives of the Company, but in no event shall such Plans provide Executive with incentive (including long-term incentives), savings and retirement benefits during the Post-Change Period that are materially less valuable or have terms materially less favorable, in the aggregate, than the most valuable and favorable of those provided by the Company for Executive under such Plans as in effect at any time during the 90-day period immediately before the Effective Date.

-9-

- (e) <u>Welfare Benefit Plans</u>. During the Post-Change Period, Executive and Executive's family shall be eligible to participate in, and receive all benefits under, welfare benefit Plans provided by the Company (including medical, prescription, dental, disability, salary continuance, individual life, group life, dependent life, accidental death and travel accident insurance Plans) and applicable to other peer executives of the Company and their families, but in no event shall such Plans provide benefits during the Post-Change Period that are materially less favorable, in the aggregate, than the most favorable of those provided to Executive under such Plans as in effect at any time during the 90-day period immediately before the Effective Date.
- (f) <u>Fringe Benefits</u>. During the Post-Change Period, Executive shall be entitled to fringe benefits in accordance with the most favorable Plans applicable to peer executives of the Company, but in no event shall such Plans provide fringe benefits that are materially less favorable, in the aggregate, than the most favorable of those provided by the Company to Executive under such Plans in effect at any time during the 90-day period immediately before the Effective Date.
- (g) <u>Expenses</u>. During the Post-Change Period, Executive shall be entitled to prompt reimbursement of all reasonable employment-related expenses incurred by Executive upon the Company's receipt of accountings in accordance with the most favorable Policies applicable to peer executives of the Company, but in no event shall such Policies be materially less favorable, in the aggregate, than the most favorable of those provided by the Company for Executive under such Policies in effect at any time during the 90-day period immediately before the Effective Date.
- (h) Office and Support Staff. During the Post-Change Period, Executive shall be entitled to an office or offices of a size and with furnishings and other appointments, and to secretarial and other assistance in accordance with the most favorable Policies applicable to peer executives of the Company, but in no event shall such Policies be materially less favorable, in the aggregate, than the most favorable of those provided by the Company for Executive under such Policies in effect at any time during the 90-day period immediately before the Effective Date.
- (i) <u>Vacation</u>. During the Post-Change Period, Executive shall be entitled to paid vacation in accordance with the most favorable Policies applicable to peer executives of the Company, but in no event shall such Policies be materially less favorable, in the aggregate, than the most favorable of those provided by the Company for Executive under such Policies in effect at any time during the 90-day period immediately before the Effective Date.

#### 2.3 Stock Incentive Awards.

On the Effective Date of a Change of Control, (i) all of Executive's unvested Stock Options then outstanding (whether granted before or after the Agreement Date) shall immediately become fully vested and exercisable, and (ii) all of Executive's Restricted Shares then outstanding shall immediately become fully vested and nonforfeitable.

-10-

- 2.4 <u>Unfunded Deferred Compensation</u>. On the Effective Date of a Change of Control, Executive shall become fully vested in all benefits previously accrued under any deferred compensation Plan (including a SERP) that is not qualified under Section 401(a) of the Code (a "Non-Qualified Plan"). Within five business days after the Effective Date of a Change of Control, the Company shall pay to Executive a lump-sum cash amount equal to:
  - (a) the sum of the Lump-Sum Values of all Maximum Annuities that are payable pursuant to all defined benefit Non-Qualified Plans, plus
    - (b) the sum of Executive's account balances under all defined contribution Non-Qualified Plans.

To the extent that, if, for any reason, any portion of such Non-Qualified Plan benefit is not so paid, the Company shall pay Executive in lieu thereof a lump-sum cash payment equal to such unpaid portion within the five-business day period specified in the preceding sentence.

### ARTICLE III. TERMINATION OF EMPLOYMENT

#### 3.1 <u>Disability</u>.

- (a) During the Post-Change Period, the Company may terminate Executive's employment because of Executive's Disability by giving Executive or his legal representative, as applicable, (i) written notice in accordance with Section 10.8 of the Company's intention to terminate Executive's employment pursuant to this Section and (ii) a certification of Executive's Disability by a physician selected by the Company or its insurers, subject to the consent of Executive or Executive's legal representative, which consent shall not be unreasonably withheld or delayed. Executive's employment shall terminate effective on the 30th day (the "Disability Effective Date") after Executive's receipt of such notice unless, before the Disability Effective Date, Executive shall have resumed the full-time performance of Executive's duties.
  - (b) "<u>Disability</u>" means any medically determinable physical or mental impairment of an Executive that:
  - (i) has lasted for a continuous period of not less than (x) six months or (y) such longer period, if any, that is available to Executive under the Company's Policies relating to the continuation of employee status after the onset of disability, as such Policies are in effect when Disability is determined, but in no event shall such Policies be materially less favorable to the Executive than the most favorable of such Policies in effect for peer executives at any time during the 90-day period immediately before the Effective Date,
    - (ii) can be expected to be permanent or of indefinite duration, and

-11-

- (iii) renders Executive unable to perform the duties required under this Agreement.
- 3.2 <u>Death</u>. Executive's employment shall terminate automatically upon Executive's death during the Post-Change Period.
  - 3.3 Cause.
  - (a) During the Post-Change Period, the Company may terminate Executive's employment for Cause solely in accordance with all of the substantive and procedural provisions of this Section.
    - (b) "Cause" means any one or more of the following:
      - (i) Executive's conviction of a felony or other crime involving fraud, dishonesty or moral turpitude;
      - (ii) Executive's willful or reckless material misconduct in the performance of Executive's duties;
      - (iii) Executive's habitual neglect of duties; or
      - (iv) Executive's willful or intentional breach of this Agreement;

provided, however, that for purposes of clauses (ii), (iii), and (iv), Cause shall not include any one or more of the following:

- (1) bad judgment or negligence;
- (2) any act or omission believed by Executive in good faith to have been in or not opposed to the interest of the Company (without intent of Executive to gain, directly or indirectly, a profit to which Executive was not legally entitled);
- (3) any act or omission with respect to which a determination could properly have been made by the Board that Executive had satisfied the applicable standard of conduct for indemnification or reimbursement under Allstate's by-laws, any applicable indemnification agreement, or applicable law, in each case as in effect at the time of such act or omission; or
- (4) any act or omission with respect to which Executive receives a Notice of Consideration (as defined below) more than six months after the earliest date on which any member of the Board, not a party to the act or omission, knew or should have known of such act or omission; and

further provided, that if a breach of this Agreement involved an act or omission based on Executive's good faith and reasonable belief that Executive's act or omission was in the best interests of the Company or was required by applicable law or administrative regulation, such breach shall not constitute Cause unless the Company gives Executive written notice of such breach that specifically refers to this Section and, within 30 days after such notice is given, Executive fails to cure such breach to the fullest extent that it is curable.

- (c) The Company shall strictly observe each of the following procedures in connection with any Termination of Employment for Cause:
  - (i) A meeting of the Board shall be called for the stated purpose of determining whether Executive's acts or omissions satisfy the requirements of Section 3.3(b) and, if so, whether to terminate Executive's employment for Cause.
  - (ii) Not less than 30 days prior to the date of such meeting, the Company shall provide Executive and each member of the Board written notice (a "Notice of Consideration") of (x) a detailed description of the acts or omissions alleged to constitute Cause, (y) the date, time and location of such meeting of the Board, and (z) Executive's rights under clause (iii) below.
  - (iii) Executive shall have the opportunity to appear before the Board in person and, at Executive's option, with legal counsel, and/or to present to the Board a written response to the Notice of Consideration.
  - (iv) Executive's employment may be terminated for Cause only if (x) the acts or omissions specified in the Notice of Consideration did in fact occur and do constitute Cause as defined in this Section, (y) the Board makes a specific determination to such effect and to the effect that Executive's employment should be terminated for Cause and (z) the Company thereafter provides Executive with a Notice of Termination that specifies in specific detail the basis of such Termination of Employment for Cause and which Notice shall be consistent with the reasons set forth in the Notice of Consideration. The Board's determination specified in clause (y) of the preceding sentence shall require the affirmative vote of at least 75% of the members of the Board.
  - (v) In the event that the existence of Cause shall become an issue in any action or proceeding between the Company and Executive, the Company shall, notwithstanding the determination referenced in clause (iv) of this Section 3.3(c), have the burden of establishing that the actions or omissions specified in the Notice of Consideration did in fact occur and do constitute Cause and that the Company has satisfied the procedural requirements of this Section 3.3(c). The satisfaction of the Company's burden shall require clear and convincing evidence.

#### 3.4 Good Reason.

(a) During the Post-Change Period, Executive may terminate his employment for Good Reason in accordance with the substantive and procedural provisions of this

-13-

Section. A Termination of Employment for Good Reason will be deemed to have occurred during the Post-Change Period if Executive gives notice as provided in Section 3.4(d) within the Post-Change Period and the Termination of Employment is no more than thirty (30) days after the expiration of the cure period described in Section 3.4(e).

- (b) "<u>Good Reason</u>" means the first to occur of the following actions or omissions that, unless otherwise specified, occurs during a Post-Change Period without the consent of Executive:
  - (i) a material diminution in Executive's base compensation;
  - (ii) any material diminution in Executive's authority, duties, or responsibilities as set forth in Paragraph 2.1(a);
  - (iii) any material diminution in the authority, duties, or responsibilities of the person to whom Executive reports, including a requirement that Executive report to a corporate officer or employee instead of reporting directly to the Board, if applicable;
    - (iv) a material change in the geographic location at which Executive must perform services; or
    - (v) any other action or inaction that constitutes a material breach of this Agreement by the Company;

- (c) Any reasonable determination by Executive that any of the events specified in subsection (b) above has occurred and constitutes Good Reason shall be conclusive and binding for all purposes, unless the Company establishes by clear and convincing evidence that Executive did not have any reasonable basis for such determination.
- (d) In the event of any Termination of Employment by Executive for Good Reason, Executive shall notify the Company of the events constituting such Good Reason by a Notice of Termination within ninety days of the date Executive should have known of the events constituting Good Reason.
- (e) Company shall have thirty days from the date Executive provides Notice of Termination pursuant to Section 3.4(d) to remedy the conditions constituting Good Reason during which period no termination for Good Reason shall be deemed to have occurred.
- (f) If the Company has not remedied the conditions constituting Good Reason within the thirty-day period described in Section 3.4(e), then, in order for Executive's termination to constitute a termination for Good Reason, the date of Termination of Employment must occur no later than twelve (12) months after the date of the first action or omission constituting Good Reason.

-14-

## ARTICLE IV. COMPANY'S OBLIGATIONS UPON A TERMINATION OF EMPLOYMENT

- 4.1 <u>If by Executive for Good Reason or by the Company Other Than for Cause or Disability</u>. If, during the Post-Change Period, the Company terminates Executive's employment other than for Cause or Disability, or if Executive terminates employment for Good Reason, the Company's sole obligations to Executive under Sections 2.1 and 2.2 and this Article shall be as follows:
  - (a) The Company shall pay Executive, in addition to all vested rights arising from Executive's employment as specified in Article II, a lump-sum cash amount equal to the sum of the following:
    - (i) all Accrued Obligations;
    - (ii) Executive's Pro-rata Annual Bonus reduced (but not below zero) by the amount of any Annual Bonus paid to Executive with respect to the Company's fiscal year in which the Termination Date occurs;
    - (iii) Executive's Pro-rata LTIP Bonus reduced (but not below zero) by the amount of any LTIP Bonus paid to Executive with respect to the Company's fiscal year in which the Termination Date occurs;
    - (iv) all amounts previously deferred by, or accrued to the benefit of, Executive under any defined contribution Non-Qualified Plans, whether or not vested, together with any accrued earnings thereon, to the extent that such amounts and earnings have not been previously paid by the Company (whether pursuant to Section 2.4 or otherwise);
    - (v) an amount equal to three (3.0) times the sum of (y) Base Salary, and (z) the Target Annual Bonus, each determined as of the Termination Date; provided, however, that any reduction in Executive's Base Salary or Target Annual Bonus that would qualify as Good Reason shall be disregarded for this purpose; and
    - (vi) to the extent not paid pursuant to clause (iv) of this Section 4.1(a), an amount equal to the sum of the value of the unvested portion of Executive's accounts or accrued benefits under any defined contribution Plan (whether or not qualified under Section 401(a) of the Code) maintained by the Company as of the Termination Date and forfeited by Executive by reason of the Termination of Employment.

Such lump-sum amount shall be paid no more than five business days after the date of Termination of Employment.

(b) The Company shall pay Executive, in lieu of all benefits under all defined benefit Non-Qualified Plans that have accrued on or before the Termination Date but

-15-

remain unpaid as of such date, a lump-sum cash amount equal to the positive difference, if any, between:

(i) the sum of the Lump-Sum Values of each Maximum Annuity that would be payable to Executive under any defined benefit Plan (whether or not qualified under Section 401(a) of the Code) if Executive had:

- (1) become fully vested in all such benefits to the extent that such benefits are unvested as of the Termination Date,
- (2) attained as of the Termination Date an age that is three years greater than Executive's actual age,
- (3) accrued a number of years of service (for purposes of determining the amount of such benefits, entitlement to early retirement benefits, and all other purposes of such defined benefit plans) that is three years greater than the number of years of service actually accrued by Executive as of the Termination Date, and
- (4) received the lump-sum severance benefits specified in Section 4.1(a) (excluding all LTIP Bonuses, and all amounts in respect of Stock Options or Restricted Shares, if any) as covered compensation in equal monthly installments during the three year period following Termination of Employment,

minus

(ii) the sum of (x) the Lump-Sum Values of the Maximum Annuity benefits vested and payable (whether currently or at some future date) to Executive under each defined benefit Plan that is qualified under Section 401(a) of the Code and (y) the aggregate amounts simultaneously or previously paid (whether pursuant to Section 2.4 or otherwise) to Executive under the defined benefit Plans (whether or not qualified under Section 401(a) of the Code) described in clause (i) of this Section 4.1(b).

Such lump-sum amount shall be paid no more than five business days after the date of Termination of Employment.

- (c) (i) On the date of Termination of Employment, all of Executive's unvested Stock Options then outstanding (whether granted before or after the Agreement Date) shall immediately become fully vested and exercisable, and (ii) all of Executive's Restricted Shares then outstanding shall immediately become fully vested and nonforfeitable. This Section 4.1(c) amends all award agreements dated as of any date before the Agreement Date.
- (d) All of Executive's then-outstanding Stock Options that were granted after the Agreement Date, whether vested on or before the date of Termination of Employment, shall thereafter remain exercisable until the last to occur of (x) the first

-16-

anniversary of the date of Termination of Employment, and (y) any period provided in the applicable stock option agreement or stock option plan as then in effect, but in no event shall such period of exercisability continue after the earlier of (i) the date on which such Stock Options would have expired if Executive had remained an employee of the Company, or (ii) the tenth anniversary of the original date of the Stock Option grant.

- (e) Within five business days after date of Executive's Termination of Employment, the Company shall deliver to Executive certificates for all Restricted Shares theretofore held by or on behalf of the Company.
- (f) The Company shall pay on behalf of Executive all reasonable fees and costs charged by the outplacement firm selected by Executive to provide outplacement services to Executive that are incurred no later than the end of the second year following the year in which the Termination of Employment occurs.
- g) During the period of time which Executive would be entitled to continuation coverage under a Company-sponsored group health plan under Section 4980 of the Code or such later date as any Plan may specify, the Company shall continue to make available to Executive and Executive's family welfare benefits (including medical, prescription, dental, disability, salary continuance, individual life, group life, accidental death and travel accident insurance plans and programs) that are at least as favorable as the most favorable Plans of the Company applicable to other peer executives and their families as of the Termination Date, but which are in no event less favorable than the most favorable Plans of the Company applicable to other peer executives and their families during the 90-day period immediately before the Effective Date. The cost of such welfare benefits, including continuation coverage required by Section 4980 of the Code ("COBRA"), to Executive shall not exceed the cost of such benefits to Executive immediately before the Termination Date or, if less, the Effective Date. Executive's rights under this Section shall be co-extensive with any post-termination continuation coverage Executive may have pursuant to applicable law, including COBRA,. Accordingly, in order to receive this coverage, Executive shall timely elect continuation coverage under COBRA for Executive and Executive's covered dependents. Notwithstanding any of the above, such welfare benefits shall be secondary to any similar welfare benefits provided by Executive's subsequent employer as provided in the Plans.

- 4.2 <u>If by the Company for Cause</u>. If the Company terminates Executive's employment for Cause during the Post-Change Period, the Company's sole obligation to Executive under Sections 2.1 and 2.2 and this Article shall be to pay Executive a lump-sum cash amount equal to all Accrued Obligations determined as of the Termination Date.
- 4.3 <u>If by Executive Other Than for Good Reason</u>. If Executive terminates employment during the Post-Change Period other than for Good Reason, Disability or death, the Company's sole obligation to Executive under Sections 2.1 and 2.2 and this Article shall be to pay Executive a lump-sum cash amount equal to all Accrued Obligations determined as of the Termination Date.

-17-

- 4.4 <u>If by the Company for Disability</u>. If the Company terminates Executive's employment by reason of Executive's Disability during the Post-Change Period, the Company's sole obligation to Executive under Sections 2.1 and 2.2 and this Article shall be as follows:
  - (a) to pay Executive a lump-sum cash amount equal to all Accrued Obligations determined as of the Termination Date, and
  - (b) to provide Executive disability and other benefits after the Termination Date that are not less favorable to Executive than the most favorable of such benefits then available under Plans of the Company to disabled peer executives of the Company.

Such disability and other benefits shall also be not materially less favorable, in the aggregate, to Executive than the most favorable of the disability and other benefits available to Executive under such Plans in effect at any time during the 90-day period immediately preceding the Effective Date.

- 4.5 <u>If Upon Death</u>. If Executive's employment is terminated by reason of Executive's death during the Post-Change Period, the Company's sole obligations to Executive under Sections 2.1 and 2.2 and this Article shall be as follows:
  - (a) to pay Executive's estate or Beneficiary a lump-sum cash amount equal to all Accrued Obligations; and
  - (b) to provide Executive's estate or Beneficiary survivor and other benefits that are not less than the most favorable survivor and other benefits then available under Plans of the Company to the estates or the surviving families of peer executives of the Company.

Such survivor benefits shall also be no less favorable, in the aggregate, than the most favorable of the survivor benefits available to Executive under such Plans in effect at any time during the 90-day period immediately preceding the Effective Date.

#### 4.6 Amount Contested.

(a) In the event of any dispute between the Company and Executive as to the nature or extent of the Company's obligation to make any payments or provide other benefits to Executive or Executive's family pursuant to Sections 4.1 or 2.4, Executive shall have the right, exercisable by written notice given to the Company within 90 days after the Executive believes a payment or provision of benefits should have occurred, to obtain, within 30 days after the Company's receipt of Executive's demand therefor, a written certificate prepared by the Company and certified by Allstate's independent auditors (a "Section 4.6 Certificate"). The Section 4.6 Certificate shall specify in detail either (i) the amount and nature of each payment or other benefit that the Company believes is then due and owing to Executive pursuant to Section 2.4 or 4.1, as applicable, or (ii) if the Company asserts that the conditions to Executive's entitlement to severance or other benefits pursuant to Section 4.1 or 2.4, as applicable, have for any reason not been satisfied, the amount and nature of each payment or other benefit that the Company believes would be due and owing to Executive pursuant to Section 4.1 or 2.4, as

-18-

applicable, if all of such applicable conditions had been fully satisfied. Executive may not demand more than one Section 4.6 Certificate in respect of his rights under Section 4.1 or more than one Section 4.6 Certificate in respect of his rights under Section 2.4.

- (b) Each Section 4.6 Certificate shall include schedules that specify in detail how each amount or other benefit specified therein was computed, together with appropriate references to specific provisions of this Agreement or of any applicable Plans or Policies of the Company, copies of which Plans or Policies shall be attached to such schedules.
- (c) The Company shall be precluded from asserting that any portion of the payments or other benefits due to Executive pursuant to Section 4.1 or 2.4, as applicable, is less than the amount specified in the Section 4.6 Certificate. The Section 4.6 Certificate shall in no event be binding on Executive and Executive shall have the right to assert that any or all

of the payments or other benefits to be provided pursuant to Section 4.1 or 2.4 are greater than or different from those specified in the Section 4.6 Certificate.

(d) If the Company shall for any reason fail to deliver to Executive a Section 4.6 Certificate in compliance with this Section within 30 days after the Company's receipt of Executive's written demand therefor, Executive's determination of the amount and nature of payments or other benefits due to Executive (i) pursuant to Section 4.1 and set forth in an Executive's Deferred Compensation Determination (as defined below) or (ii) pursuant to Section 2.4 and set forth in an Executive's Deferred Compensation Determination or Executive's Severance Determination or Executive's Deferred Compensation Determination, as applicable, is incorrect and that a different amount (which may be zero or a positive amount) or nature of payments or other benefits is correct. "Executive's Severance Determination" means an opinion of nationally recognized executive compensation counsel to the effect that the amount and nature of severance and other benefits due to Executive pursuant to Section 4.1 is the amount and nature that a court of competent jurisdiction, based on a final judgment not subject to further appeal, is most likely to decide to have been calculated in accordance with this Agreement and applicable law. "Executive's Deferred Compensation Determination" means an opinion of nationally recognized executive compensation counsel to the effect that the amount of payments due to Executive pursuant to Section 2.4 is the amount that a court of competent jurisdiction, based on a final judgment not subject to further appeal, is most likely to decide to have been calculated in accordance with this Agreement and applicable law.

-19-

## ARTICLE V. REDUCTIONS IN PAYMENTS BY THE COMPANY

#### 5.1 Reduction of Potential Parachute Payments.

- (a) If it is determined by Allstate's independent auditors that any monetary or other benefit received or deemed received by Executive from the Company or any Affiliate pursuant to this Agreement or otherwise, whether or not in connection with a Change of Control (such monetary or other benefits collectively, the "Potential Parachute Payments"), is or will become subject to any excise tax under Section 4999 of the Code or any similar tax under any United States federal, state, local or other law other than Section 409A of the Code (such excise tax and all such similar taxes collectively, "Excise Taxes"), and reducing the Potential Parachute Payments would increase the aggregate amount received by Executive (after taking into consideration the payment of all income and Excise Taxes that would be owing as the result of the Potential Parachute Payments) the Company shall reduce the Potential Parachute Payments by the amount necessary to maximize such amounts for Executive. The reduction in Potential Parachute Payments will be determined on an after-tax basis in the following order:
  - (i) all amounts that are not deferred compensation for purposes of Section 409A of the Code, and
  - (ii) all amounts that are deferred compensation for purposes for Section 409A of the Code,

in each case such amounts to be reduced in the order that results in the greatest reduction in the amount valued as a "parachute payment" as that term is used in Section 280G of the Code and that results in the least economic impact to Executive as determined by Allstate's independent auditors.

- (b) For purposes of determining the amount of an Executive's Change in Control Benefits on an after-tax basis, the Executive shall be deemed to be subject to the highest effective marginal rate of Taxes.
- 5.2 <u>Determination by Auditors</u>. The determination of Allstate's independent auditors described in Section 5.1(a), including the detailed calculations of the amounts of the Potential Parachute Payments, and the assumptions relating thereto, shall be set forth in a written certificate of such auditors (the "<u>Company Certificate</u>") delivered to Executive. Executive or the Company may at any time request the preparation and delivery to Executive of a Company Certificate. The Company shall cause the Company Certificate to be delivered to Executive as soon as reasonably possible after such request.

-20-

# ARTICLE VI. EXPENSES AND INTEREST

#### 6.1 Legal and Other Expenses.

(a) If Executive incurs legal fees or other expenses (including expert witness and accounting fees) in an effort to determine, secure, preserve, establish entitlement to, or obtain benefits under this Agreement (collectively, "<u>Legal and Other</u>

<u>Expenses</u>"), the Company shall, regardless of the outcome of such effort, pay or reimburse Executive for such Legal and Other Expenses in accordance with Section 6.1(b).

- (b) All Legal and Other Expenses shall be paid or reimbursed on a monthly basis within 10 days after Allstate's receipt of Executive's written request accompanied by evidence that such Legal and Other Expenses were incurred.
- (c) If Executive does not prevail (after exhaustion of all available judicial remedies) in respect of a claim by Executive or by the Company hereunder, and the Company establishes before a court of competent jurisdiction, by clear and convincing evidence, that Executive had no reasonable basis for his claim hereunder, or for his response to the Company's claim hereunder, or acted in bad faith, no further payment of or reimbursement for Legal and Other Expenses shall be due to Executive in respect of such claim and Executive shall refund any amounts previously paid or reimbursed hereunder with respect to such claim.
- 6.2 <u>Interest</u>. If the Company does not pay an amount due to Executive under this Agreement within five business days after such amount first became due and owing, interest shall accrue on such amount from the date it became due and owing until the date of payment at an annual rate equal to 200 basis points above the base commercial lending rate published in *The Wall Street Journal* in effect from time to time during the period of such nonpayment.

### ARTICLE VII. NO SET-OFF OR MITIGATION

- 7.1 No Set-off by Company. Executive's right to receive when due the payments and other benefits provided for under this Agreement is absolute, unconditional and subject to no set-off, counterclaim or legal or equitable defense. Time is of the essence in the performance by the Company of its obligations under this Agreement. Any claim that the Company may have against Executive, whether for a breach of this Agreement or otherwise, shall be brought in a separate action or proceeding and not as part of any action or proceeding brought by Executive to enforce any rights against the Company under this Agreement, except if (i) the Company's claim is determined by a court to be a compulsory counterclaim under applicable law or (ii) if a court determines that the Company would otherwise be materially prejudiced if its claim were to be brought in a separate action.
- 7.2 <u>No Mitigation</u>. Executive shall not have any duty to mitigate the amounts payable by the Company under this Agreement by seeking new employment or self-employment following termination. Except as specifically otherwise provided in this Agreement, all amounts

-21-

payable pursuant to this Agreement shall be paid without reduction regardless of any amounts of salary, compensation or other amounts that may be paid or payable to Executive as the result of Executive's employment by another employer or self-employment.

### ARTICLE VIII. RESTRICTIVE COVENANTS

- 8.1 <u>Non-Competition</u>. If Executive remains employed by the Company on the Effective Date, Executive shall not at any time during the period beginning on the Effective Date and ending on the first anniversary of the Termination Date, directly or indirectly, in any capacity:
  - (a) engage or participate in, become employed by, serve as a director of, or render advisory or consulting or other services in connection with, any Competitive Business; provided, however, that this Section 8.1(a) shall not preclude Executive from being an employee of, or consultant to, any business unit of a Competitive Business if (i) such business unit does not qualify as a Competitive Business in its own right and (ii) Executive does not have any direct or indirect involvement in, or responsibility for, any operations of such Competitive Business that cause it to qualify as a Competitive Business; or
  - (b) make or retain any financial investment, whether in the form of equity or debt, or own any interest, in any Competitive Business; provided, however, that nothing in this subsection shall restrict Executive from making an investment in any Competitive Business if such investment (i) represents no more than 1% of the aggregate market value of the outstanding capital stock or debt (as applicable) of such Competitive Business, (ii) does not give Executive any right or ability, directly or indirectly, to control or influence the policy decisions or management of such Competitive Business, and (iii) does not create a conflict of interest between Executive's duties under this Agreement and his interest in such investment.
- 8.2 <u>Non-Solicitation</u>. If Executive remains employed by the Company on the Effective Date, Executive shall not at any time during the period beginning on the Effective Date and ending on the first anniversary of the Termination Date, directly or indirectly:

- (a) other than in connection with the good-faith performance of his duties as an officer of the Company, encourage any employee or agent of the Company to terminate his relationship with the Company;
- (b) employ, engage as a consultant or adviser, or solicit the employment or engagement as a consultant or adviser, of any employee or agent of the Company (other than by the Company or its Affiliates), or cause or encourage any Person to do any of the foregoing;
- (c) establish (or take preliminary steps to establish) a business with, or encourage others to establish (or take preliminary steps to establish) a business with, any employee or agent of the Company; or

-22-

(d) interfere with the relationship of the Company with, or endeavor to entice away from the Company, any Person who or which at any time during the period commencing one year prior to the Agreement Date was or is a material customer or material supplier of, or maintained a material business relationship with, the Company.

#### 8.3 Reasonableness of Restrictive Covenants.

- (a) Executive acknowledges that the covenants contained in Sections 8.1 and 8.2 are reasonable in the scope of the activities restricted, the geographic area covered by the restrictions, and the duration of the restrictions, and that such covenants are reasonably necessary to protect the Company's relationships with its employees, customers and suppliers. Executive further acknowledges such covenants are essential elements of this Agreement and that, but for such covenants, the Company would not have entered into this Agreement.
- (b) The Company and Executive have each consulted with their respective legal counsel and have been advised concerning the reasonableness and propriety of such covenants. Executive acknowledges that his observance of the covenants contained in Sections 8.1 and 8.2 will not deprive him of the ability to earn a livelihood or to support his dependents.

#### 8.4 <u>Right to Injunction; Survival of Undertakings</u>.

- (a) In recognition of the necessity of the limited restrictions imposed by Sections 8.1 and 8.2, the parties agree that it would be impossible to measure solely in money the damages that the Company would suffer if Executive were to breach any of his obligations under such Sections. Executive acknowledges that any breach of any provision of such Sections would irreparably injure the Company. Accordingly, Executive agrees that the Company shall be entitled, in addition to any other remedies to which the Company may be entitled under this Agreement or otherwise, to an injunction to be issued by a court of competent jurisdiction, to restrain any actual breach, or threatened breach, of such provisions, and Executive hereby waives any right to assert any defense that the Company has an adequate remedy at law for any such breach.
- (b) If a court determines that any of the covenants included in this Article VIII is unenforceable in whole or in part because of such covenant's duration or geographical or other scope, such court may modify the duration or scope of such provision, as the case may be, so as to cause such covenant as so modified to be enforceable.
- (c) All of the provisions of this Article VIII shall survive any Termination of Employment without regard to (i) the reasons for such termination or (ii) the expiration of the Agreement Term.
- 8.5 <u>Non-Disparagement</u>. If Executive remains employed by the Company on the Effective Date, Executive shall not at any time during the two-year period commencing on the Termination Date (a) make any written or oral statement that brings the Company or any of its then-current or former employees, officers or agents into disrepute, or tarnishes any of their images or reputations or (b) publish, comment on or disseminate any statements suggesting or

-23-

accusing the Company or any of its then-current or former agents, employees or officers of any misconduct or unlawful behavior. This Section shall not be deemed to be breached by testimony of Executive given in any judicial or governmental proceeding that Executive reasonably believes to be truthful at the time given or by any other action of Executive that he reasonably believes is taken in accordance with the requirements of applicable law or administrative regulation.

# ARTICLE IX. NON-EXCLUSIVITY OF RIGHTS

9.1 <u>Waiver of Certain Other Rights</u>. To the extent that Executive shall have received severance payments or other severance benefits under any other Plan or agreement of the Company prior to receiving severance payments or other severance

benefits pursuant to Article IV, the severance payments and other severance benefits under such other Plan or agreement shall reduce (but not below zero) the corresponding severance payments or other severance benefits to which Executive shall be entitled under Article IV. To the extent that Executive receives payments or other benefits pursuant to Article IV, Executive hereby waives the right to receive a corresponding amount of future severance payments or other severance benefits under any other Plan or agreement of the Company. To the extent that Executive receives payments pursuant to Section 4.1(b), Executive hereby waives the right to receive payments or other benefits under any Non-Qualified Plan that have accrued as of the Termination Date. To the extent that Executive received payments or other benefits pursuant to Section 4.1(a)(ii) or (iii), Executive hereby waives the right to receive any Annual Bonus or LTIP Bonus payments with respect to the Annual Performance Period or LTIP Performance Periods in effect as of the Termination Date.

9.2 Other Rights. Except as expressly provided in Section 9.1, this Agreement shall not prevent or limit Executive's continuing or future participation in any benefit, bonus, incentive or other Plans provided by the Company and for which Executive may qualify, nor shall this Agreement limit or otherwise affect such rights as Executive may have under any other agreements with the Company. Amounts that are vested benefits or which Executive is otherwise entitled to receive under any Plan and any other payment or benefit required by law at or after the Termination Date shall be payable in accordance with such Plan or applicable law except as expressly modified by this Agreement.

### ARTICLE X. MISCELLANEOUS

- 10.1 <u>No Assignability</u>. This Agreement is personal to Executive and without the prior written consent of the Company shall not be assignable by Executive otherwise than by will or the laws of descent and distribution. This Agreement shall inure to the benefit of and be enforceable by Executive's legal representatives.
- 10.2 <u>Successors</u>. This Agreement shall inure to the benefit of and be binding on the Company and its successors and assigns. The Company will require any successor (whether

-24-

direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of the Company to assume expressly and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform it if no such succession had taken place. Any successor to the business or assets of the Company that assumes or agrees to perform this Agreement by operation of law, contract, or otherwise shall be jointly and severally liable with the Company under this Agreement as if such successor were the Company.

- 10.3 <u>Payments to Beneficiary</u>. If Executive dies before receiving amounts to which Executive is entitled under this Agreement, such amounts shall be paid in a lump sum to one or more beneficiaries designated in writing by Executive (each, a "Beneficiary"), or if none is so designated, to Executive's estate.
- 10.4 <u>Non-Alienation of Benefits</u>. Benefits payable under this Agreement shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge, garnishment, execution or levy of any kind, either voluntary or involuntary, before actually being received by Executive, and any such attempt to dispose of any right to benefits payable under this Agreement shall be void.
- 10.5 <u>No Deference</u>. Unless otherwise expressly provided in this Agreement, no determination pursuant to, or interpretation of, this Agreement made by the board of directors (or any committee thereof) of Allstate or any Successor Corporation shall be entitled to any presumptive validity or other deference in connection with any judicial or administrative proceeding relating to or arising under this Agreement.
- 10.6 <u>Severability</u>. If any one or more Articles, Sections or other portions of this Agreement are declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity shall not serve to invalidate any Article, Section or other portion not so declared to be unlawful or invalid. Any Article, Section or other portion so declared to be unlawful or invalid shall be construed so as to effectuate the terms of such Article, Section or other portion to the fullest extent possible while remaining lawful and valid.
- 10.7 <u>Amendments</u>. This Agreement shall not be amended or modified except by written instrument executed by Executive, Allstate and AIC.
- 10.8 <u>Notices</u>. All notices and other communications under this Agreement shall be in writing and delivered by hand, by nationally recognized delivery service that promises overnight delivery, or by first-class registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to Executive, to Executive at his most recent home address on file with the Company.

If to Allstate or AIC:

Attention: General Counsel

or to such other address as either party shall have furnished to the other in writing. Notice and communications shall be effective when actually received by the addressee.

- 10.9 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts, each of which shall be deemed an original, but all of which together constitute one and the same instrument.
- 10.10 <u>Governing Law</u>. This Agreement shall be interpreted and construed in accordance with the laws of the State of Illinois, without regard to its choice of law principles.
  - 10.11 <u>Captions</u>. The captions of this Agreement are not a part of the provisions hereof and shall have no force or effect.
- 10.12 <u>Number and Gender</u>. Wherever appropriate, the singular shall include the plural, the plural shall include the singular, and the masculine shall include the feminine.
- 10.13 <u>Tax Withholding</u>. The Company may withhold from any amounts payable under this Agreement any Taxes that are required to be withheld by any applicable law or regulation.
- 10.14 <u>No Waiver</u>. Executive's failure to insist upon strict compliance with any provision of this Agreement shall not be deemed a waiver of such provision or any other provision of this Agreement. A waiver of any provision of this Agreement shall not be deemed a waiver of any other provision, and any waiver of any default in any such provision shall not be deemed a waiver of any later default thereof or of any other provision.
- 10.15 <u>Joint and Several Liability</u>. The obligations of Allstate and AIC to Executive under this Agreement shall be joint and several.
- 10.16 <u>No Rights Prior to Effective Date</u>. Notwithstanding any provision of this Agreement to the contrary, this Agreement shall not entitle Executive to any compensation, severance or other benefits of any kind prior to an Effective Date.
- 10.17 <u>Six-month Delay.</u> Any payment considered to be deferred compensation under Section 409A of the Code and not subject to an exception or exemption thereunder, shall not be paid to Executive prior to the first business day following the date that is six (6) months after the date of Executive's Termination of Employment. Any payments that would otherwise have been made to Executive during such six (6) month period shall instead be aggregated and not paid to Executive prior to the first business day following the date that is six (6) months after the date of Executive's Termination of Employment. Any payments subject to a sixmonth delay shall be paid with interest which shall accrue from the date such payment became due and owing until the date of payment at an annual rate equal to 200 basis points above the base commercial lending rate published in *The Wall Street Journal* in effect from time to time during the six-month delay period. Any payments scheduled to be made after the date that is six (6) months after the Termination of Employment shall be paid to Executive in accordance with the other provisions of this Agreement or applicable plan.

-26-

- 10.18 <u>Interpretation to Avoid 409A Penalties</u>. This Agreement is intended to comply with the provisions of Section 409A of the Code so as to avoid the imposition of excise taxes and penalties on the Executive under Section 409A of the Code. The Agreement shall be interpreted, construed and administered consistent with that intent.
- 10.19 <u>Entire Agreement</u>. This Agreement contains the entire understanding of Allstate, AIC and Executive with respect to its subject matter.

IN WITNESS WHEREOF, Executive, Allstate and AIC have executed this Change of Control Employment Agreement as of the date first above written.

[insert name of Executive]
THE ALLSTATE CORPORATION
By: Title:
ALLSTATE INSURANCE COMPANY
By: Title:
-27-

The Allstate Corporation 2775 Sanders Road Northbrook, IL 60062

We have reviewed, in accordance with the standards of the Public Company Accounting Oversight Board (United States), the unaudited interim financial information of The Allstate Corporation and subsidiaries (the "Company") for the periods ended June 30, 2011 and 2010, as indicated in our report dated July 29, 2011; because we did not perform an audit, we expressed no opinion on that information.

We are aware that our report referred to above, which is included in your Quarterly Report on Form 10-Q for the quarter ended June 30, 2011, is incorporated by reference in the following Registration Statements:

Form S-3 Registration Statement Nos.	Form S-8 Registration Statement Nos.
333-34583	33-77928
333-159071	33-93762
	333-04919
	333-16129
	333-40283
	333-60916
	333-120344
	333-134242
	333-134243
	333-144691
	333-144692
	333-158581
	333-159343
	333-175526
	333-175528

We also are aware that the aforementioned report, pursuant to Rule 436(c) under the Securities Act of 1933, is not considered a part of the Registration Statement prepared or certified by an accountant or a report prepared or certified by an accountant within the meaning of Sections 7 and 11 of that Act.

/s/ Deloitte & Touche LLP Chicago, Illinois July 29, 2011 CERTIFICATIONS EXHIBIT 31 (i)

- I, Thomas J. Wilson, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of The Allstate Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

July 29, 2011

/s/ Thomas J. Wilson
Thomas J. Wilson
Chairman of the Board,
President and Chief Executive Officer

E-3

CERTIFICATIONS EXHIBIT 31 (i)

- I, Don Civgin, certify that:
- 1. I have reviewed this quarterly report on Form 10-Q of The Allstate Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
- a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
- b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
- c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
- d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and

- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
- a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
- b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

July 29, 2011

<u>/s/ Don Civgin</u>
Don Civgin
Executive Vice President and Chief Financial Officer

#### **SECTION 1350 CERTIFICATIONS**

Each of the undersigned hereby certifies that to his knowledge the quarterly report on Form 10-Q for the fiscal period ended June 30, 2011 of The Allstate Corporation filed with the Securities and Exchange Commission fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934 and that the information contained in such report fairly presents, in all material respects, the financial condition and result of operations of The Allstate Corporation.

July 29, 2011

/s/ Thomas J. Wilson Thomas J. Wilson Chairman of the Board, President and Chief Executive Officer

<u>/s/ Don Civgin</u>
Don Civgin
Executive Vice President and Chief Financial Officer